UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): February 11, 2013

ModusLink Global Solutions, Inc.

(Exact name of registrant as specified in charter)

Delaware (State or other jurisdiction of incorporation)

001-35319 (Commission File Number) 04-2921333 (IRS Employer Identification No.)

1601 Trapelo Road Waltham, Massachusetts 02451 (Address of Principal Executive Offices)

(781) 663-5000

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- □ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry into a Material Definitive Agreement

Investment Agreement

Generally

On February 11, 2013, ModusLink Global Solutions, Inc. (the "Company") and with Steel Partners Holdings L.P. ("Steel Holdings"), an affiliate of Handy & Harman, Ltd. ("HNH"), reached an agreement regarding the sale and issuance of 7,500,000 shares of the Company's common stock at a price of \$4.00 per share and warrants to purchase 2,000,000 shares of the Company's common stock at an exercise price of \$5.00 per share (and the issuance of such shares upon exercise of such warrants) to Steel Holdings pursuant to the terms and subject to the conditions set forth in the Investment Agreement (the "Investment Agreement"), dated February 11, 2013, by and between the Company and Steel Holdings (the "Steel Holdings Investment").

The Investment Agreement contains customary representations and warranties by the Company relating to, among other things, its corporate organization and subsidiaries, the due authorization of the Investment Agreement and the shares of Common Stock and Warrants to be issued, the capitalization of the Company, the lack of required governmental consents, compliance with laws and other instruments, litigation, its filings with the SEC, including the financial statements included therein, undisclosed liabilities, absence of certain events and changes, litigation, taxes, the broker fees payable by the Company, solvency and internal controls. The Company also represented that the Company's board of directors (the "Board") had taken all actions necessary to render inapplicable the provisions of Section 203 of the General Corporation Law of the State of Delaware ("Section 203"), solely by virtue of the Steel Holdings Investment and the acquisitions of securities permitted to be made by HNH and its affiliates under the Settlement Agreement, for so long as HNH and its affiliates and associates continue to collectively own at least fifteen percent of the Company's outstanding Common Stock. The representations and warranties are included in the Investment Agreement are made solely for purposes of the Investment Agreement and may not be accurate or complete as of any particular date because they are subject to a contractual standard of materiality or material adverse effect different from that generally applicable to public disclosure to stockholders or used for the purpose of allocating risk between the parties to the Investment Agreement rather than establishing matters of fact. Stockholders should not rely on these representations and warranties as statements of factual information.

The Investment Agreement also contains representations and warranties by Steel Holdings relating to, among other things, its corporate organization, the due authorization and execution of the Investment Agreement, the shares of Common Stock owned by Steel Holdings and its affiliates, its investment intent and its status as an accredited investor.

Until the closing of the Steel Holdings Investment or the earlier termination of the Investment Agreement, the Company has agreed, without the consent of Steel Holdings, to conduct its business in the ordinary course in a manner consistent with past practice, not to grant any waiver under the Company's Tax Benefit Preservation Plan or Shareholder Rights Plan and not to issue or sell any equity interests, other than in accordance with the Company's Tax Benefit Preservation Plan, Shareholder Rights Plan and equity incentive plans that are currently effective.

The sale of Common Stock and the Warrants pursuant to the Investment Agreement is intended to be exempt from the registration requirements of the Securities Act of 1933, as amended (the "Securities Act"), and the Company expects to rely upon Section 4(2) of the Securities Act for an exemption from registration. Upon the closing of the Steel Holdings Investment, the Company has agreed to reimburse Steel Holdings' reasonably documented out-of-pocket expenses up to a maximum of \$200,000.

Competing Proposals

The Investment Agreement restricts the ability of the Board to directly or indirectly solicit, initiate, facilitate or engage in discussions with a third party regarding an a Competing Proposal (as defined below). However, in the event that the Board receives an unsolicited Competing Proposal from a third party that, prior to receipt of stockholder approval, the Board determines in good faith, after consultation with the Company's legal and financial advisors, constitutes or could reasonably be expected to result in a Superior Proposal (as defined below), and that the

failure to take the following actions would be inconsistent with the proper exercise of its fiduciary duties under applicable law, the Board may furnish information to the third-party and engage in discussions or negotiations regarding such Competing Proposal with the third party, subject to specified conditions. If the Board concludes in good faith, after consultation with the Company's legal and financial advisors, that a Competing Proposal constitutes a Superior Proposal and that the failure to take the following actions would be inconsistent with the proper exercise of its fiduciary duties under applicable law, the Board may, prior to obtaining stockholder approval, withdraw its recommendation of the Steel Holdings Investment or accept the Superior Proposal, subject to specified conditions, including the ability of Steel Holdings to propose revisions to the terms of the Steel Holdings Investment. The Board may also withdraw its approval or recommendation of the Steel Holdings Investment prior to obtaining stockholder approval, subject to certain conditions, if the Board first determines in good faith, after consulting with outside counsel, that the failure to take such action would be inconsistent with the proper exercise of its fiduciary duties under applicable law.

Under the Investment Agreement, a "Superior Proposal" means a written Competing Proposal made by any person on terms that the Board determines in good faith, after consultation with the Company's financial and legal advisors, and considering all relevant legal, financial, regulatory and other aspects of the proposal and the person making the proposal, are more favorable to the Company and its stockholders than the transactions contemplated by the Investment Agreement (considering any changes thereto agreed in writing by Steel Holdings in response thereto) and which the Board determines in good faith is reasonably likely to be consummated. A "Competing Proposal" is defined as any of the following transactions with a person other than Steel Holdings or any of its affiliates, or any inquiry, proposal or offer relating to, or which could reasonably be expected to lead to, any such transaction, directly or indirectly, whether in one transaction or a series of transactions: (i) any merger, consolidation, tender offer, exchange offer, recapitalization, share exchange or similar business combination involving the Company or any of its subsidiaries, (ii) any sale, lease, transfer or other disposition of a material portion of the assets of the Company or any of its subsidiaries, (iii) any issuance of, or transaction involving the sale or transfer of, the Company's equity interests to non-employees of the Company representing, or exercisable or exchangeable for or convertible into securities representing, 5% or more of the voting power of the Company, or (iv) any other transaction directly or indirectly having the purpose or effect of any of the foregoing.

Conditions and Termination

Steel Holdings' obligation to close the Steel Holdings Investment is subject to certain conditions, including the election of both Mr. Lichtenstein and Mr. Kassan to the Board, the approval of the Steel Holdings Investment by the Company's stockholders, that no Company Material Adverse Effect (as defined in the Investment Agreement) shall have occurred and other customary closing conditions.

The Company's obligation to close the Steel Holdings Investment is subject to certain conditions, including the approval of the Steel Holdings Investment by the Company's stockholders and other customary closing conditions.

The Investment Agreement may be terminated, among other reasons, by Steel Holdings if the Board changes its recommendation to the Company's stockholders regarding the Steel Holdings Investment or recommends a Competing Proposal, or if either Mr. Lichtenstein or Mr. Kassan is not elected to the Board at the 2012 Meeting, and may be terminated by the Company if the Board has received a Superior Proposal and determined to accept it prior to receiving stockholder approval of the Steel Holdings Investment. The Investment Agreement may be terminated by either party if stockholder approval of the Steel Holdings Investment is not obtained. In the event of the termination of the Investment Agreement due to any of the foregoing reasons, the Company will be required to pay to Steel Holdings a termination fee equal to \$900,000, plus an additional amount equal to Steel Holdings' reasonably documented out-of-pocket expenses up to a maximum of \$200,000. The Investment Agreement may also be terminated by either party if the closing of the Steel Holdings Investment has not occurred prior to May 6, 2013, subject to extension in the event specified regulatory approvals are required and not obtained.

Registration Rights

Pursuant to the Investment Agreement, the Company agreed to grant Steel Holdings certain registration rights. The Company agreed to file a resale registration statement on Form S-3 as soon as practicable after it is eligible to do so,

covering the shares of Common Stock purchased by Steel Holdings and the shares of Common Stock issuable upon exercise of the Warrants. The Company is required to keep the resale registration statement effective for three years following the date it is declared effective. Steel Holdings also has the right, until such time as it owns less than one-third of the Common Stock originally issued to it under the Investment Agreement, to require that the Company file a prospectus supplement or amendment to cover sales of Common Stock through a firm commitment underwritten public offering. The underwriters of any underwritten offering have the right to limit the number of shares to be included in any such offering. In addition, the Company has agreed to certain "piggyback registration rights." If the Company registers any securities for public sale, Steel Holdings has the right to include its shares in the registration, subject to certain exceptions. The underwriters of any underwritten offering have the right to limit the number of Steel Holdings' shares to be included in any such offering for marketing reasons. The Company has agreed to pay the expenses of Steel Holdings in connection with any registration of the securities issued in the Steel Holdings Investment and to provide customary indemnification to Steel Holdings in connection with such registration.

Warrants

The per share exercise price of the Warrants is \$5.00. The Warrants will be exercisable (including on a cashless basis) beginning on the original date of issuance and will expire on the date that is five years following the closing of the Steel Holdings Investment. The terms of the Warrants provide for customary adjustments to the number of shares issuable upon exercise of the Warrants and the exercise price of the Warrants in the event of a stock split, capital stock dividend, subdivision or other similar event.

The foregoing summary is not a complete description of the terms of the Investment Agreement and is qualified in its entirety by reference to the full text of the Investment Agreement, a copy of which is attached as Exhibit 10.1 hereto and is incorporated by reference herein. The Company's press release, dated February 11, 2013, announcing entry into the Settlement Agreement and the Investment Agreement is attached hereto as Exhibit 99.1.

Settlement Agreement

In connection with the Investment Agreement, the Company entered into an agreement (the "Settlement Agreement") with HNH and certain of HNH's affiliates party thereto, pursuant to which HNH agreed to withdraw its notice of nomination of two candidates to be elected as directors at the Company's 2012 annual meeting of stockholders (the "2012 Annual Meeting") and its previously filed proxy statement with respect to the 2012 Annual Meeting, and the Company agreed to nominate Warren G. Lichtenstein and Glen M. Kassan for election to the Board at the 2012 Annual Meeting.

The Settlement Agreement also provides that:

- HNH has agreed to vote its shares of the Company's common stock in favor of the directors nominated by the Board for election at the 2012 Annual Meeting and each other matter recommended for stockholder approval by the Board and to certain customary restrictions during the Settlement Agreement Period (as defined below).
- If Mr. Lichtenstein is elected to the Board and the Steel Holdings Investment is approved by the Company's stockholders, following the consummation of the Steel Holdings Investment, Mr. Lichtenstein will be appointed as Chairman of the Board.
- The Company has agreed not to increase the size of the Board to more than 7 directors at any time from the date of the Settlement Agreement until ten days preceding the opening of the advance notice period for nominations of directors for the Company's 2013 annual meeting of stockholders (the "Settlement Agreement Period").
- The Company has agreed, subject to consummation of the Steel Holdings Investment, that Edward E. Lucente and Joseph M. O'Donnell will resign from the Board in connection with the closing of the Steel Holdings Investment and Francis J. Jules and Mardy J. Mardy will be appointed to fill the vacancies created thereby.
- The Company agreed, subject to consummation of the Steel Holdings Investment, to grant a limited waiver under Section 203 of the Delaware General Corporation Law in respect of the shares of Common Stock and Warrants (and shares of Common Stock underlying the Warrants) purchased pursuant to the Investment Agreement, as well as an additional 1,405,374 shares of Common Stock (subject to proportionate adjustment as provided in the Settlement Agreement) that HNH and its affiliates may purchase in the open market. The Company also agreed, subject to consummation of the Steel Holdings Investment, to grant HNH a waiver under its Tax Benefit Preservation Plan and to amend its Shareholder Rights Plan to permit the purchase of such securities.
- HNH has agreed that for a period of two years following the date of the Settlement Agreement, HNH and its affiliates ownership in the Company
 may not exceed 30% of the Company's outstanding Common Stock in the aggregate (subject to proportionate adjustment as provided in the
 Settlement Agreement), except for any acquisitions of Common Stock in connection with the exercise of the warrants issued in connection with the
 Steel Holdings Investment.

- The Company agreed to include a proposal in its proxy statement in respect of the 2012 Annual Meeting to amend the Company's restated certificate
 of incorporation providing for the declassification of the Board and the annual election of all directors beginning at the Company's 2013 annual
 meeting of stockholders.
- HNH has agreed that, if elected to the Board, each of Mr. Kassan and Mr. Lichtenstein will be required to resign from the Board in the event that a court of competent jurisdiction determines that HNH or any of its affiliates, including Steel Holdings, has breached its obligations to consummate the Steel Holdings Investment. In addition Mr. Kassan, if elected to the Board, will be required to resign on the date upon which HNH and its affiliates have sold (other than to an affiliate) in excess of 4,900,000 shares of Common Stock in the aggregate (subject to proportionate adjustment as provided in the Settlement Agreement) and Mr. Lichtenstein, if elected to the Board, will be required to resign on the date upon which HNH and its affiliates have sold (other than to an affiliate) in excess of 9,100,000 shares of Common Stock in the aggregate (subject to proportionate adjustment as provided in the Settlement Agreement).
- The Settlement Agreement may be terminated, among other reasons, by the Company or HNH if the Investment Agreement is terminated because the Board has effected a change of recommendation or if the Company has terminated the Investment Agreement following the receipt of a Superior Proposal (as defined in the Investment Agreement) from a third party prior to receiving stockholder approval of the Steel Holdings Investment. The Settlement Agreement may also be terminated by the Company in the event the Investment Agreement is terminated because of an uncured material breach by Steel Holdings or its affiliates. Further, the Settlement Agreement may be terminated by the Company or HNH if Mr. Lichtenstein and Mr. Kassan are not elected to the Board at the 2012 Meeting and the Steel Holdings Investment is not consummated.

The foregoing summary is not a complete description of the terms of the Settlement Agreement and is qualified in its entirety by reference to the full text of the Settlement Agreement, a copy of which is attached as Exhibit 10.2 hereto and is incorporated by reference herein.

Amendment to Rights Agreement

The information set forth under "Item 3.03 Material Modification to Rights of Security Holders" of this Current Report on Form 8-K with respect to the amendment to the Company's Rights Agreement is incorporated into this Item 1.01 by reference.

Consulting Agreement

In June of 2012, following the resignation of the Company's chief executive officer, the Board requested that Mr. Lucente act as a Board resource to management and provide greater board oversight of management by spending additional time at the Company's headquarters in Waltham, Massachusetts. In light of this experience and his knowledge of the Company's current operational plans, the Board has determined, in the event that Mr. Lucente's resignation as a member of the Board described below under Item 5.02 becomes effective, to retain Mr. Lucente as a consultant to assist the Board and the Company's management team, including the Company's new executive leadership, pursuant to a one-year Consulting Agreement. Mr. Lucente would receive a fee of \$175,000 for his services under the Consulting Agreement. The foregoing summary is not a complete description of the terms of the Consulting Agreement and is qualified in its entirety by reference to the full text of the Consulting Agreement, a copy of which is attached as Exhibit 10.3 hereto and is incorporated by reference herein.

Item 3.03. Material Modification to Rights of Security Holders

Amendment to Rights Agreement

On March 21, 2012, the Board adopted the Rights Agreement between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent (the "Agreement"). On February 11, 2013, the Board approved, and the Company and the Rights Agent entered into, an Amendment No. 1 to the Agreement (the "Amendment"). The purpose of the Amendment is to clarify that (a) neither a Distribution Date (under and as defined in the Agreement) nor a Trigger Event (under and as defined in the Agreement) shall be deemed to have occurred, (b) neither Steel

Holdings nor any of its affiliates shall be deemed an Acquiring Person (under and as defined in the Agreement) and (c) no Person shall be entitled to exercise any Rights (under and as defined in the Agreement), in each case, by reason of the entry into or the consummation of the transactions contemplated by (i) the Investment Agreement (ii) the Settlement Agreement.

The foregoing summary is not a complete description of the Amendment and is qualified in its entirety by reference to the full text of the Amendment, a copy of which is attached as Exhibit 4.1 hereto and is incorporated by reference herein. A copy of the Agreement and a summary of its material terms were filed with the Securities and Exchange Commission on Form 8-K on March 22, 2012 (incorporated herein by reference to Exhibit 4.2).

Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers

Pursuant to the terms of the Settlement Agreement, on February 11, 2013, Edward E. Lucente and Joseph M. O'Donnell agreed to resign as members of the Board effective upon the consummation of the Steel Holdings Investment. Pursuant to Section 2(a)(iii) of the Settlement Agreement, subject to and effective immediately following the consummation of the Steel Holdings Investment, Mr. Lucente and Mr. O'Donnell will resign as members of the Board and the Board will appoint Francis J. Jules and Michael J. Mardy to fill the vacancies created by such resignations.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit

No.	<u>Description</u>		
4.1	Amendment No. 1, dated as of February 11, 2013 to Rights Agreement, dated as of March 21, 2012, between ModusLink Global Solutions, Inc. and American Stock Transfer & Trust Company, LLC		
4.2	Rights Agreement, dated as of March 21, 2012, between ModusLink Global Solutions, Inc. and American Stock Transfer & Trust Company, LLC, which includes the Form of Certificate of Designations of Series B Junior Participating Preferred Stock as Exhibit A, the Form of Right Certificate as Exhibit B and the Summary of Rights to Purchase Preferred Shares as Exhibit C. (incorporated herein by reference to Exhibit 4.1 of the registrant's Current Report on Form 8-K, filed with the Securities and Exchange Commission on March 22, 2012).		
10.1	Investment Agreement, dated February 11, 2013, between ModusLink Global Solutions, Inc. and Steel Partners Holdings, L.P.		
10.2	Settlement Agreement, dated February 11, 2013, among ModusLink Global Solutions, Inc., Handy & Harman, Ltd. and certain of its affiliates party thereto		
10.3	Consulting Agreement, dated February 11, 2013, between Edward E. Lucente and ModusLink Global Solutions, Inc.		
99.1	Press Release dated February 11, 2013		

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

MODUSLINK GLOBAL SOLUTIONS, INC.

Date: February 13, 2013

By: /s/ Peter L. Gray

Name: Peter L. Gray

Title: Executive Vice President, Chief Administrative Officer, General

Counsel and Secretary

EXHIBIT INDEX

Exhibit No.	<u>Description</u>
4.1	Amendment No. 1, dated as of February 11, 2013 to Rights Agreement, dated as of March 21, 2012, between ModusLink Global Solutions, Inc. and American Stock Transfer & Trust Company, LLC
4.2	Rights Agreement, dated as of March 21, 2012, between ModusLink Global Solutions, Inc. and American Stock Transfer & Trust Company, LLC, which includes the Form of Certificate of Designations of Series B Junior Participating Preferred Stock as Exhibit A, the Form of Right Certificate as Exhibit B and the Summary of Rights to Purchase Preferred Shares as Exhibit C. (incorporated herein by reference to Exhibit 4.1 of the registrant's Current Report on Form 8-K, filed with the Securities and Exchange Commission on March 22, 2012).
10.1	Investment Agreement, dated February 11, 2013, between ModusLink Global Solutions, Inc. and Steel Partners Holdings, L.P.
10.2	Settlement Agreement, dated February 11, 2013, among ModusLink Global Solutions, Inc., Handy & Harman, Ltd. and certain of its affiliates party thereto
10.3	Consulting Agreement, dated February 11, 2013, between Edward E. Lucente and ModusLink Global Solutions, Inc.
99.1	Press Release dated February 11, 2013

AMENDMENT NO. 1 TO RIGHTS AGREEMENT

THIS AMENDMENT NO. 1 TO RIGHTS AGREEMENT (this "Amendment"), dated as of February 11, 2013, is entered into by and between ModusLink Global Solutions, Inc., a Delaware corporation (the "Company") and American Stock Transfer & Trust Company, LLC, a New York limited liability trust company, as Rights Agent (the "Rights Agent").

WITNESSETH:

WHEREAS, the Company and the Rights Agent entered into that certain Rights Agreement, dated as of March 21, 2012 (as amended, supplemented or otherwise modified, the "Agreement"); and

WHEREAS, the Company and the Rights Agent wish to amend the Agreement as more specifically provided herein.

NOW, THEREFORE, in consideration of the premises and the mutual agreements herein set forth, the parties hereby agree as follows:

- 1. Capitalized Terms. Unless otherwise defined herein, all capitalized terms used in this Amendment shall have the meanings set forth in the Agreement.
- 2. Amendment to the Agreement.

The Agreement is hereby amended by adding a new Section 36 thereof which shall read as follows:

"Section 36. Exception for Investment Agreement and Settlement Agreement. Notwithstanding any provision of this Agreement to the contrary, neither a Distribution Date nor a Trigger Event shall be deemed to have occurred, and none of the Investors (as defined below) or any of their Affiliates or Associates shall be deemed to have become an Acquiring Person, and no holder of any Rights shall be entitled to exercise such Rights under, or be entitled to any rights pursuant to, any of Sections 3, 7, 11 or 13 of this Agreement, in any such case by reason of (a) the approval, execution or delivery of (i) the Investment Agreement (the "Investment Agreement"), dated as of February 10, 2013, by and between the Company and Steel Partners Holdings L.P., a Delaware limited partnership and (ii) the Settlement Agreement (the "Settlement Agreement"), dated as of February 10, 2013, by and between the Company and Handy & Harman Ltd., a Delaware corporation ("HNH" and, together with the parties listed on Exhibit A thereto, the "Investors") or (b) the consummation of any of the transactions contemplated or permitted by the Investment Agreement and Settlement Agreement, pursuant to and in accordance with the terms of such agreements, in each case prior to the termination of any such agreements."

3. Miscellaneous.

- (a) *Governing Law; Jurisdiction*. This Amendment shall be deemed to be a contract made under the internal laws of the State of Delaware and for all purposes shall be governed by and construed in accordance with the laws of such State applicable to contracts to be made and performed entirely within such State.
- (b) *Counterparts; Electronic Signatures*. This Amendment may be executed in any number of counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument. A signature to this Amendment transmitted electronically shall have the same authority, effect and enforceability as an original signature.
- (c) *References*. Any reference to the Agreement contained in any agreement, instrument, notice, request, certificate, or other document executed concurrently with or after the execution and delivery of this Amendment shall be deemed to include this Amendment unless the context shall otherwise require.
- (e) *Confirmation of the Agreement*. Except as expressly modified hereby, the Agreement shall remain in full force and effect and is hereby ratified and confirmed in all respects.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed, as of the day and year first above written.

MODUSLINK GLOBAL SOLUTIONS, INC.

By /s/ Peter L. Gray

Name: Peter L. Gray

Title: Executive Vice President, Chief Administrative Officer, General Counsel and Secretary

AMERICAN STOCK TRANSFER & TRUST COMPANY, LLC

By /s/ Michael A. Nespoli

Name: Michael A. Nespoli Title: Senior Vice President

[Signature Page to Amendment No. 1 to Rights Agreement]

INVESTMENT AGREEMENT

by and between

MODUSLINK GLOBAL SOLUTIONS, INC.

and

STEEL PARTNERS HOLDINGS L.P.

Dated as of February 11, 2013

TABLE OF CONTENTS

		Page
ARTICLE	1 DEFINITIONS	2
ARTICLE	2 PURCHASE AND SALE OF SHARES	8
2.1	Purchase and Sale of the Shares	8
2.2	Purchase Price	8
2.3	Closing	8
2.4	Closing Deliverables	8
ARTICLE	3 REPRESENTATIONS AND WARRANTIES OF THE COMPANY	9
3.1	Organization and Qualifications; Subsidiaries	9
3.2	Due Authorization	10
3.3	Due Execution; Enforceability	11
3.4	Board of Directors	11
3.5	Due Issuance and Authorization of Capital Stock	11
3.6	Capitalization	12
3.7	Consents	13
3.8	No Conflicts	13
3.9	Compliance with Laws and Regulations	14
3.10	SEC Filings	14
3.11	Financial Statements	14
3.12	Litigation	15
3.13	Taxes	15
3.14	Broker Fees	16
3.15	Insolvency, Creditors	16
3.16	Information in the Proxy Statement	17
3.17	Required Vote	17
3.18	Sarbanes-Oxley; Internal Accounting Controls	17
3.19	Private Placement	18
3.20	No Other Representations or Warranties	18
ARTICLE 4	4 REPRESENTATIONS AND WARRANTIES OF THE INVESTOR	18
4.1	Organization	18
4.2	Due Authorization	18
4.3	Due Execution; Enforceability	18
4.4	No Registration Under the Securities Act	18
4.5	Shares of Common Stock Beneficially Owned	19
4.6	Acquisition for Investment	19
4.7	Evaluation of Merits and Risks of Investment	19
4.8	Additional Information	19
49	IIS Person	19

20

4.10

Broker Fees

4.12	Information in the Proxy Statement	20
4.13	No Other Representations or Warranties	20
ARTICLE 5	S ADDITIONAL COVENANTS	20
5.1	No Solicitation of Transactions	20
5.2	Legend	22
5.3	Preparation of Proxy Statement; Stockholder Meeting	22
5.4	Conduct of Business	23
5.5	Further Assurances	24
5.6	Indemnification	24
5.7	Fees	26
5.8	Integration	26
5.9	Reservation of Common Stock	26
5.10	Governmental Approvals	26
ARTICLE 6	CONDITIONS TO CLOSING	27
6.1	Conditions to Investor's Obligations	27
6.2	Conditions to Company's Obligations	28
ADTICLE 7	7 TERMINATION	29
7.1	Termination	29
7.2	Effect of Termination	30
7.3	Termination Fee and Investor Expenses	31
ARTICLE 8 MISCELLANEOUS		31
8.1	Notices	31
8.2	Survival of Representations and Warranties, etc.	33
8.3	Assignment	33
8.4	Entire Agreement	33
8.5	Waivers and Amendments	33
8.6	Governing Law; Jurisdiction; Venue; Process; Jury Trial	33
8.7	Remedies	34
8.8	Counterparts	34
8.9	Headings	34
8.10	Severability	35
Exhibit A	Form of Registration Rights Agreement	
Exhibit B	Form of Warrant	

20

4.11

Sufficient Funds

INVESTMENT AGREEMENT

THIS INVESTMENT AGREEMENT, dated as of February 11, 2013 (this "<u>Agreement</u>"), by and between ModusLink Global Solutions, Inc., a Delaware corporation (the "<u>Company</u>"), and Steel Partners Holdings L.P., a Delaware limited partnership (the "<u>Investor</u>"). Each of the Company and the Investor is referred to herein as a "<u>Party</u>" and collectively as the "<u>Parties</u>." All capitalized terms used in this Agreement shall have the meanings assigned to such terms in <u>Article 1</u> or as otherwise defined elsewhere in this Agreement unless the context clearly provides otherwise.

WITNESSETH:

WHEREAS, the Investor wishes to purchase from the Company, and the Company wishes to sell and issue to the Investor, upon the terms and subject to the conditions set forth in this Agreement, an aggregate of seven million five hundred thousand (7,500,000) newly issued shares of the common stock of the Company, par value \$0.01 per share (the "Common Stock") (as adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction after the date hereof and prior to the Closing), along with the associated right to acquire Series A Preferred Stock and Series B Preferred Stock issued under the Tax Plan and the Rights Plan, respectively (collectively, the "Shares"), at a per-share purchase price of \$4.00 (as adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction after the date hereof and prior to the Closing), together with warrants to purchase two million (2,000,0000) shares of Common Stock (as adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction after the date hereof and prior to the Closing, and subject to adjustment as provided therein) at an exercise price of \$5.00 per share (as adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction after the date hereof and prior to the Closing, and subject to adjustment as provided therein) in the form of Exhibit B hereto (the "Warrants"), for an aggregate purchase price of thirty million dollars (\$30,000,000);

WHEREAS, with respect to the purchase and sale of the Shares and the Warrants, and the shares of Common Stock issuable upon exercise of the Warrants (the "<u>Underlying Shares</u>"), the Company and the Investor are executing and delivering this Agreement in reliance upon the exemption from securities registration afforded by the provisions of Section 4(2) of the Securities Act, and Regulation D, as promulgated by the SEC under the Securities Act; and

WHEREAS, the Board of Directors of the Company (the "Board") has determined that this Agreement, the issuance of Common Stock, the Warrants and the Underlying Shares, and the other transactions contemplated by this Agreement are advisable and in the best interests of the Company and the Stockholders; and

WHEREAS, in connection with the Stockholder Meeting, the Company intends to seek the Stockholders' approval of the issuance of the Shares, the Warrants and the Underlying Shares to the Investor by a majority of all the votes cast at the Stockholder Meeting, whether in person or by proxy, pursuant to NASDAQ Marketplace Rule 5635(b) (the "Stockholder Approval").

NOW, THEREFORE, in consideration of the mutual covenants and premises contained in this Agreement and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Parties agree as follows:

ARTICLE 1 DEFINITIONS

For purposes of this Agreement, the following terms will have the meaning set forth below:

"Acceptable Confidentiality Agreement" means a confidentiality agreement that contains confidentiality provisions that are no less favorable in any material respect to the Company than those contained in that certain confidentiality agreement, dated June 27, 2012, by and between the Company and Handy & Harman Ltd.

"Affiliate" of any Person means any Person that directly or indirectly controls, or is under common control with, or is controlled by, such Person. As used in this definition, "control" (including with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise).

- "Agreement" has the meaning assigned to it in the Preamble.
- "Board" has the meaning assigned to it in the Recitals.
- "Business Day" has the meaning set forth in Rule 14d-1(g)(3) of the Exchange Act.
- "Change of Recommendation" has the meaning assigned to it in Section 5.1(a).
- "Claim" has the meaning assigned to it in Section 3.15.
- "Closing" has the meaning assigned to it in Section 2.3.
- "Closing Date" has the meaning assigned to it in Section 2.3.
- "Code" means the Internal Revenue Code of 1986, as amended.
- "Common Stock" has the meaning assigned to it in the Recitals.
- "Company" has the meaning assigned to it in the Preamble.
- "Company Bylaws" has the meaning assigned to it in Section 3.1(b).
- "Company Certificate" has the meaning assigned to it in Section 3.1(b).
- "Company Disclosure Schedule" has the meaning assigned to it in Article 3.

"Company Indemnitees" has the meaning assigned to it in Section 5.12.

"Company Material Adverse Effect" means any change, event, development, condition, occurrence or effect that is, or would reasonably be expected to be, materially adverse to the business, financial condition, results of operations, assets, liabilities or prospects of the Company and the Company Subsidiaries, taken as a whole; provided, however, that none of the following shall be deemed in themselves, either alone or in combination, to constitute, and that none of the following shall be taken into account in determining whether there has been or will be, a Company Material Adverse Effect (but, in the case of clauses (a), (b), (d), (e), (f) and (g) below, only to the extent the Company and the Company Subsidiaries are not materially and disproportionately affected thereby): (a) any change generally affecting the economy, financial markets or political, economic or regulatory conditions in the United States or any other geographic region in which the Company and the Company Subsidiaries conduct business; (b) general financial, credit or capital market conditions, including interest rates or exchange rates, or any changes therein; (c) the taking of any action expressly provided by this Agreement or to which Investor has consented in writing; (d) acts of war (whether or not declared), the commencement, continuation or escalation of a war, acts of armed hostility, sabotage or terrorism or other international or national calamity or any material worsening of such conditions existing as of the date of this Agreement; (e) any hurricane, earthquake, flood, or other natural disasters or acts of God; (f) changes in Laws after the date hereof; (g) changes in GAAP after the date hereof; (h) any failure by the Company to meet any published or internally prepared estimates of revenues, earnings or other economic performance for any period ending on or after the date of this Agreement and prior to the Closing (it being understood that the facts and circumstances giving rise to such failure may be deemed to constitute, and may be taken into account in determining whether there has been, a Company Material Adverse Effect if such facts and circumstances are not otherwise described in clauses (a)-(g) of the definition); or (i) a decline in the price of the Common Stock on NASDAQ or any other market in which such securities are quoted for purchase and sale (it being understood that the facts and circumstances giving rise to such decline may be deemed to constitute, and may be taken into account in determining whether there has been, a Company Material Adverse Effect if such facts and circumstances are not otherwise described in clauses (a)-(g) of the definition). For the avoidance of doubt, a Company Material Adverse Effect shall be deemed to have occurred if any client with annual revenue in excess of \$70 million for the fiscal year ended July 31, 2012 terminates or reduces its level of business, or any material program, with the reasonably likely effect of a reduction of anticipated revenues for the next twelve (12) months from such client, or material program, by more than fifty percent (50%).

"Company SEC Documents" has the meaning assigned to it in Section 3.10.

"Company Subsidiary" has the meaning assigned to it in Section 3.1(a).

"Competing Proposal" means any of the following transactions with a person other than Investor or any of its Affiliates, or any inquiry, proposal or offer relating to, or which could reasonably be expected to lead to, any such transaction, directly or indirectly, whether in one transaction or a series of transactions: (i) any merger, consolidation, tender offer, exchange offer, recapitalization, share exchange or similar business combination involving the Company or any of its Subsidiaries, (ii) any sale, lease, transfer or other disposition of a material portion of

the assets of the Company or any of its Subsidiaries, (iii) any issuance of, or transaction involving the sale or transfer of, the Company's Equity Interests to non-employees of the Company representing, or exercisable or exchangeable for or convertible into securities representing, 5% or more of the voting power of the Company, or (iv) any other transaction directly or indirectly having the purpose or effect of any of the foregoing. For the avoidance of doubt, any transaction of a type described in the section entitled "Background to the Other Solicitations" contained in the Company's preliminary proxy statement filed with the SEC on December 21, 2012, as amended, shall be deemed to be a Competing Proposal.

"Contract" means any of the agreements, arrangements, commitments, understandings, contracts, leases (whether for real or personal property), powers of attorney, notes, bonds, mortgages, indentures, deeds of trust, loans, evidences of indebtedness, purchase orders, letters of credit, settlement agreements, franchise agreements, undertakings, covenants not to compete, employment agreements, licenses, instruments, obligations, commitments, understandings, policies, purchase and sales orders, quotations and other commitments to which a Person is a party or to which any of the assets of such Person or any of its Subsidiaries are subject, whether oral or written, express or implied.

"Debt" has the meaning assigned to it in Section 3.15.

"DGCL" means Delaware General Corporation Law, as amended.

"dollars" and "\$" mean dollars in lawful currency of the United States of America.

"Equity Interest" means any share of capital stock, partnership, member or similar equity interests in any Person, and any securities (including debt securities) convertible into, or exchangeable or exercisable for, any such shares, capital stock, partnership, member or similar equity interests, or any options, warrants or other rights of any kind to acquire or that are linked to the value of any such shares, capital stock, partnership, member or similar equity interests or such convertible or exchangeable securities, or any other ownership interest (including, without limitation, any such interest represented by Contract right), of such Person.

"ESPP" means the Amended and Restated 1995 Employee Stock Purchase Plan.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Financial Statements" has the meaning assigned to it in Section 3.11.

"GAAP" means generally accepted accounting principles as applied in the United States.

"Governmental Entity" means any nation, federal, state, county, municipal, local or foreign government, or other political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory, taxing or administrative functions of or pertaining to government.

- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations thereunder.
- "Incentive Plans" means any stock option plan of the Company, including the 1999 Stock Option Plan for Non-Employee Directors, 2000 Stock Incentive Plan, 2002 Non-Officer Employee Stock Incentive Plan, 2004 Stock Incentive Plan, 2005 Non-Employee Director Plan and the 2010 Incentive Award Plan or any other plan, agreement or arrangement.
 - "Indemnified Liabilities" has the meaning assigned to it in Section 5.6(a).
 - "Indemnitees" has the meaning assigned to it in Section 5.6(a).
 - "Investor" has the meaning assigned to it in the Preamble.
- "Investor Expenses" means up to \$200,000 of the Investor's reasonably documented out-of-pocket fees and expenses (including legal expenses) incurred in connection with matters related to any of the following: (i) the Investors' and/or any of its Affiliates' investment in the Company, (ii) the Stockholder Meeting and (iii) the negotiation and execution of this Agreement, the Settlement Agreement, the Registration Rights Agreement and the Warrants.
- "knowledge" as to the Company means the actual knowledge of Steven G. Crane, Peter L. Gray or Scott R. Crawley and as to the Investor means the actual knowledge of Warren G. Lichtenstein, Glen M. Kassan and Leonard McGill.
- "<u>Law</u>" means any federal, state, local or foreign law, statute, code, ordinance, rule, regulation, order, judgment, writ, stipulation, award, injunction, decree or other pronouncement of any Governmental Entity having the effect of law.
- "<u>Lien</u>" means any lien, mortgage, pledge, conditional or installment sale agreement, encumbrance, covenant, condition, restriction, charge, option, right of first refusal, easement, security interest, deed of trust, right-of-way, encroachment, community property interest or other claim or restriction of any nature, whether voluntarily incurred or arising by operation of Law (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).
 - "Matching Bid" has the meaning assigned to it in Section 5.1(b).
 - "NASDAQ" means the NASDAQ Global Select Market.
 - "Offer" has the meaning assigned to it in Section 5.1(b).
 - "Offer Notice" has the meaning assigned to it in Section 5.1(b).
- "Options" means any unexpired and unexercised option to purchase shares of Common Stock or Preferred Stock under any Incentive Plan or otherwise.

- "Organizational Documents" means the certificate of incorporation and bylaws, or equivalent organizational or governing documents, each as amended to date, of a Person.
 - "Party" or "Parties" has the meaning assigned to it in the Preamble.
- "Permitted Liens" means (a) Liens for Taxes not yet due and payable or that are being contested in good faith by appropriate proceedings and for which adequate reserves have been established in accordance with GAAP, (b) Liens in favor of vendors, carriers, warehousemen, repairmen, mechanics, workmen, materialmen, landlords construction or similar liens or other encumbrances arising by operation of Law in the ordinary course of business and (c) Liens that do not materially interfere with or detract from the use or value of any material property of the Company or any Company Subsidiary.
- "Person" means an individual, corporation, limited liability company, partnership, association, trust, unincorporated organization, other entity or group (as defined in Section 13(d) of the Exchange Act).
 - "Preferred Stock" means the preferred stock, par value \$0.01 per share, of the Company.
- "Proxy Statement" means the preliminary proxy statement filed, relating to the Stockholder Meeting, as amended, including the definitive proxy statement relating thereto.
 - "Purchase Price" has the meaning assigned to it in Section 2.2.
 - "Rights Agreements" means the Rights Plan and the Tax Plan.
 - "Registration Statement" has the meaning assigned to it in Section 2.3(a).
- "Rights Plan" means the Rights Agreement, dated as of March 21, 2012, between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent.
 - $\label{eq:control_equation} \begin{tabular}{ll} \tt "Rights Plan Rights" means the associated preferred share purchase rights issued pursuant to the Rights Plan. \\ \end{tabular}$
 - "Sarbanes-Oxley Act" has the meaning assigned to it in Section 3.10.
 - "SEC" means the Securities and Exchange Commission.
 - "Section 203 Approval" has the meaning assigned to it in Section 3.2(c).
 - "Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.
- "Settlement Agreement" means that certain Settlement Agreement, dated as of the date hereof, by and between the Company and Handy & Harman Ltd.
 - "Series A Preferred Stock" has the meaning assigned to it in Section 3.6(b)(i).

- "Series B Preferred Stock" has the meaning assigned to it in Section 3.6(b)(ii).
- "Shares" has the meaning assigned to it in the Recitals.
- "Solvent" has the meaning assigned to it in Section 3.15.
- "Stockholder Meeting" means the 2012 annual meeting of the Stockholders.
- "Stockholders" means the stockholders of the Company.
- "Subsidiary" of the Company, the Investor or any other Person means any corporation, partnership, joint venture or other legal entity of which the Company, the Investor or such other Person, as the case may be (either alone or through or together with any other Subsidiary), owns, directly or indirectly, a majority of the stock or other equity interests the holders of which are generally entitled to vote for the election of the board of directors or other governing body of such corporation, partnership, joint venture or other legal entity, or any Person that would otherwise be deemed a "subsidiary" under Rule 12b-2 promulgated under the Exchange Act.

"Superior Proposal" means a written Competing Proposal made by any Person on terms that the Board determines in good faith, after consultation with the Company's financial and legal advisors, and considering all relevant legal, financial, regulatory and other aspects of the proposal and the Person making the proposal, are more favorable to the Company and the Stockholders than the transactions contemplated by this Agreement (considering any changes to this Agreement agreed in writing by the Investor in response thereto) and which the Board determines in good faith is reasonably likely to be consummated.

"<u>Tax Plan</u>" means the Tax Benefit Preservation Plan, dated as of October 17, 2011, between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent (as amended by Amendment No. 1 to Tax Benefit Preservation Plan, dated as of March 21, 2012, by and between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent).

"Tax Plan Rights" means the associated preferred share purchase rights issued pursuant to the Tax Plan.

"<u>Tax Return</u>" means any report, return (including information return), claim for refund or declaration required to be supplied to any Governmental Entity with respect to Taxes, including any schedule or attachment thereto, and including any amendments thereof.

"Taxes" means any and all taxes, fees, levies, duties, tariffs, imposts and other similar charges (together with any and all interest, penalties and additions to tax) imposed by any Governmental Entity, including income, franchise, windfall or other profits, gross receipts, property, sales, use, net worth, capital stock, payroll, employment, social security, workers' compensation, unemployment compensation, excise, withholding, ad valorem, stamp, transfer, value-added, gains tax and license, registration and documentation fees.

- "Transfer Agent" means American Stock Transfer & Trust Company, the Company's transfer agent.
- "Underlying Shares" has the meaning assigned to it in the Recitals.
- "Warrants" has the meaning assigned to it in the Recitals.

ARTICLE 2 PURCHASE AND SALE OF SHARES

- 2.1 <u>Purchase and Sale of the Shares</u>. Upon the terms and subject to the conditions of this Agreement, the Company agrees to issue and sell to the Investor, and the Investor agrees to purchase from the Company, on the Closing Date, all of the Shares and the Warrants.
- 2.2 <u>Purchase Price</u>. The per-share purchase price for the Shares shall be \$4.00, for an aggregate purchase price of thirty million dollars (\$30,000,000) (the "Purchase Price") for the Shares and the Warrants.
- 2.3 <u>Closing</u>. The closing of the transactions contemplated by this Agreement (the "<u>Closing</u>") shall take place at 10:00 a.m., New York time, on the second (2nd) Business Day after satisfaction or waiver of all of the conditions set forth in <u>Sections 2.4</u> and <u>2.5</u> (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions at the Closing) (the "<u>Closing Date</u>"), at the offices of Latham & Watkins LLP located at 885 Third Avenue, New York, New York 10022, unless another time, date or place is agreed to in writing by the Parties.

2.4 Closing Deliverables.

- (a) <u>Closing Deliverables of the Investor</u>. The following deliveries shall be made by the Investor to the Company prior to the Closing:
 - (i) a duly executed copy of this Agreement;
- (ii) a duly executed copy of the Registration Rights Agreement substantially in the form of Exhibit A hereto (the "Registration Rights Agreement");
- (iii) the full amount of the Purchase Price for the Shares and the Warrants in immediately available funds by wire transfer to an account or accounts designated by the Company by written notice to the Investor, which written notice shall be delivered not later than two (2) Business Days prior to the Closing Date; and
 - (iv) an officer's certificate as described in Section 6.2(h).
- (b) <u>Closing Deliverables of the Company</u>. The following deliveries shall be made by the Company to the Investor prior to the Closing:
 - (i) a duly executed copy of this Agreement;

- (ii) a duly executed copy of the Registration Rights Agreement;
- (iii) (x) evidence of one or more stock certificates representing the Shares registered in the name of the Investor or (y) upon the request of the Investor, credit to the Investor's or its designee's balance account with The Depository Trust Company through its Deposit Withdrawal At Custodian system, or to the Investor's or its designee's direct registration account;
 - (iv) the Warrants, duly executed by the Company in favor of the Investor;
 - (v) a certificate from the Transfer Agent certifying the number of issued and outstanding shares of Common Stock of the Company;
- (vi) payment of the Investor Expenses in immediately available funds by wire transfer to an account or accounts designated by the Investor by written notice to the Company, which written notice shall be delivered not later than two (2) Business Days prior to the Closing Date: and
 - (vii) an officer's certificate as described in Section 6.1(k).

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in, or qualified by any matters set forth in, (i) any information contained, or incorporated by reference, in any report, schedule, form, statement or other Company SEC Document filed with, or furnished to, the SEC by the Company and publicly available prior to the date hereof (excluding any disclosures set forth in any "risk factor" section thereof or any statements that constitute forward-looking statements in that such statements are predictive or forward-looking in nature) or (ii) the section of the disclosure schedule delivered by the Company to the Investor prior to the execution of this Agreement (the "Company Disclosure Schedule") that corresponds to any Section or Subsection hereof or in any other section of the Company Disclosure Schedule to the extent it is reasonably/readily apparent from the face of such disclosure or the context in which such disclosure is made that such disclosure is applicable to any other Section or Subsection of this Agreement, the Company represents and warrants to the Investor, as of the date hereof, as follows:

3.1 Organization and Qualifications; Subsidiaries.

(a) The Company and each of its Subsidiaries (each a "<u>Company Subsidiary</u>") is a corporation or other legal entity duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization (in such jurisdictions where the concept exists) and has all requisite corporate or organizational, as the case may be, power and authority to own, lease and operate its properties and assets and to carry on its business as it is now being conducted, except where the failure to be so qualified or in good standing, individually or in the aggregate, has not had and would not reasonably be expected to have a Company Material Adverse Effect. The Company and each Company Subsidiary is duly qualified to do business and is in good standing in each jurisdiction where the

ownership, leasing or operation of its properties or assets or the conduct of its business requires such qualification, except where the failure to be so qualified or in good standing, individually or in the aggregate, has not had and would not reasonably be expected to have a Company Material Adverse Effect.

- (b) The Company has delivered or made available to the Investor true and complete copies of the Restated Certificate of Incorporation of the Company (the "<u>Company Certificate</u>") and Third Amended and Restated Bylaws of the Company (the "<u>Company Bylaws</u>"), each as amended to date, and the Organizational Documents of each Company Subsidiary. Each of the foregoing documents is in full force and effect. The Company has not violated any provision of the Company Certificate or the Company Bylaws in a manner that has not been cured and that materially and adversely affects the Company.
- (c) Section 3.1(c) of the Company Disclosure Schedule sets forth a true and complete list of the Company Subsidiaries, together with the jurisdiction of organization or incorporation, as the case may be, of each Company Subsidiary and the percentage ownership thereof by the Company and/or another Company Subsidiary. Each Company Subsidiary is in compliance in all material respects with the terms of its Organizational Documents.

3.2 Due Authorization.

- (a) The Company has the requisite corporate power and authority to enter into, execute and deliver this Agreement, the Warrants and the Registration Rights Agreement and to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereunder and thereunder. The execution and delivery of this Agreement, the Warrants and the Registration Rights Agreement by the Company and the consummation by the Company of the transactions contemplated hereunder and thereunder have been duly and validly authorized by all requisite corporate action, and no other corporate proceedings on the part of the Company, pursuant to the General Corporation Law of the State of Delaware, as amended, or otherwise, are necessary to authorize the execution and delivery of this Agreement, the Warrants and the Registration Rights Agreement or to consummate the transactions contemplated hereunder or thereunder other than the Stockholder Approval.
- (b) The Investor has formally requested and the Board has granted a waiver and exemption to the Investor under the Tax Plan and otherwise modified the Rights Plan effective concurrently with the execution of this Agreement, in each case, to permit the entering into of this Agreement and the Settlement Agreement, and, subject to the consummation of the transactions contemplated hereunder, the acquisition by the Investor and/or any of its Affiliates of the Shares, the Warrants and the Underlying Shares and the acquisition by the Investor and/or any of its Affiliates of additional securities as set forth in the Settlement Agreement. The Rights Agreements have not been further amended or modified, as of the date hereof, and will not be further amended or modified to prohibit or restrict any of the foregoing permitted acquisitions. The Company (i) has not granted any currently effective waiver or exemption to any Person under any Rights Agreement, except as set forth on Section 3.2(b) of the Company Disclosure Schedule, (ii) will not grant any waiver or exemption to any Person under any Rights Agreement prior to the Closing and (iii) will not grant any waiver or exemption to any Person under restrict any of the foregoing permitted acquisitions by the Investor and/or any of its Affiliates.

- (c) The Board has taken the necessary actions such that the restrictions on business combinations contained in DGCL Section 203 will not apply to the Investor and its Affiliates and Associates (as defined in DGCL Section 203) solely by virtue of the transactions contemplated hereby and the acquisitions permitted by Section 5(a) of the Settlement Agreement, and neither the Investor nor any of its Affiliates and Associates shall be deemed to be an "interested stockholder" (as defined in and contemplated by Section 203(c)(5) of the DGCL) solely as a result of the transactions contemplated hereby or the foregoing permitted acquisitions (the "Section 203 Approval"). Notwithstanding the foregoing, from and after such time as the Investor, together with its Affiliates and Associates, no longer has collective beneficial ownership of at least fifteen percent (15%) in voting power of the Company's then-outstanding capital stock by virtue of having made a disposition, the Section 203 Approval shall not be applicable to any subsequent acquisitions of the Company's capital stock by the Investor and its Affiliates and Associates that would result in such Persons having beneficial ownership of fifteen percent (15%) or more in voting power of the Company's capital stock. The Investor hereby acknowledges and agrees that the Section 203 Approval is limited in accordance with the terms of the Board resolutions as set forth above, and is effective only to the extent of such limitation, as set forth above, and that Section 203 (and the restrictions on business combinations contained therein) shall otherwise apply to the Investor and its Affiliates and Associates and shall apply as a matter of contract pursuant to this Agreement.
- 3.3 <u>Due Execution; Enforceability</u>. Each of this Agreement, the Warrants and the Registration Rights Agreement has been duly authorized and validly executed and delivered by the Company and, assuming due authorization, execution and delivery by the Investor, constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws, now or hereafter in effect, affecting creditors' rights generally and general principles of equity.
- 3.4 <u>Board of Directors</u>. The Board has determined that this Agreement, the Warrants and the Registration Rights Agreement and the transactions contemplated hereby and thereby, including the terms of issuance of the Common Stock, Warrants and Underlying Shares, are advisable and in the best interests of the Company and its Stockholders.
- 3.5 <u>Due Issuance and Authorization of Capital Stock</u>. No shares of capital stock of the Company are subject to preemptive rights or any other similar rights of any or all of the Stockholders. The shares of Common Stock and Warrants to be issued and delivered to the Investor pursuant to the terms hereof have been duly authorized and will be, upon issuance, validly issued, fully paid and non-assessable, and will be free from all taxes, Liens, preemptive rights and charges with respect to the issue thereof. The Underlying Shares to be issued and delivered to the Investor pursuant to the terms of the Warrants have been duly authorized and will be, upon issuance, validly issued, fully paid and non-assessable, and will be free from all taxes, Liens, preemptive rights and charges with respect to the issue thereof. The Company has reserved from its duly authorized capital stock solely for the benefit of the Investor and its Affiliates the maximum number of shares of Common Stock issuable pursuant to this Agreement and the Warrants.

3.6 Capitalization.

- (a) The authorized capital stock of the Company consists of (i) 1,400,000,000 shares of Common Stock and (ii) 5,000,000 shares of Preferred Stock. As of the close of business on February 5, 2013, (x) there were 44,021,864 shares of Common Stock issued and outstanding, (y) no shares of Common Stock were issued and held in the treasury of the Company or otherwise owned by the Company, and (z) there were no shares of Preferred Stock issued and outstanding. All of the outstanding shares of Common Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, and have been issued in compliance with all federal and state securities Laws.
- (b) As of the close of business on February 5, 2013, the Company has no shares of Common Stock or Preferred Stock reserved for or otherwise subject to issuance, except for (i) 140,000 shares of Series A Junior Participating Preferred Stock, par value \$0.01 per share, reserved for issuance in connection with the exercise of Tax Plan Rights issued pursuant to the Tax Plan ("Series A Preferred Stock"), (ii) 140,000 shares of Series B Junior Participating Preferred Stock, par value \$0.01 per share, reserved for issuance in connection with the exercise of Rights Plan Rights issued pursuant to the Rights Plan ("Series B Preferred Stock") and (iii) 1,691,388 shares of Common Stock reserved for issuance upon exercise of outstanding options under the Incentive Plans and the ESPP. All shares of Preferred Stock and Common Stock subject to issuance under the Rights Agreements or Incentive Plans, as the case may be, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights.
- (c) As of the date hereof, except for the Tax Plan Rights, Rights Plan Rights, Options and offerings under the terms of the ESPP, there are no (i) options, warrants or other rights relating to any Equity Interests of the Company or (ii) agreements or arrangements obligating the Company to issue, acquire or sell any Equity Interests of the Company.
- (d) There are no outstanding obligations of the Company (i) restricting the transfer of, (ii) affecting the voting rights of, (iii) requiring the repurchase, redemption or disposition of, or containing any right of first refusal with respect to, (iv) requiring the registration for sale of or (v) except for the Rights Plan Rights, granting any preemptive or antidilutive rights with respect to, any shares of Common Stock or other Equity Interests in the Company. The authorization, execution and delivery of this Agreement, the Warrants and the Registration Rights Agreement, the consummation of any of the transactions contemplated herein and therein or the execution or performance of any of the rights or obligations pursuant to the terms herein or therein will not give rise to, trigger or otherwise give the right to any antidilution rights to any holder of Common Stock or securities convertible into, or exercisable or exchangeable for, shares of Common Stock.
- (e) Except for the Company Subsidiaries listed on Section 3.6(e)(i) of the Company Disclosure Schedule, the Company or another Company Subsidiary owns, directly or

indirectly, all of the issued and outstanding shares of capital stock or other Equity Interests of each of the Company Subsidiaries, free and clear of any Liens (other than transfer and other restrictions under applicable federal and state securities Laws), and all of such shares of capital stock or other Equity Interests have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights. Except for Equity Interests in the Company Subsidiaries and the Equity Interests described on Section 3.6(e)(ii) of the Company Disclosure Schedule, neither the Company nor any Company Subsidiary owns directly or indirectly any Equity Interest in any Person, or has any obligation to acquire any such Equity Interest, to provide funds to, or to make any investment (in the form of a loan, capital contribution or otherwise) in, any Company Subsidiary or any other Person. Since the close of business on July 31, 2011, no Company Subsidiary has issued any shares of capital stock or other Equity Interests.

3.7 <u>Consents</u>. None of the execution, delivery or performance of this Agreement, the Warrants or the Registration Rights Agreement, or the consummation of the transactions contemplated hereby or thereby, will require (with or without notice or lapse of time, or both) any consent, approval, authorization or permit of, or filing or registration with or notification to, any Governmental Entity or any other Person, other than (a) the Stockholder Approval, (b) compliance with the applicable requirements of the Exchange Act, including filings with the SEC as may be required by the Company in connection with this Agreement and the transactions contemplated hereunder, including the filing of a Form D with respect to the transactions contemplated hereby and the filing of one or more registration statements, as contemplated pursuant to the Registration Rights Agreement, (c) compliance with any applicable foreign or state securities Laws or Blue Sky Laws, (d) such filings as may be required under the rules and regulations of NASDAQ, (e) compliance with any applicable requirements of the HSR Act and other applicable foreign or supranational antitrust and competition Laws, and (f) such other consents, authorizations, filings, approvals and registrations set forth in Section 3.7 of the Company Disclosure Schedule.

3.8 No Conflicts. The execution, delivery and performance of Agreement, the Warrants and the Registration Rights Agreement and the consummation of the transactions contemplated hereunder and thereunder, do not and will not (a) conflict with or violate any provision of the Company Certificate or Company Bylaws or the Organizational Documents of any Company Subsidiary; (b) assuming that all consents, approvals, authorizations and permits described in Section 3.7 have been obtained and all filings and notifications described in Section 3.7 have been made and any waiting periods thereunder have terminated or expired, conflict with or violate any Law applicable to the Company or any Company Subsidiary or any of their respective properties or assets; or (c) subject to obtaining the Stockholder Approval, result in any breach of, or constitute a default under (with or without notice or lapse of time, or both), or result in termination or give to others any right of termination, vesting, amendment, acceleration or cancellation of, or result in the creation of a Lien upon any of the respective properties or assets of the Company or any Company Subsidiary pursuant to, any Contract, except, with respect to clauses (b) and (c), where such conflicts or violations would not reasonably be expected to have a Company Material Adverse Effect or prevent or materially delay its ability to consummate the transactions contemplated by Agreement, the Warrants and the Registration Rights Agreement. The Company is not in violation of the listing requirements of NASDAQ and has no knowledge of any facts that would reasonably lead to delisting or suspension of the Common Stock in the foreseeable future. Assuming that Stockholder Approval

has been obtained, the issuance by the Company of any of the Shares, the Warrants or the Underlying Shares shall not have the effect of delisting or suspending the Common Stock from NASDAQ.

- 3.9 <u>Compliance with Laws and Regulations</u>. Except as set forth on Section 3.9 of the Company Disclosure Schedule, as of the date hereof, each of the Company and the Company Subsidiaries has complied and is in compliance with all Laws which affect the business, properties or assets of the Company or such Company Subsidiary, except for any non-compliance that, individually or in the aggregate, has not had and would not reasonably be expected to have a Company Material Adverse Effect.
- 3.10 SEC Filings. Except as set forth on Section 3.10 of the Company Disclosure Schedule, since August 1, 2009, the Company has filed or otherwise furnished (as applicable) all registration statements, prospectuses, forms, reports, definitive proxy statements, schedules and documents required to be filed or furnished by it under the Securities Act or the Exchange Act, as the case may be, together with all certifications required pursuant to the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act") (such documents and any other documents filed by the Company or any Company Subsidiary with the SEC, as have been supplemented, modified or amended since the time of filing, collectively, the "Company SEC Documents"). Except as provided on Section 3.10 of the Company Disclosure Schedule, as of their respective filing dates and, if supplemented, modified or amended since the time of filing, as of the date of the most recent supplement, modification or amendment, the Company SEC Documents (a) did not at the time each such document was filed contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading and (b) complied in all material respects with the applicable requirements of the Exchange Act or the Securities Act, as the case may be, and the Sarbanes-Oxley Act. The Company has not filed any confidential material change reports with the SEC that is still maintained on a confidential basis.
- 3.11 Financial Statements. Except as set forth in Section 3.11(a) of the Company Disclosure Schedule, the audited consolidated financial statements for the fiscal year ended July 31, 2012 and the unaudited consolidated interim financial statements for the fiscal quarter ended October 31, 2012 of the Company and the consolidated Company Subsidiaries included in the Company SEC Documents (collectively, the "Financial Statements") (i) have been prepared in accordance with GAAP (as in effect in the United States of the date of such Financial Statement) applied on a consistent basis during the periods involved (except for normal and recurring year-end adjustments that are not expected to be material and as may be permitted by the SEC on Form 10-Q, Form 8-K or any successor or like form under the Exchange Act) and (ii) fairly present in all material respects the consolidated financial position and the consolidated results of operations, cash flows and changes in stockholders' equity of the Company and the consolidated Company Subsidiaries as of the dates and for the periods referred to therein (except for normal and recurring year-end adjustments that are not expected to be material and as may be permitted by the SEC on Form 10-Q, Form 8-K or any successor or like form under the Exchange Act). Except as set forth in Section 3.11(b) of the Company Disclosure Schedule or any liability (contingent or otherwise) that would not be required by GAAP to be disclosed or reflected in a consolidated balance sheet of the Company, neither the Company nor any of the

Company Subsidiaries has incurred any material liabilities (contingent or otherwise) other than liabilities incurred in the ordinary course of business consistent with past practice since October 31, 2012. Since October 31, 2012, there has been no change, event, development, condition, occurrence or effect that has had or would reasonably be expected to have a Company Material Adverse Effect.

3.12 <u>Litigation</u>. As of the date hereof, except as listed on Section 3.12 of the Company Disclosure Schedule, there is no suit, claim, action, arbitration, investigation or proceeding pending or, to the knowledge of the Company, threatened that (i) if determined adversely to the Company or any Company Subsidiary, has had or would reasonably be expected to result in losses greater than one million dollars (\$1,000,000), or (ii) could reasonably be expected to result in a Company Material Adverse Effect. Neither the Company nor any Company Subsidiary is subject to any outstanding order, writ, injunction, judgment, decree or arbitration ruling, award or other finding that, individually or in the aggregate, has had or would reasonably be expected to have a Company Material Adverse Effect.

3.13 Taxes.

- (a) Except as has not had and would not reasonably be expected to have, individually or in the aggregate a materially adverse impact on the business, financial condition, results of operations, assets, liabilities or prospects of the Company and the Company Subsidiaries, taken as a whole: (i) the Company and each Company Subsidiary has timely filed all Tax Returns required to be filed through the date hereof (taking into account any applicable extensions) and (ii) all Taxes of the Company and each Company Subsidiary that are due and payable have been paid, other than Taxes which are being contested in good faith and as to which the Company or the relevant Company Subsidiary has set aside on its books appropriate reserves in accordance with GAAP.
- (b) Except as has not had and would not reasonably be expected to have, individually or in the aggregate a materially adverse impact on the business, financial condition, results of operations, assets, liabilities or prospects of the Company and the Company Subsidiaries, taken as a whole: (i) no Tax Return of the Company or any Company Subsidiary is currently the subject of an audit by any Governmental Entity, and no written notice of such an audit has been received by the Company or any Company Subsidiary with respect to Taxes for any open Tax year; (ii) neither the Company nor any of the Company Subsidiaries has waived any statute of limitations with respect to Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency, in each case, for any open Tax year; and (iii) there is no deficiency with respect to Taxes against the Company or any Company Subsidiary that remains unpaid.
 - (c) There are no Liens for Taxes other than Permitted Liens on any assets of the Company or any of the Company Subsidiaries.
- (d) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, none of the Company or any Company Subsidiary has any liability under any Tax-sharing, Tax-indemnity or Tax-allocation agreements other than any such agreements that are between or among the Company and/or any Company Subsidiaries or that are customary commercial contracts not primarily related to Taxes.

- (e) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, neither the Company nor any Company Subsidiary has any liability for the Taxes of any Person (other than the Company and the Company Subsidiaries) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law).
- (f) Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, each of the Company and the Company Subsidiaries has withheld and paid all Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, stockholder or other third party.
- (g) None of the Company nor any Company Subsidiary has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.
- (h) Neither the Company nor any Company Subsidiary has entered into any "listed transaction" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).
- (i) Neither the Company nor any Company Subsidiary has distributed the capital stock of any corporation in a transaction intended to qualify under Section 355 of the Code within the past two years, nor has the Company or any Company Subsidiary been distributed in a transaction intended to qualify under Section 355 of the Code within the past two years.
- (j) As of the date hereof, the consummation of the transactions contemplated hereby, including, without limitation, the issuance of the Shares and the Warrants to the Investor and/or any of its Affiliates, will not result in an "ownership change" under Section 382 of the Code. The Company will not grant any waiver or exemption to any Person under the Tax Plan that, if fully utilized by such Person, would cause any exercise of the Warrants to result in an "ownership change" under Section 382 of the Code.
- 3.14 <u>Broker Fees</u>. Except as set forth on Section 3.14 of the Company Disclosure Schedule, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of the Company. The Company has disclosed to the Investor the amounts of any such fees and commissions, all of which shall be paid by the Company.
- 3.15 <u>Insolvency, Creditors</u>. No proceedings have been taken, instituted or, to the knowledge of the Company, are pending for the dissolution or liquidation of the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries has taken any steps to seek protection pursuant to any bankruptcy or insolvency laws, nor does the Company have any knowledge or reason to believe that its creditors intend to initiate involuntary bankruptcy proceedings or any knowledge of any fact that would reasonably lead a creditor to do so. The

Company and its Subsidiaries, taken as a whole, are, as of the date hereof, and after giving effect to the transactions contemplated hereby, will be Solvent. As used herein, (x) "Solvent", with regard to any Person, means that (a) the sum of the assets of such Person, both at a fair valuation and at present fair salable value, exceeds its liabilities, including contingent, subordinated, unmatured, unliquidated and disputed liabilities, (b) such Person has sufficient capital with which to conduct its business, and (c) such Person has not incurred Debts, and does not intend to incur Debts, beyond its ability to pay such Debts as they mature, (y) "Debt" means any liability on a Claim, and (z) "Claim" means (i) a right to payment, whether or not such right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, disputed, undisputed, legal, equitable, secured or unsecured, or (ii) a right to an equitable remedy for breach of performance if such breach gives rise to a payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured; with respect to any such contingent liabilities, such liabilities shall be computed at the amount which, in light of all of the facts and circumstances existing at the time, represents the amount which can reasonably be expected to become an actual or matured liability.

- 3.16 <u>Information in the Proxy Statement</u>. None of the information in or incorporated by reference in the Proxy Statement (or any amendment thereof or supplement thereto), other than information provided by the Investor in writing or electronically specifically for inclusion in the Proxy Statement (or any amendment or supplement thereto) will, at the date mailed to the Stockholders or at the time of the Stockholder Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they are made, not misleading.
- 3.17 <u>Required Vote</u>. The Stockholder Approval is the only vote of the Stockholders required, if any, to approve and adopt this Agreement and the transactions contemplated hereunder, including, without limitation, the issuance of the Shares, the Warrants and the Underlying Shares.
- 3.18 Sarbanes-Oxley; Internal Accounting Controls. The Company is in material compliance with all applicable requirements of the Sarbanes-Oxley Act that are effective as of the date hereof, and any and all applicable rules and regulations promulgated by the SEC thereunder that are effective as of the date hereof. The Company maintains a system of internal accounting controls sufficient to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability, (iii) access to assets is permitted only in accordance with management's general or specific authorization, and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. The Company has established disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the Company and designed such disclosure controls and procedures to ensure that information required to be disclosed by the Company in the reports it files or submits under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms. The Company's certifying officers have evaluated the effectiveness of the Company's disclosure

controls and procedures as of the end of the period covered by the Company's most recently filed periodic report under the Exchange Act. For the avoidance of doubt, the foregoing is qualified by, and should be read in conjunction with, the Company SEC Documents, including Item 9A of the Form 10-K for the fiscal year ended July 31, 2012 filed with the SEC on January 11, 2013.

- 3.19 <u>Private Placement</u>. Assuming the accuracy of the Investor's representations and warranties set forth in Article 4, no registration under the Securities Act is required for the offer and sale of the Shares, the Warrants or the Underlying Shares by the Company to the Investor as contemplated hereby.
- 3.20 No Other Representations or Warranties. Except for the representations and warranties set forth in this Article 3 or as provided in Section 8.2, neither the Company nor any other Person makes any express or implied representation or warranty with respect to the Company or with respect to any other information provided to the Investor in connection with the transactions contemplated hereunder.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF THE INVESTOR

The Investor represents and warrants to the Company as follows:

- 4.1 <u>Organization</u>. The Investor is duly organized, validly existing and in good standing under the Laws of the jurisdiction of its incorporation or organization and has all requisite partnership power and authority to own, lease and operate its properties and assets and to carry on its business as it is now being conducted.
- 4.2 <u>Due Authorization</u>. The Investor has all necessary partnership power and authority to execute and deliver this Agreement and the Registration Rights Agreement, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby.
- 4.3 <u>Due Execution; Enforceability</u>. The execution and delivery of this Agreement and the Registration Rights Agreement by the Investor and the consummation by the Investor of the transactions contemplated hereby have been duly and validly authorized by all necessary action of the Investor, and no other proceedings on the part of the Investor are necessary to authorize this Agreement and the Registration Rights Agreement or to consummate the transactions contemplated hereby or thereby. Each of this Agreement and the Registration Rights Agreement has been duly authorized and validly executed and delivered by the Investor, and assuming due authorization, execution and delivery by the Company, constitutes a legal, valid and binding obligation of the Investor, enforceable against the Investor in accordance with its terms, subject to applicable bankruptcy, insolvency and similar Laws, now or hereafter in effect, affecting creditors' rights generally and general principles of equity.
- 4.4 No Registration Under the Securities Act. The Investor understands that the Shares, the Warrants and the Underlying Shares (collectively, the "Securities") have not been registered, and that except as provided in the Registration Rights Agreement, the Company shall not be required to effect any registration thereof, under the Securities Act or any state securities law and will be issued in reliance upon exemptions contained in the Securities Act or

interpretations thereof and in the applicable state securities laws, and cannot be offered for sale, sold or otherwise transferred unless pursuant to a registration statement or in a transaction exempt from or not subject to registration under the Securities Act.

- 4.5 <u>Shares of Common Stock Beneficially Owned</u>. As of the date hereof, the Investor is the beneficial owner of 540,015 shares of Common Stock, and, except for the beneficial ownership of shares disclosed in the Schedule 13D with respect to the Company filed by Handy & Harman Ltd. with the SEC on October 14, 2011, as amended (the "<u>Schedule 13D</u>"), neither the Investor nor any Affiliate of the Investor is the beneficial owner of any other shares of Common Stock as a result of the constructive ownership rules of Treasury Regulation Section 1.382-2T(h). Except for the understandings disclosed in the Schedule 13D, neither the Investor nor any Affiliate of the Investor is a member of a "group of persons who have a formal or informal understanding among themselves to make a coordinated acquisition of stock" within the meaning of Treasury Regulation Section 1.382-3(a)(1).
- 4.6 Acquisition for Investment. The Securities are being acquired under this Agreement by the Investor in good faith solely for its own account, for investment and not with a view toward resale or other distribution in violation of the Securities Act; provided, however, that the disposition of the Investor's property shall at all times be under its control; provided, further, that by making the representations herein, the Investor does not agree to hold any of the Securities for any minimum or other specific term and reserves the right to dispose of the Securities at any time in accordance with or pursuant to a registration statement or an exemption from registration under the Securities Act. The Securities will not be offered for sale, sold or otherwise transferred by the Investor except pursuant to a registration statement or in a transaction exempt from or not subject to registration under the Securities Act and any applicable state securities laws.
- 4.7 Evaluation of Merits and Risks of Investment. The Investor has such knowledge and experience in financial and business matters such that it is capable of evaluating the merits and risks of its investment in the Securities being acquired hereunder. The Investor is an "accredited investor" within the meaning of Rule 501(a) under the Securities Act. The Investor understands and is able to bear any economic risks associated with such investment (including, without limitation, the necessity of holding the Securities for an indefinite period of time, inasmuch as the Securities have not been registered under the Securities Act).
- 4.8 <u>Additional Information</u>. The Investor acknowledges that it has been afforded the opportunity to ask questions and receive answers concerning the Company and to obtain additional information that it has requested to verify the accuracy of the information contained herein. Notwithstanding the foregoing, nothing contained herein shall operate to modify or limit in any respect the representations and warranties of the Company or to relieve it from any obligations to the Investor for breach thereof or the making of misleading statements or the omission of material facts in connection with the transactions contemplated herein.
- 4.9 <u>U.S. Person</u>. The Investor certifies that (i) it is a natural person resident in the United States or (ii) it is an entity organized or incorporated under the laws of any state of the Unites States or the District of Columbia and having its principal place of business in the United States.

- 4.10 <u>Broker Fees</u>. No broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of the Investor.
- 4.11 <u>Sufficient Funds</u>. The Investor has on hand as of the date hereof, and will have as of the Closing Date, all of the funds available that are necessary to consummate the transactions contemplated by this Agreement.
- 4.12 <u>Information in the Proxy Statement</u>. None of the information supplied by the Investor in writing or electronically specifically for inclusion or incorporation by reference in the Proxy Statement (or any amendment thereof or supplement thereto) will, at the date mailed to the Stockholders or at the time of the Stockholder Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they are made, not misleading.
- 4.13 <u>No Other Representations or Warranties</u>. Except for the representations and warranties set forth in this <u>Article 4</u> or as provided in <u>Section 8.2</u>, neither the Investor nor any other Person makes any express or implied representation or warranty with respect to the Investor or with respect to any other information provided to the Company in connection with the transactions contemplated hereunder.

ARTICLE 5 ADDITIONAL COVENANTS

The Company and the Investor hereby agree to do the following:

5.1 No Solicitation of Transactions.

(a) From and after the date hereof until the Closing Date or, if earlier, the termination of this Agreement in accordance with Article 7, the Company agrees that it and its Subsidiaries and Affiliates shall not, and that it shall cause its directors, officers and other representatives (including legal and financial advisors) not to, directly or indirectly, (i) initiate or solicit, or take any other action to facilitate, a Competing Proposal, (ii) initiate, continue or participate in discussions or negotiations with respect to any Competing Proposal, (iii) furnish or exchange any information to or with any Person in connection with a Competing Proposal, (iv) enter into any letter of intent, agreement in principle or other agreement, arrangement, commitment or understanding with respect to a Competing Proposal, (v) approve or recommend any Competing Proposal, (vi) withdraw, change, amend, modify or qualify, in a manner adverse to the Investor, any recommendation of the Board with respect to the transactions contemplated hereunder, or (vii) resolve or agree to do any of the foregoing (any act or failure to act relating to clauses (v) and (vi) above, a "Change of Recommendation"); provided that the Parties acknowledge and agree that the Company may take any action that would otherwise be prohibited or restricted pursuant to this Section 5.1(a) with respect to, but limited to, (i) any business or assets with a book value of less than five million dollars (\$5,000,000) and (ii) the real property owned by the Company and located in (A) Singapore, (B) Orléans, France and (3)

Kildare, Ireland, including, in each case, entering into any letter of intent, agreement in principle or other agreement or arrangement with respect to such properties or otherwise consummating a sale thereof.

(b) Notwithstanding anything to the contrary contained in Section 5.1(a), if at any time following the date hereof and prior to obtaining the Stockholder Approval, (i) the Company receives a written Competing Proposal that the Board believes in good faith to be bona fide and (ii) the Board determines in good faith, after consultation with the Company's outside legal and financial advisors, that (A) such Competing Proposal constitutes or could reasonably be expected to result, after the taking of any of the actions referred to in either of clause (x) or (y) below, in a Superior Proposal and (B) the failure to take the actions referred to in clauses (x) and (y) below would be inconsistent with the proper exercise of the fiduciary duties of the Board to the Stockholders under applicable Law, then the Company may take the following actions: (x) furnish nonpublic and other information and provide access to the Company's facilities, personnel and representatives of the Company to the third party making such Competing Proposal, if, and only if, prior to so furnishing such information and providing such access, the Company receives from the third party an executed Acceptable Confidentiality Agreement and (y) engage in discussions or negotiations with the third party with respect to the Competing Proposal. The Company shall promptly (and in any event within two Business Days) (A) notify the Investor if it receives a Competing Proposal, including the material terms and conditions thereof and the identity of the Person making such Competing Proposal; (B) notify the Investor of the determination of the Board as provided for in clause (ii) above; (C) keep the Investor reasonably apprised and update the Investor as to any material developments, discussions and negotiations concerning such Competing Proposal; and (D) provide to the Investor any material non-public information concerning the Company or any Company Subsidiary provided to such third party pursuant to clause (x) or (y) above, which was not previously provided to the Investor. Prior to terminating this Agreement to enter into a definitive agreement with respect to any Superior Proposal, the Company shall promptly deliver to the Investor a written notice (the "Offer Notice") of such Superior Proposal (the "Offer"), which Offer Notice shall describe all material terms and conditions of the Superior Proposal, including the party that made the Superior Proposal, and shall include a copy of the relevant proposed transaction agreements with such party. In the case of the initial Superior Proposal, the Company shall give the Investor forty-eight (48) hours after delivery of the Offer Notice to propose revisions to the terms of this Agreement or the transactions contemplated hereby (or make any other proposals) (a "Matching Bid") and during such time shall negotiate and cause its representatives to negotiate (if the Investor notifies the Company that it desires to negotiate), in good faith with the Investor as to such Matching Bid. If the Competing Proposal to which the initial Offer Notice applies is modified in any material respect in response to a Matching Bid (including any subsequent Matching Bid with respect thereto) or otherwise, the Company shall give the Investor a new Offer Notice following which the Investor shall have twenty-four (24) hours to propose a new Matching Bid, and during such time the Company shall negotiate and cause its representatives to negotiate (if the Investor notifies the Company that it desires to negotiate), in good faith with the Investor as to such Matching Bid. The Board shall take into account the terms and conditions of any Matching Bid in determining whether the Competing Proposal continues to constitute a Superior Proposal.

- (c) Notwithstanding the limitations set forth in Section 5.1(a), if the Board has concluded in good faith after consultation with the Company's outside legal and financial advisors that a Competing Proposal constitutes a Superior Proposal, after and subject to compliance with the requirements contained in Section 5.1(b), and that the failure to take the following actions would be inconsistent with the proper exercise of the fiduciary duties of the Board to the Stockholders under applicable Law, then the Board may prior to obtaining Stockholder Approval cause the Company to effect a Change of Recommendation, terminate this Agreement in accordance with Section 7.1(e) and concurrently with such termination enter into a binding written agreement with respect to such Superior Proposal; provided, that the Company shall not terminate this Agreement in accordance with Section 7.1(e) and any purported termination in accordance therewith shall be void and of no force and effect unless concurrently with such termination the Company pays the Termination Fee to the Investor.
- (d) Notwithstanding the limitations set forth in Section 5.1(a) and prior to obtaining Stockholder Approval, the Board may, other than in connection with a Competing Proposal (which is subject to Sections 5.1(b) and (c)), effect a Change of Recommendation under Section 5.1(a)(vi) if the Board has concluded in good faith after consultation with the Company's outside legal and financial advisors that the failure of the Board to effect such Change of Recommendation would be inconsistent with the proper exercise of the fiduciary duties of the Board to the Stockholders under applicable Law.
- (e) Nothing contained in this Agreement shall prohibit the Company or the Board from making any disclosure to the Stockholders if the Board has reasonably determined in good faith, after consultation with outside legal counsel, that the failure to do so would be inconsistent with any applicable Law; *provided*, that disclosures under this Section 5.1(e) shall not be a basis, in themselves, for the Investor to terminate this Agreement pursuant to Section 7.1(b).
 - 5.2 Legend. The Investor agrees with the Company that the certificates evidencing the Shares will bear the following legend:

"THE SECURITIES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD OR TRANSFERRED UNLESS THERE IS AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT COVERING SUCH SECURITIES OR THE SECURITIES ARE SOLD AND TRANSFERRED IN A TRANSACTION THAT IS EXEMPT FROM OR NOT SUBJECT TO THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF SUCH ACT."

5.3 Preparation of Proxy Statement; Stockholder Meeting.

(a) In connection with the Stockholder Meeting, the Company shall prepare and file with the SEC the Proxy Statement, amended to reflect the transactions and terms contemplated hereby and by the Settlement Agreement, as promptly as reasonably practicable, and in any event by no later than five (5) Business Days following the date hereof. The Investor and the Company shall cooperate in the preparation and filing of the Proxy Statement to the extent related to the transactions contemplated hereby. The Company will advise the Investor

promptly after it receives any oral or written request by the SEC for amendment of the Proxy Statement or comments thereon and responses thereto or requests by the SEC for additional information insofar as any such request for amendment, comment, response or request for additional information relates to the transactions contemplated hereby and will promptly provide the Investor with copies of any written communication from the SEC or any state securities commission. The Company shall give the Investor and its counsel a reasonable opportunity to review and comment on the Proxy Statement, any amendments thereto and any responses of the Company in response to any request or comment of the SEC, in each case, to the extent related to the transactions contemplated hereby, and shall give due consideration to all reasonable additions, deletions or changes suggested thereto by the Investor and its counsel. Other than the Proxy Statement, or as otherwise required by a court order or the Company's governance documents, the Company shall agree not to file a proxy statement or amend the Proxy Statement, or take any action that would establish a new nomination deadline, relating to its next annual meeting of Stockholders until after the earlier of the Closing or the termination of this Agreement. The Company shall cause the Proxy Statement to be mailed to the Stockholders as promptly as reasonably practicable after it has been cleared by the SEC.

- (b) If, at any time prior to the Closing Date, any information relating to the Parties, or any of their respective Affiliates, officers or directors, is discovered by such Party that should be set forth in an amendment or supplement to the Proxy Statement so that the Proxy Statement would not include any misstatement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they are made, not misleading, such Party shall as promptly as practicable notify the other Party and an appropriate amendment or supplement describing such information shall be filed by the Company with the SEC as promptly as practicable after the Investor has had a reasonable opportunity to review and comment thereon (to the extent related to the transactions contemplated hereby) and, to the extent required by applicable Law, disseminated to the Stockholders. The Proxy Statement will comply as to form in all material respects with the applicable requirements of the Securities Act, the Exchange Act and other applicable Law, including the regulations and requirements of NASDAQ.
- (c) In addition to their obligations pursuant to Section 5.3(a) and Section 5.3(b), the Company shall make all necessary filings with respect to the transactions contemplated hereunder, including the issuance of the Shares, the Warrants and the Underlying Shares, under the Securities Act, the Exchange Act and applicable Blue Sky Laws and make available to the Investor copies of such filings.
- (d) As promptly as reasonably practicable after the Proxy Statement is definitive, the Company shall duly call, give notice to the Stockholders of, convene and hold the Stockholder Meeting. The Company shall (i) solicit the Stockholder Approval and (ii) subject to a Change in Recommendation permitted by Sections 5.1(c) or (d), include in the Proxy Statement the Board's recommendation to the Stockholders that they approve the transactions contemplated hereby, and include therein disclosure regarding the approval of the Board.
- 5.4 <u>Conduct of Business</u>. During the period between the date hereof and the Closing, the Company shall: (a) provide the Investor with all monthly financial statements, budgets and forecasts provided to the Board concurrently with their provision to the Board; (b)

conduct its business in the ordinary course in a manner consistent with past practice, including, without limitation, subject to Section 5.1; (c) not grant any waiver or exemption to any Person under any Rights Agreement or modify or amend any Rights Agreement for the benefit of any Person; and (d) subject to Section 5.1, not, in any manner, issue or sell, agree to issue or sell, enter into or affect any issuance or sale, or be deemed to have entered into or affected any issuance or sale, of any Equity Interests (other than in accordance with the Rights Agreements, the Incentive Plans or the ESPP or repurchases of cashless exercises of options or tax withholdings upon the vesting of options of incentive awards). In furtherance of the foregoing, the Parties acknowledge and agree that the Company may take any action that would otherwise be prohibited or restricted pursuant to Section 5.1(a) with respect to, but limited to, (i) any business or assets with a book value of less than five million dollars (\$5,000,000) and (ii) the real property owned by the Company and located in (A) Singapore, provided that the Company shall not consummate such transaction (without consent of the Investor, which shall not be unreasonably delayed, conditioned or withheld) unless such transaction is on terms and conditions substantially similar to, or more favorable to the Company than, those set forth on the term sheet delivered to the Investor on February 4, 2013 or as otherwise set forth on Section 5.4 of the Company Disclosure Schedule, (B) Orléans, France and (3) Kildare, Ireland, including, in each case, entering into any letter of intent, agreement in principle or other agreement or arrangement with respect to such properties or otherwise consummating a sale thereof.

5.5 <u>Further Assurances</u>. From time to time after the date of this Agreement, the Parties shall execute, acknowledge and deliver to the other Party such other instruments, documents, and certificates and will take such other actions as the other Party may reasonably request in order to consummate the transactions contemplated by this Agreement.

5.6 Indemnification.

(a) In consideration of the Investor's execution and delivery of this Agreement, the Registration Rights Agreement and the Settlement Agreement and acquiring the Securities thereunder and in addition to all of the Company's other obligations hereunder and thereunder, the Company shall defend, protect, indemnify and hold harmless the Investor and each Affiliate of the Investor and all of their respective shareholders, partners, members, officers, directors, employees and direct or indirect investors and all of the foregoing Persons' agents and other representatives (collectively, the "Indemnitees") from and against any and all actions, causes of action, suits, claims, losses, costs, penalties, fees, liabilities and damages, and expenses in connection therewith, including all judgments, amounts paid in settlement, court costs and costs of investigation, and including reasonable attorneys' fees and disbursements (irrespective of whether any such Indemnitee is a party to the action for which indemnification hereunder is sought) (the "Indemnified Liabilities"), incurred by any Indemnitee as a result of, or arising out of, or relating to (i) any misrepresentation or breach of any representation or warranty made by the Company in this Agreement, the Warrants, the Registration Rights Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby or (ii) any breach of any covenant, agreement or obligation of the Company contained in this Agreement, the Warrants, the Registration Rights Agreement, the Settlement Agreement or any other agreement, certificate, instrument or document contemplated hereby or thereby.

(b) Promptly after receipt by an Indemnitee under this Section 5.6 of notice of the commencement of any action or proceeding (including any governmental action or proceeding) involving an Indemnified Liability, such Indemnitee shall, if a claim for indemnification in respect thereof is to be made against the Company under this Section 5.6, deliver to the Company a written notice of the commencement thereof, and the Company shall have the right to participate in, and, to the extent the Company so desires, to assume control of the defense thereof with counsel reasonably satisfactory to the Company and the Indemnitee; provided, however, that an Indemnitee shall have the right to retain its own counsel at its own expense, except that the reasonable fees and expenses of such counsel shall be borne by the Company if (i) the employment of such counsel has been specifically authorized by the Company in writing, (ii) the Company has failed after a reasonable period of time to assume such defense and to employ counsel or (iii) in such action there is, based upon the advice of counsel to the Investor, a material conflict on any material issue between the position of the Company and the position of the Investor, in which case the Company shall be responsible for the reasonable fees and expenses of no more than one such separate counsel. The Indemnitee shall reasonably cooperate with the Company in connection with any negotiation or defense of any such action or Indemnified Liabilities by the Company and shall furnish to the Company all information reasonably available to the Indemnitee that relates to such action or Indemnified Liabilities. The Company shall keep the Indemnitee fully apprised at all times as to the status of the defense or any settlement negotiations with respect thereto. The Company shall not be liable for any settlement of any action, claim or proceeding effected without its prior written consent, provided, however, that the Company shall not unreasonably withhold, delay or condition its consent. The Company shall not, without the prior written consent of the Indemnitee, consent to entry of any judgment or enter into any settlement or other compromise which (i) does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Indemnitee of a release from all liability in respect to such Indemnified Liabilities, (ii) requires any admission of wrongdoing by such Indemnitee, or (iii) obligates or requires an Indemnitee to take, or refrain from taking, any action. Following indemnification as provided for hereunder, the Company shall be subrogated to all rights of the Indemnitee with respect to all third parties, firms or corporations relating to the matter for which indemnification has been made. The failure to deliver written notice to the Company within a reasonable time after the Indemnitee has received notice of the commencement of any such action shall not relieve the Company of any liability to the Indemnitee under this Section 5.6, except to the extent that the Company is materially prejudiced in its ability to defend such action.

(c) The parties acknowledge and agree that in no event shall the Indemnitees be entitled to payments under this <u>Section 5.6</u> in excess of the Purchase Price.

(d) Subject to Section 8.7 and other than with respect to fraud, (i) the indemnity agreements contained herein shall be the sole and exclusive remedy of the Investor for any and all claims, damages or other matters arising under, out of, or related to this Agreement, the Warrants, the Registration Rights Agreement, the Settlement Agreement and agreements, certificates, instruments or documents contemplated hereby or thereby, and (ii) no Person will have any other entitlement, remedy or recourse, whether in contract, tort, strict liability, equitable remedy or otherwise, it being agreed that all of such other remedies, entitlements and recourse are expressly waived and released by the Parties to the fullest extent permitted by Law.

- 5.7 <u>Fees</u>. Except as otherwise set forth herein or as otherwise agreed upon in writing by the Company and the Investor, each Party shall bear its own expenses in connection with the sale of the Shares and the Warrants to the Investor.
- 5.8 <u>Integration</u>. The Company shall not sell, offer for sale or solicit offers to buy or otherwise negotiate in respect of any security (as defined in Section 2 of the Securities Act) that would be integrated with the offer or sale of any of the Securities in a manner that would require the registration under the Securities Act of the sale of any of the Securities.
- 5.9 Reservation of Common Stock. As of the date hereof, the Company has reserved and the Company shall continue to reserve and keep available at all times, free of preemptive rights, a sufficient number of shares of Common Stock for the purpose of enabling the Company to issue Shares pursuant to this Agreement and Underlying Shares pursuant to any exercise of the Warrants. The Company shall reserve and so long as any Securities are outstanding, the Company shall continue to reserve and keep available at all times, free of preemptive rights, a number of shares of Common Stock equal to the maximum number of Shares issuable pursuant to this Agreement and Underlying Shares issuable pursuant to any exercise of the Warrants.
- 5.10 Governmental Approvals. Each of the Company and the Investor undertakes and agrees to file as soon as practicable after the Parties have determined in their reasonable discretion that consent is required under the HSR Act to consummate the sale of the Securities (which determination and filing shall be made no later than the latest to occur of (i) 35 days prior to the Stockholder Meeting and (ii) five Business Days after the date on which the Company notifies the Investor of the date on which the Stockholder Meeting will be held), a Notification and Report Form and documentary materials in respect of the transactions contemplated by this Agreement that substantially comply with the provisions of the HSR Act, and the rules thereunder, with the United States Federal Trade Commission and the Antitrust Division of the United States Department of Justice (and shall file as soon as practicable any form or report required by any other Governmental Entity relating to antitrust matters) if such filing is required. In furtherance of the foregoing, each Party agrees to (A) use commercially reasonable efforts to comply at the earliest practicable date with any request under the HSR Act for additional information, documents, or other materials received by each of them or any of their respective Affiliates from any Governmental Entity in respect of such filings or such transactions, and (B) cooperate with each other in connection with any such filing and in connection with resolving any investigation or other inquiry of any Governmental Entity under the HSR Act with respect to any such filing or any such transaction. Each Party shall use its commercially reasonable efforts to furnish to the other Party all information required for any application or other filing to be made pursuant to any Law in connection with the transactions contemplated by this Agreement (including, to the extent permitted by Law, responding to any reasonable requests for copies of documents filed with the other Party's prior filings). The Investor and the Company shall permit each other to review in advance any proposed communication to any Governmental Entity, subject to applicable Law and provided that the Parties shall not be required to provide each other material related to a Party's valuation of the transactions contemplated hereby or any other confidential information of a Party. Each such Party shall promptly inform the other Party of any oral communication with, and provide copies of written communications with, any Governmental Entity regarding any such filings or any such

transaction. No Party shall independently participate in any meeting with any Governmental Entity in respect of any such filings, investigation, or other inquiry without giving the other Party prior notice of the meeting and, to the extent permitted by such Governmental Entity, the opportunity to attend and/or participate. Subject to Law, the Parties will consult and cooperate with one another in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any Party relating to proceedings under the HSR Act. The Investor and the Company shall each pay one half of all filing fees in connection with all filings under the HSR Act.

5.12 Financial Information. At the request and expense of the Investor or any of its Affiliates, the Company shall permit the Investor or such Affiliate to include the Company's audited financial statements and any other financial information of the Company in any filing or report of the Investor or such Affiliate filed with the SEC under the Exchange Act or the Securities Act to the extent required under the SEC's rules and regulations relating to significant subsidiaries or equity method accounting, and the Company shall, and shall cause its independent registered public accounting firm to, reasonably cooperate with the Investor or such Affiliate with respect to the provision of such required information as promptly as practicable. The Investor shall, or shall cause its applicable Affiliate to, defend, protect, indemnify and hold harmless the Company and each Affiliate of the Company and all of their respective officers, directors, employees and all of the foregoing Persons' agents and other representatives (collectively, the "Company Indemnitees") from and against any and all Indemnified Liabilities incurred by any Company Indemnitee as a result of, or arising out of, or relating to the inclusion of such audited financial statements and other financial information of the Company in any filing or report of the Investor or such Affiliate filed with the SEC under the Exchange Act or the Securities Act; provided, however, that the Investor and its Affiliates will not have any liability, and will not be required to indemnify the Company Indemnitees, with respect to, and the Company will be solely responsible for, the content and substance of such audited financial statements and other financial information provided by or on behalf of the Company.

ARTICLE 6 CONDITIONS TO CLOSING

- 6.1 <u>Conditions to Investor's Obligations</u>. The obligation of the Investor to purchase the Shares and Warrants as set forth in <u>Section 2.1</u> shall be subject to the satisfaction or waiver by the Investor of each of the following conditions, in each case, on the Closing Date:
- (a) The Company shall have delivered to the Investor a copy of the executed and delivered instruction letter from the Company to the Transfer Agent instructing the Transfer Agent to deliver a stock certificate representing the Shares to the Investor or to credit the Shares to the Investor's or its designee's balance account with The Depository Trust Company through its Deposit Withdrawal At Custodian system, or to the Investor's or its designee's direct registration account.
 - (b) The Company shall have executed and delivered to the Investor the Registration Rights Agreement.

- (c) The Company shall have executed and delivered to the Investor the Warrants in favor of the Investor.
- (d) No change, event, development, condition, occurrence or effect shall have occurred or come into existence since the date of this Agreement that has had or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
 - (e) The Stockholder Nominees shall have been elected to serve on the Board at the Stockholder Meeting.
- (f) There shall have been no law, rule, regulation, order, judgment or decree enacted, entered or promulgated by any Governmental Entity or NASDAQ which would have the effect of prohibiting the transactions contemplated by this Agreement.
- (g) All representations and warranties of the Company contained in this Agreement (without giving effect to any references to any Company Material Adverse Effect or materiality qualifications and other qualifications based upon the concept of materiality or similar phrases contained therein) shall be true and correct as of the Closing Date with the same force and effect as if made as of such date, except for representations and warranties that relate to a specific date or time (which need only be true and correct as of such date or time), except as has not had and would not reasonably be expected to have, individually or in the aggregate with all other such failures to be true or correct, a Company Material Adverse Effect.
- (h) The Company shall not have breached or failed, in any material respect, to perform or to comply with any agreement or covenant to be performed or complied with by it under this Agreement or any such breach or failure shall have been cured, and there shall not be in effect any legal requirement or injunction that prohibits the consummation of the transactions contemplated under this Agreement.
 - (i) The Stockholder Approval has been obtained.
- (j) The Company shall have obtained all governmental, regulatory, NASDAQ and third-party consents and approvals, if any (including any consent required under the HSR Act), necessary for the issuance and sale of the Securities and the listing of the Shares and the Underlying Shares on NASDAQ, subject to official notice of issuance, and the Investor shall have obtained any consent required under the HSR Act necessary for its purchase of the Securities.
- (k) The Company shall have delivered to the Investor an officer's certificate dated as of the Closing Date attaching copies of resolutions duly adopted by the Board, authorizing and approving the execution and delivery of this Agreement, the Warrants and the Registration Rights Agreement and the consummation of the transactions contemplated hereby and thereby.
- 6.2 <u>Conditions to Company's Obligations</u>. The obligation of the Company to consummate the Closing shall be subject to the satisfaction or waiver by the Company of each of the following conditions, in each case, on the Closing Date:
 - (a) The Investor shall have executed and delivered to the Company the Registration Rights Agreement.

- (b) The Investor shall have delivered the Purchase Price.
- (c) The Stockholder Approval shall have been obtained.
- (d) There shall have been no law, rule, regulation, order, judgment or decree enacted, entered or promulgated by any Governmental Entity or NASDAQ which would have the effect of prohibiting the transactions contemplated by this Agreement.
- (e) All representations and warranties of the Investor contained in this Agreement shall be true and correct in all material respects as of the Closing Date with the same force and effect as if made as of such date, except for representations and warranties that relate to a specific date or time (which need only be true and correct in all material respects as of such date or time).
- (f) The Investor shall not have breached or failed, in any material respect, to perform or to comply with any agreement or covenant to be performed or complied with by it under this Agreement or any such breach or failure shall have been cured, and there shall not be in effect any legal requirement or injunction that prohibits the consummation of the transactions contemplated under this Agreement.
- (g) The Company shall have obtained all governmental, regulatory, NASDAQ or third-party consents and approvals, if any (including any consent required under the HSR Act), necessary for the issuance and sale of the Securities, and the Investor shall have obtained any consent required under the HSR Act necessary for its purchase of the Securities.
- (h) The Investor shall have delivered to the Company an officer's certificate dated as of the Closing Date attaching copies of resolutions duly adopted by the board of directors or equivalent body of the Investor, authorizing and approving the execution and delivery of this Agreement and the Registration Rights Agreement and the consummation of the transactions contemplated hereby and thereby.

ARTICLE 7 TERMINATION

- 7.1 <u>Termination</u>. This Agreement may be terminated at any time:
 - (a) by mutual written agreement of the Company and the Investor;
- (b) by the Investor, at any time prior to the Closing Date if: (i) there shall be an inaccuracy or breach of any representation or warranty of the Company contained in this Agreement or breach of any covenant of the Company contained in this Agreement, in any case, such that any closing condition contained in Section 6.1 is not or is reasonably likely not to be satisfied, (ii) the Investor shall have delivered to the Company written notice of such inaccuracy or breach, and (iii) either such inaccuracy or breach is not capable of cure or at least twenty (20) calendar days shall have elapsed since the date of delivery of such written notice to the Company

and such inaccuracy or breach shall not have been cured; *provided*, *however*, that the Investor shall not be permitted to terminate this Agreement pursuant to this Section 7.1(b) if: (A) any material covenant of the Investor contained in this Agreement shall have been breached in any respect, and such breach shall not have been cured; or (B) there shall be an inaccuracy or breach of any representation or warranty of the Investor contained in this Agreement, in each of the foregoing cases such that any closing condition contained in Section 6.2 is not or is reasonably not likely to be satisfied;

- (c) by the Company, at any time prior to the Closing Date if: (i) there shall be an inaccuracy or breach of any representation or warranty of the Investor contained in this Agreement or breach of any covenant of the Investor contained in this Agreement, in any case, such that any closing condition contained in Section 6.2 is not or is reasonably likely not to be satisfied, (ii) the Company shall have delivered to the Investor written notice of such inaccuracy or breach, and (iii) either such inaccuracy or breach is not capable of cure or at least twenty (20) calendar days shall have elapsed since the date of delivery of such written notice to the Investor and such inaccuracy or breach shall not have been cured; *provided, however*, that the Company shall not be permitted to terminate this Agreement pursuant to this Section 7.1(c) if: (A) any material covenant of the Company contained in this Agreement shall have been breached in any respect, and such breach shall not have been cured; or (B) there shall be an inaccuracy or breach of any representation or warranty of the Company contained in this Agreement, in each of the foregoing cases such that any closing condition contained in Section 6.1 is not or is reasonably not likely to be satisfied;
 - (d) by the Investor if the Board shall have effected a Change of Recommendation;
- (e) by the Company if, prior to obtaining the Stockholder Approval, the Board (or any committee thereof) has received a Superior Proposal and has determined to accept it, subject to compliance by the Company with <u>Section 5.1</u> in all material respects;
- (f) by the Investor if either of the Stockholder Nominees has not been elected to serve as a director of the Company at the Stockholder Meeting or any adjournment thereof;
- (g) by either the Company or the Investor if the Stockholder Approval has not been obtained at the Stockholder Meeting or any adjournment thereof; or
- (h) by either the Company or the Investor if the Closing has not occurred by May 6, 2013, unless delayed by any governmental or regulatory consent required under the HSR Act that has not been obtained such that any closing condition contained in Section 6.1(j) or Section 6.2(g) has not been satisfied, in which case such date shall be extended until May 31, 2013; provided, however, that the right to terminate this Agreement under this Section 7.1(h) shall not be available to any Party whose failure to fulfill any obligation under this Agreement has been the cause of, or resulted in, the failure of the Closing to occur by such date.
- 7.2 <u>Effect of Termination</u>. If this Agreement is terminated by either Party pursuant to the provisions of <u>Section 7.1</u>, except as set forth in the immediately following

sentence, this Agreement shall immediately become void and there shall be no further obligations on the part of either Party or their respective directors, officers, employees, agents or representatives, except (a) with respect to the provisions of this Article 7 and Article 8, which shall survive the termination of this Agreement, and (b) with respect to any liabilities or damages incurred or suffered as a result of the willful breach by the Company or the Investor of any of their respective representations, warranties, covenants or other agreements set forth in this Agreement. Notwithstanding anything to the contrary contained in this Agreement, nothing shall limit or prevent any Party from exercising any rights or remedies it may have under Section 8.7 in lieu of terminating this Agreement to the provisions of Section 7.1.

7.3 Termination Fee and Investor Expenses.

(a) Notwithstanding anything to the contrary contained herein, the Company agrees that if this Agreement is terminated pursuant to Section 7.1(e), Section 7.1(f) or Section 7.1(g) then the Company shall pay the Investor the Termination Fee and the Investor Expenses in immediately available funds (x) within two Business Days after the date of termination, in the case of a termination pursuant to Section 7.1(g) and (y) concurrently with such termination, in the case of a termination pursuant to Section 7.1(e). "Termination Fee" means \$900,000. In no event shall payment of more than one Termination Fee be made. Notwithstanding anything to the contrary contained in this Agreement: (i) the Investor's right to receive the Termination Fee and the Investor Expenses pursuant to this Section 7.3(a) shall be the Investor's sole and exclusive remedy against the Company or any of its Affiliates, stockholders, directors, officers, employees, agents or representatives for any loss, claim, damage, liability or expense suffered as a result of the failure of any of the transactions contemplated hereby to be consummated in circumstances giving rise to the obligation of the Company to pay the Termination Fee and the Investor Expenses under this Section 7.3(a); and (ii) upon payment of all amounts that are required to be paid pursuant to this Section 7.3(a), none of the Company or any of its Affiliates, stockholders, directors, officers, employees, agents or representatives shall have any further liability or obligation relating to or arising out of this Agreement (other than any obligation to pay any amounts due pursuant to the second sentence of Section 7.3(b).

(b) The Company acknowledges that the agreements contained in this Section 7.3 are an integral part of the transactions contemplated by this Agreement and that without these agreements, the Investor would not enter into this Agreement. If the Company shall fail to pay any amount payable pursuant to Section 7.3(a) when due, the Company shall reimburse the Investor for all costs and expenses actually incurred or accrued by the Investor (including reasonable fees and expenses of counsel) in connection with the collection under and enforcement of such Section.

ARTICLE 8 MISCELLANEOUS

8.1 <u>Notices</u>. Any notices or other communications required or permitted under, or otherwise given in connection with, this Agreement shall be in writing and shall be deemed to have been duly given (a) when delivered or sent if delivered in Person or sent by facsimile transmission (provided confirmation of facsimile transmission is obtained), (b) on the

fifth (5th) Business Day after dispatch by registered or certified mail, return receipt requested, postage prepaid, (c) on the next Business Day if transmitted by national overnight courier or (d) on the date delivered if sent by email (provided confirmation of email receipt is obtained from the recipient and if followed by transmittal by national overnight courier or hand for delivery on the next Business Day), in each case as follows (or to such other address for a Party as shall be specified by like notice):

(i) if to the Investor, to:

c/o Steel Partners II, L.P. 590 Madison Avenue New York, New York 10022 Attention: Warren Lichtenstein Leonard McGill

Phone: (212) 520-2300 Fax: (212) 520-2337

Email: warren@steelpartners.com len@steelpartners.com

with a copy (for informational purposes only) to:

Olshan Frome Wolosky LLP 65 East 55th Street New York, New York 10022 Attention: Steven Wolosky Michael R. Neidell

Phone: (212) 451-2300 Fax: (212) 451-2222

Email: swolosky@olshanlaw.com mneidell@olshanlaw.com

(ii) if to the Company, to:

ModusLink Global Solutions, Inc. 1601 Trapelo Road Suite 170 Waltham, MA 02451 Attn: Peter I. Gray

Attn: Peter L. Gray Phone: (781) 663-5024 Fax: (781) 663-5095

Email: peter_gray@moduslink.com

with a copy to:

Latham & Watkins LLP 233 South Wacker Drive, Suite 5800 Chicago, IL 60606 Attn: Mark D. Gerstein

Zachary A. Judd
Phone: (312) 876-7700
Fax: (312) 993-9769
Email: mark.gerstein@lw.com
zachary.judd@lw.com

- 8.2 <u>Survival of Representations and Warranties, etc.</u> All representations and warranties made in, pursuant to or in connection with this Agreement will survive the execution and delivery of this Agreement indefinitely, notwithstanding any investigation at any time made by or on behalf of any Party hereto; and all statements contained in any certificate, instrument or other writing delivered by or on behalf of any Party required to be made pursuant to the terms of this Agreement or required to be made in connection with or in contemplation of the transactions contemplated by this Agreement will constitute representations and warranties by such Party pursuant to this Agreement.
- 8.3 <u>Assignment</u>. This Agreement will be binding upon and will inure to the benefit of each and all of the Parties, and neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by either Party without the prior written consent of the other Party, except that the Investor may assign or delegate any or all of its rights, interests or obligations hereunder to an Affiliate without such consent, *provided* that such Affiliate agrees in a writing reasonably satisfactory to the Company to be bound by the terms and conditions of the Settlement Agreement.
- 8.4 Entire Agreement. Except for the Registration Rights Agreement, the Warrants and the Settlement Agreement, this Agreement contains the entire agreement by and between the Company and the Investor with respect to the transactions contemplated by this Agreement and supersedes all prior agreements and representations, written or oral, with respect thereto.
- 8.5 Waivers and Amendments. This Agreement may be amended, modified, superseded, cancelled, renewed or extended, and the terms and conditions of this Agreement may be waived, only by a written instrument signed by the Parties or, in the case of a waiver, by the Party waiving compliance. No delay on the part of either Party in exercising any right, power or privilege pursuant to this Agreement will operate as a waiver thereof, nor will any waiver on the part of either Party of any right, power or privilege pursuant to this Agreement, nor will any single or partial exercise of any right, power or privilege pursuant to this Agreement, preclude any other or further exercise thereof or the exercise of any other right, power or privilege pursuant to this Agreement. The rights and remedies provided pursuant to this Agreement are cumulative and are not exclusive of any rights or remedies which any party otherwise may have at law or in equity.

8.6 Governing Law; Jurisdiction; Venue; Process; Jury Trial. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO ANY CHOICE OF LAW OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE

APPLICATION OF THE LAWS OF ANY JURISDICTION OTHER THAN THE STATE OF NEW YORK. Each of the Parties hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of any Delaware State court, or Federal court of the United States of America, sitting in Delaware, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement or the agreements delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each of the Parties hereby irrevocably and unconditionally (i) agrees not to commence any such action or proceeding except in such courts, (ii) agrees that any claim in respect of any such action or proceeding may be heard and determined in such Delaware State court or, to the extent permitted by law, in such Federal court, (iii) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such Delaware State or Federal court, and (iv) waives, to the fullest extent permitted by Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such Delaware State or Federal court. Each of the Parties agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each Party irrevocably consents to service of process in the manner provided for notices in Section 8.1. Nothing in this Agreement will affect the right of any Party to serve process in any other manner permitted by Law. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY.

- 8.7 Remedies. Each Party acknowledges and agrees that (a) the covenants, obligations and agreements of each Party contained in this Agreement relate to special, unique and extraordinary matters and (b) a violation of any of the covenants, obligations or agreements contained in this Agreement by a Party will cause the other Party irreparable injury for which adequate remedies are not available at law. Therefore, the Parties agree that each Party shall be entitled to an injunction, restraining order or such other equitable relief (without the requirement to post bond) as a court of competent jurisdiction may deem necessary or appropriate to restrain the other Party from committing any violation of such covenants, obligations or agreements and to specifically enforce the terms of this Agreement, in addition to any other remedies available at law or in equity. Furthermore, the Parties hereby waive, and shall use their commercially reasonable best efforts to cause their stockholders, members, partners, Affiliates and representatives to waive, any requirement for the securing or posting of any bond in connection with such remedy.
- 8.8 <u>Counterparts</u>. This Agreement may be executed by the Parties by facsimile or electronic mail transmission in one or more counterparts, and by the different Parties in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.
- 8.9 <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

8.10 <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the extent possible.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the date first above written.

MODUSLINK GLOBAL SOLUTIONS, INC.

By: /s/ Peter L. Gray

Name: Peter L. Gray

Title: Executive Vice President, Chief Administrative Officer and General

Counsel

STEEL PARTNERS HOLDINGS L.P.

By: Steel Partners Holdings GP Inc., its general partner

By: /s/ Warren Lichtenstein

Name: Warren Lichtenstein

Title:

EXHIBIT A

FORM OF REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT, dated as of , 2013 (this "<u>Agreement</u>"), by and between ModusLink Global Solutions, Inc., a Delaware corporation (the "<u>Company</u>"), and Steel Partners Holdings L.P., a Delaware limited partnership (the "<u>Investor</u>"). Unless otherwise defined herein, all capitalized terms used in this Agreement shall have the meanings set forth in the Investment Agreement (defined below).

WITNESSETH:

WHEREAS, the Company and the Investor entered into that certain Investment Agreement, dated as of February 11, 2013 (the "Investment Agreement");

WHEREAS, in connection with transactions contemplated by, and pursuant to the terms and conditions of, the Investment Agreement, the Company will issue and sell to the Investor (i) shares (the "Shares") of the common stock of the Company, par value \$0.01 (the "Common Stock") and (ii) warrants (the "Warrants") to purchase newly issued shares of Common Stock (the shares of Common Stock issuable upon exercise of the Warrants, the "Underlying Shares"); and

WHEREAS, in connection with the transactions contemplated by the Investment Agreement, the Company and the Investor have agreed to enter into this Agreement with respect to the Shares, the Underlying Shares and the Common Stock.

NOW, THEREFORE, in consideration of the mutual covenants and premises contained in this Agreement and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

1. Representations and Warranties of the Investor.

The Investor hereby represents and warrants to the Company as follows:

- (a) The Investor is a limited partnership duly organized and validly existing under the laws of the State of Delaware.
- (b) The Investor has full legal right, power and authority to enter into and perform this Agreement, and the execution and delivery of this Agreement by the Investor and the consummation by the Investor of the transactions contemplated hereby have been duly authorized by all necessary partnership action on behalf of the Investor. This Agreement constitutes a legally valid and binding agreement of the Investor, enforceable against it in accordance with its terms.
- (c) The Investor is an "accredited investor" within the meaning set forth in the rules and regulations promulgated under the Securities Act of 1933, as amended (the "Securities Act"). The Investor has such knowledge and experience in financial and business matters that the Investor is capable of evaluating the risks and merits of the transactions contemplated hereby,

including an investment in the Shares and Underlying Shares to be received by the Investor pursuant to the Investment Agreement and the Warrants and is able to bear the economic risk inherent in holding such Shares and Underlying Shares. The Shares and Underlying Shares are being acquired by the Investor on behalf of itself and not for any other Person and for the account of the Investor, not as a nominee or agent and not for the account of any other Person. The Investor is not obligated to transfer all or any portion of the Shares and Underlying Shares to any other Person, nor does the Investor have any present intention, agreement or understanding to do so. Notwithstanding the foregoing, the disposition of the Investor's property shall at all times be under its control, and by making the representations herein, the Investor does not agree to hold any of the Shares or Underlying Shares for any minimum or other specific term and reserves the right to dispose of the Shares and the Underlying Shares at any time in accordance with or pursuant to a registration statement or an exemption from registration under the Securities Act.

2. Representations and Warranties of the Company.

The Company hereby represents and warrants to the Investor as follows:

- (a) The Company is a corporation duly organized and validly existing under the laws of the State of Delaware.
- (b) The Company has full legal right, power and authority to enter into and perform this Agreement, and the execution and delivery of this Agreement by the Company and the consummation by the Company of the transactions contemplated hereby have been duly authorized by the Board of Directors of the Company, authorization by no other body or party being required by law. This Agreement constitutes a legally valid and binding agreement of the Company, enforceable against it in accordance with its terms.
- (c) As of the date hereof, the Company's authorized capital stock consists of: (i) 1,400,000,000 shares of Common Stock and (ii) 5,000,000 shares of Preferred Stock. As of the close of business on February 5, 2013, (x) there were 44,021,864 shares of Common Stock issued and outstanding, (y) no shares of Common Stock were issued and held in the treasury of the Company or otherwise owned by the Company, and (z) there were no shares of Preferred Stock issued and outstanding. As of the close of business on February 5, 2013, the Company has no shares of Common Stock or Preferred Stock reserved for or otherwise subject to issuance, except for (i) 140,000 shares of Series A Junior Participating Preferred Stock, par value \$0.01 per share, reserved for issuance in connection with the exercise of Tax Plan Rights issued pursuant to the Tax Plan, (ii) 140,000 shares of Series B Junior Participating Preferred Stock, par value \$0.01 per share, reserved for issuance in connection with the exercise of Rights Plan Rights issued pursuant to the Rights Plan and (iii) 1,691,388 shares of Common Stock reserved for issuance upon exercise of outstanding options under the Incentive Plans and the ESPP.
- (d) The Shares have been duly authorized for issuance and sale to the Investor pursuant to the Investment Agreement and, when issued and delivered by the Company pursuant to the terms of the Investment Agreement against payment of the consideration set forth therein, shall be validly issued, fully paid and nonassessable and will be free from all taxes, Liens, preemptive rights and charges with respect to the issue thereof. The Underlying Shares to be

issued and delivered to the Investor pursuant to the terms of the Warrants have been duly authorized and will be, upon issuance, validly issued, fully paid and non-assessable, and will be free from all taxes, Liens, preemptive rights and charges with respect to the issue thereof. The Company has reserved from its duly authorized capital stock solely for the benefit of the Investor and its Affiliates the maximum number of shares of Common Stock issuable pursuant to the Investment Agreement and the Warrants. "Registrable Securities" means (i) the Shares, (ii) the Underlying Shares and (iii) any shares of capital stock issued or issuable with respect to any Shares, Warrants or Underlying Shares as a result of any stock split, stock dividend, recapitalization, reclassification, exchange or similar event or otherwise.

3. Shelf Registration Statement; Sales of Shares Under the Registration; Sales Pursuant to Rule 144.

(a) As soon as practicable after the date of this Agreement and after the date on which the Company has completed in a timely manner the filing with the Securities and Exchange Commission (the "SEC") of all reports required to be filed by Section 13 of the Securities Exchange Act of 1934, as amended (the "Exchange Act") during the twelve (12) months immediately preceding such filing(s) as required by the instructions to Form S-3, and following receipt of a written consent of KPMG LLP (or such other independent public registered accounting firm) to include or incorporate by reference in the Shelf Registration Statement its report pertaining to the financial statements of the Company to be filed therewith or included or incorporated by reference therein and to all references to KPMG LLP (or such other independent public registered accounting firm) in the Shelf Registration Statement (it being acknowledged that the obtaining of such consent shall be at the sole expense of the Company and that the Company shall use its commercially reasonable best efforts to obtain such consent, and the Investor shall cooperate in good faith with such efforts, provided, however, that the Investor shall not be required to incur any cost, expense or liability in connection therewith), the Company shall file with the SEC under the Securities Act a shelf registration statement on Form S-3 (or any successor form thereto) (the "Shelf Registration Statement" and, together with any other registration statement of the Company that covers any of the Registrable Securities pursuant to any of the provisions of this Agreement, including any prospectus, amendments (including post-effective amendments) and supplements thereto, all exhibits thereto and all material incorporated by reference therein, each a "Registration Statement") registering all of the Registrable Securities for resale to the public. The Company may include in the Shelf Registration Statement the registration of shares of Common Stock for sale by the Company for its own account ("Primary Shares") but no other shares. The Shelf Registration Statement when declared effective (including the documents incorporated therein by reference) will comply as to form with all applicable requirements of the Securities Act and the Securities Exchange Act of 1934, as amended (the "Exchange Act") and will not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading. The Company shall use commercially reasonable best efforts to (i) cause the Shelf Registration Statement to be declared effective by the SEC as soon as is reasonably practicable following the filing thereof and (ii) keep the Registration Statement continuously effective for the period (the "Effectiveness Period") ending on the earlier of (A) three (3) years following the date the Shelf Registration Statement is declared effective; provided, that if for any portion of such threeyear period the Shelf Registration Statement is not effective, then such three-year requirement for maintaining the effectiveness of the Shelf Registration Statement shall be extended by the length of such interruption(s); and (B) the date on which the Investor no longer owns any Registrable Securities.

- (b) The Investor shall have the right to sell the Registrable Securities under the Shelf Registration Statement (pursuant to an appropriate prospectus or supplement or amendment) at any time during the Effectiveness Period.
- (c) Notwithstanding anything to the contrary herein, in the event the SEC does not permit the Company to register for resale all of the Registrable Securities in the Shelf Registration Statement because of the SEC's application of Rule 415 under the Securities Act, the Company shall register in the Shelf Registration Statement such number of Registrable Securities as is permitted by the SEC. In the event the SEC does not permit the Company to register for resale all of the Registrable Securities in the initial Shelf Registration Statement, the Company shall use its commercially reasonable best efforts to file one or more subsequent Shelf Registration Statements to register for resale the Registrable Securities that were not registered in the initial Shelf Registration Statement as promptly as reasonably practicable and in a manner permitted by the SEC.

4. <u>Demand Registration Rights</u>.

(a) During the Effectiveness Period, until such time that the Investor owns less than one-third (1/3) of the Shares originally issued to Investor pursuant to the Investment Agreement, the Investor may make one (1) written demand (subject to this <u>Section 4(a)</u> and the last sentence of <u>Section 4(b)</u>) of the Company on the form attached hereto as Exhibit A (a "Demand Registration Request") to file a prospectus supplement or amendment, as may be required under the Securities Act, to the Shelf Registration Statement to cover the sale of Registrable Securities through a firm commitment underwritten public offering without regard to volume or other limitations (an "Underwritten Offering"); provided, however, that the Investor may not make a Demand Registration Request for less than that number of Registrable Securities equal to one-third (1/3) of the aggregate number of Shares originally issued to Investor pursuant to the Investment Agreement. If required by the underwriters, the Investor agrees to execute a customary lock-up agreement (for a duration not to exceed 90 days) in connection with an offering of Registrable Securities pursuant to this Section 4(a). The Company may include in the Underwritten Offering contemplated by this section Primary Shares but no other shares. If the Managing Underwriter or Underwriters (as defined below) with respect to an Underwritten Offering pursuant to this Section 4 advise the Company in writing that, in their opinion, the inclusion of the number of Registrable Securities requested to be included by the Investor and/or any Primary Shares to be included creates a risk that the price per share of the securities to be included in the Underwritten Offering will be reduced, then the Company will promptly give the Investor written notice thereof and, upon the election of the Investor the Company will include all such Registrable Securities then held by the Investor that in the opinion of such underwriters can be sold without creating such a risk prior to including any Primary Shares: provided, that if all Registrable Securities that the Investor has requested to be included in such Underwritten Offering pursuant to this Section 4 are not so included, then the Investor will be entitled to an additional Demand Registration Request hereunder on the same terms and conditions as the initial Demand Registration Request. If the Underwritten Offering made pursuant to a Demand Registration Request is not consummated for any reason (other than as a result of a withdrawal

by the Investor in accordance with <u>Section 4(b)</u> that counts as the one Demand Registration Request hereunder), then such Demand Registration Request shall not be deemed to have been made.

(b) Subject to Section 6(d), in the event that the Investor delivers a Demand Registration Request, the Company shall enter into an underwriting agreement in customary form with the underwriter or underwriters (including the book running lead manager of such Underwritten Offering, referred to herein as the "Managing Underwriter"), which shall include, among other provisions, indemnities to the effect and to the extent provided in Section 8, and shall take all such other reasonable actions as are requested by the Managing Underwriter or the Investor in order to expedite or facilitate the registration and disposition of the Registrable Securities subject to the Underwritten Offering. In connection with any Underwritten Offering under this Agreement, the Company shall be entitled to select the Managing Underwriter or Underwriters (which shall be of nationally recognized standing), subject to the written consent of the Investor not to be unreasonably withheld. The Investor shall not be required to make any representations or warranties to or agreements with the Company or the Managing Underwriter other than representations, warranties or agreements regarding the Investor and its ownership of the securities being registered on its behalf and its intended method of distribution and any other representation required by Law. If the Investor disapproves of the terms of an underwriting, the Investor may elect to withdraw therefrom by notice to the Company and the Managing Underwriter; provided, however, that such withdrawal must be made up to and including the time of pricing of such offering to be effective; and *provided*, *further*, that any such withdrawal shall count as the one Demand Registration Request hereunder unless such withdrawal is due to (i) a material and adverse change in the Company's business, condition (financial or otherwise), results of operations, properties, assets, liabilities or prospects or (ii) any action or omission on the part of the Company in connection with the Underwritten Offering, including, without limitation, the Company's failure to comply in any material respect with its obligations hereunder. No such withdrawal or abandonment shall affect the Company's obligation to pay Registration Expenses. Notwithstanding anything to the contrary in this Agreement, in the event that the Investor elects to withdraw from an Underwritten Offering pursuant to this Section 4(b) in circumstances where such withdrawal counts as the one Demand Registration Request hereunder, the Investor shall be entitled to make one additional Demand Registration Request; provided that, at its election, the Investor shall be responsible for all Registration Expenses (as defined below) incurred in connection with the initial or such additional Demand Registration Request.

5. <u>Piggy-Back Registration Rights</u>.

(a) At any time after the date hereof, if the Company decides to undertake any marketing efforts relating to the public offering of Common Stock for its own account or for the account of any other holder of its Common Stock, then the Company will promptly give the Investor written notice thereof and, upon the election of the Investor to participate therein, not less than fifty percent (50%) of the shares of Common Stock being offered in such marketing efforts shall be Registrable Securities requested by the Investor to be included therein; *provided*, that in the event all of the Registrable Securities then held by the Investor constitute less than fifty percent (50%) of the shares of Common Stock being offered in such marketing effort, then the Investor shall be entitled to offer all such Registrable Securities then held by the Investor in

such marketing efforts. The Investor must give its request for inclusion of Registrable Securities in a marketing effort under this Section 5(a) to the Company in writing within ten (10) calendar days after receipt from the Company of notice of such pending marketing effort. The Company shall establish the pricing for the sale of shares of Common Stock (including the Registrable Securities) in any such offering in a commercially reasonable manner; provided that if the Investor disapproves of the terms of offering, the Investor may elect to withdraw therefrom by notice to the Company and the Managing Underwriter; provided, however, that such withdrawal must be made up to and including the time of pricing of such offering to be effective. The parties acknowledge that the Company may in its sole discretion terminate any offering under this Section 5 at any time. No such withdrawal or abandonment shall affect the Company's obligation to pay Registration Expenses.

- (b) This Section 5 shall not apply to any issuance, offering or sale of shares of Common Stock pursuant to any registration statements of the Company on Form S-4 or Form S-8 or their equivalents (relating to equity securities to be issued in connection with an acquisition of any entity or business, an exchange offer, an offering of equity securities issuable in connection with stock option or other employee benefit or director plans) or to registration statements that would otherwise not permit the registration of resales of previously issued securities.
- (c) If the Managing Underwriter or Underwriters with respect to an offering pursuant to this <u>Section 5</u> advise the Company in writing that, in their opinion, the inclusion of the number of Registrable Securities requested to be included by the Investor and/or any other securities to be included therein (including any Primary Shares) creates a risk that the price per share of the securities to be included in such offering will be reduced, then the Company will promptly give the Investor written notice thereof and, upon the election of the Investor the Company will include all such Registrable Securities then held by the Investor that in the opinion of such underwriters can be sold without creating such a risk prior to including any other securities to be included therein (including any Primary Shares).
- (d) The Company shall enter into an underwriting agreement in customary form with the Managing Underwriter, which shall include, among other provisions, indemnities to the effect and to the extent provided in Section 8, and shall take all such other reasonable actions as are requested by the Managing Underwriter or the Investor in order to expedite or facilitate the registration and disposition of the Registrable Securities subject to such offering. In connection with any offering under this Section 5, the Company shall be entitled to select the Managing Underwriter or Underwriters (which shall be of nationally recognized standing), subject to the written consent of the Investor not to be unreasonably withheld. The Investor shall not be required to make any representations or warranties to or agreements with the Company or the Managing Underwriter other than representations, warranties or agreements regarding the Investor and its ownership of the securities being registered on its behalf and its intended method of distribution and any other representation required by Law.

6. Registration Procedures.

- (a) Subject to Section 6(d), in connection with its obligations in this Agreement, the Company will, as expeditiously as possible:
- (i) prepare and file with the Commission such amendments (including post-effective amendments) and supplements to a Registration Statement and any prospectus used in connection therewith as may be necessary, in the case of the Shelf Registration Statement, to keep the Shelf Registration Statement effective at all times during the Effectiveness Period and as may be necessary, in the case of any Registration Statement, to (x) comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such Registration Statement and (y) permit the disposition of the Registrable Securities by the Investor in accordance with its intended method of disposition.
- (ii) furnish to the Investor (A) as far in advance as reasonably practicable (and in any event no less than two (2) Business Days) before filing a Registration Statement, any prospectus used in connection therewith or any supplement or amendment thereto, copies of substantially complete drafts of all such documents proposed to be filed (including furnishing or making available exhibits and each document incorporated by reference therein to the extent then required by the rules and regulations of the SEC), and provide the Investor the opportunity to comment upon or object to any information pertaining to the Investor and/or its plan of distribution that is contained therein and make the corrections reasonably requested by the Investor with respect to such information prior to filing a Registration Statement, prospectus or supplement or amendment thereto, and (B) such number of copies of a Registration Statement and the prospectus included therein and any supplements and amendments thereto as the Investor may reasonably request in order to facilitate the public sale or other disposition of the Registrable Securities covered by such Registration Statement;
- (iii) if applicable, use its commercially reasonable best efforts to register or qualify the Registrable Securities covered by a Registration Statement under the securities or blue sky laws of such jurisdictions as the Investor or, in the case of an Underwritten Offering or an offering contemplated by Section 5, the Managing Underwriter, shall reasonably request, provided that the Company will not be required to qualify generally to transact business in any jurisdiction where it is not then required to so qualify or to take any action which would subject it to general service of process in any such jurisdiction where it is not then so subject;
- (iv) promptly notify the Investor and each underwriter of (A) the filing of a Registration Statement or any prospectus or prospectus supplement to be used in connection therewith, or any amendment or supplement thereto, and, with respect to such Registration Statement, when the same has become effective; (B) in connection with an Underwritten Offering filed pursuant to Section 4 or an offering contemplated by Section 5, any written comments from the SEC with respect to any filing referred to in clause (A) and reasonably cooperate with the Investor in responding thereto; and (C) any written request by the SEC for amendments or supplements to a Registration Statement or any prospectus or prospectus supplement thereto;
- (v) immediately notify the Investor and each underwriter of (A) the happening of any event as a result of which the prospectus or prospectus supplement contained in a Registration Statement, as then in effect, includes an untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein not misleading, in the light of the circumstances then existing; (B) the issuance or threat of issuance by the SEC of any stop order suspending the effectiveness of a Registration

Statement, or the initiation of any proceedings for that purpose; or (C) the receipt by the Company of any notification with respect to the suspension of the qualification of any Registrable Securities for sale under the applicable securities or blue sky laws of any jurisdiction. Following the provision of such notice, the Company agrees to as promptly as practicable amend or supplement the prospectus or prospectus supplement or take other appropriate action so that the prospectus or prospectus supplement does not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading, in the light of the circumstances then existing, and to take such other action as is necessary to remove or prevent a stop order, suspension, threat thereof or proceedings related thereto;

(vi) in connection with any an Underwritten Offering filed pursuant to <u>Section 4</u> or an offering contemplated by <u>Section 5</u>, furnish to the Investor copies of any and all transmittal letters or other correspondence with the SEC or any other governmental agency or self-regulatory body or other body having jurisdiction (including any domestic or foreign securities exchange) relating to such offering of the Registrable Securities;

(vii) in the case of an Underwritten Offering or an offering contemplated by Section 5, furnish upon request and address to the underwriters (with a copy to the Investor), (A) an opinion of counsel for the Company, dated the effective date of the closing under the underwriting agreement, and (B) a "comfort letter", dated the effective date of the applicable Registration Statement or the date of any amendment or supplement thereto and a letter of like kind dated the date of the closing under the underwriting agreement, in each case, signed by the independent public accountants who have certified the Company's financial statements included or incorporated by reference into the applicable Registration Statement, and each of the opinion and the "comfort letter" shall be in customary form and covering substantially the same matters with respect to such Registration Statement (and the prospectus and any prospectus supplement included therein) and as are customarily covered in opinions of issuer's counsel and in accountants' letters delivered to the underwriters in underwritten offerings of securities, and such other matters as such underwriters may reasonably request;

(viii) otherwise use its commercially reasonable best efforts to comply with all applicable rules and regulations of the SEC, and make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months, but not more than eighteen (18) months, beginning with the first full calendar month after the effective date of such Registration Statement, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 promulgated thereunder;

(ix) make available to the appropriate representatives of the Managing Underwriter and the Investor access to such information and personnel of the Company as is reasonable and customary to enable such parties to establish a due diligence defense under the Securities Act; *provided* that the Company need not disclose any information to any such representative of the Investor unless and until such representative and/or the Investor has entered into a customary confidentiality agreement with the Company;

- (x) cause all such Registrable Securities registered pursuant to this Agreement to be listed on each securities exchange or nationally recognized quotation system on which similar securities issued by the Company are then listed;
- (xi) use its commercially reasonable best efforts to cause the Registrable Securities to be registered with or approved by such other governmental agencies or authorities as may be necessary by virtue of the business and operations of the Company to enable the Investor to consummate the disposition of the Registrable Securities;
- (xii) provide a transfer agent and registrar for all Registrable Securities covered by such Registration Statement not later than the effective date of such Registration Statement; and
- (xiii) in the case of an Underwritten Offering or an offering contemplated by <u>Section 5</u>, enter into customary agreements and take such other actions as are reasonably required by the underwriters, including participation in "roadshows," as are reasonably required in order to expedite or facilitate the disposition of the Registrable Securities.
- (b) The Investor, upon receipt of notice from the Company of the happening of any event of the kind described in Section 6(a)(v), shall forthwith discontinue disposition of the Registrable Securities until the Investor's receipt of the copies of the supplemented or amended prospectus contemplated by Section 6(a)(v) or until it is advised in writing by the Company that the use of the prospectus may be resumed, and has received copies of any additional or supplemental filings incorporated by reference in the prospectus, and, if so directed by the Company, the Investor will, or will request the Managing Underwriter or underwriters, if any, to deliver to the Company (at the Company's expense) all copies in their possession or control, other than permanent file copies then in the Investor's possession, of the prospectus covering the Registrable Securities current at the time of receipt of such notice that contain an uncorrected material misstatement or omission.
- (c) If at any time, or from time to time, the Company is not eligible to file or register its shares of Common Stock (including the Registrable Securities) on a shelf registration statement on Form S-3 (or any successor form), the Company will be under no obligation pursuant to Sections 3 through 6 of this Agreement to register with the SEC or otherwise facilitate the disposition of the Registrable Securities pursuant to any other applicable forms of registration statements until such time as the Company is eligible to file or register its shares of Common Stock (including the Registrable Securities) on a shelf registration statement on Form S-3 (or any successor form thereof).
- (d) The Company will be entitled at any time, upon prior written notice to the Investor (a "<u>Suspension Notice</u>") and for a period not to exceed sixty (60) days thereafter (the "<u>Suspension Period</u>"), to suspend the use or effectiveness of any Registration Statement (and the Investor hereby agrees not to offer or sell any Registrable Securities pursuant to such Registration Statement during the Suspension Period) if the Company in its reasonable judgment determines that the offering would interfere, in any material respect, with or require premature public disclosure of any financing, acquisition, corporate reorganization or other significant transaction involving the Company or the Company Subsidiaries. No more than one (1)

Suspension Period shall occur in any twelve (12) month period. In the event that the Company shall exercise its rights hereunder, the Effectiveness Period and the applicable time period during which any applicable Registration Statement is to remain effective shall be extended by a period of time equal to the duration of each such Suspension Period.

- (e) In connection with any offering of the Registrable Securities pursuant to this Agreement:
- (i) the Investor shall furnish such information, and render such cooperation, to the Company and any underwriter as the Company or such underwriter may reasonably request for inclusion in or in connection with, as the case may be, the Shelf Registration Statement (and any supplement or amendment thereto) or any other Registration Statement pursuant to which Registrable Securities are to be offered for sale; and
- (ii) the Investor covenants and agrees that it will comply with the prospectus delivery requirements of the Securities Act as applicable to the Investor in connection with any sales of Registrable Securities pursuant to this Agreement.

Expenses

- (a) The Company will pay all Registration Expenses in connection with any Registration Statement that includes Registrable Securities, whether or not such Registration Statement becomes effective or any sale is made pursuant to such Registration Statement. The Investor shall pay all underwriting fees, discounts and selling commissions allocable to the sale of the Registrable Securities. For the avoidance of doubt, in the event the Investor makes a second Demand Registration Request pursuant to the last sentence of Section 4(b), the Investor will pay all Registration Expenses as and to the extent provided in such last sentence of Section 4(b).
- (b) For purposes of this Agreement, "Registration Expenses" means all expenses incident to the Company's performance under or compliance with this Agreement, including, without limitation, to effect the registration of the Registrable Securities in a Registration Statement and the disposition of such Registrable Securities, including, without limitation, all registration and filing fees, all securities exchange listing fees, all registration, filing, qualification and other fees and expenses of complying with securities or blue sky laws, fees of the Financial Industry Regulatory Authority and fees of transfer agents and registrars, all word processing, duplicating and printing expenses, the fees and disbursements of counsel and independent public accountants for the Company, including the expenses of any special audits or "comfort letters" required by or incident to such performance and compliance, and the fees and disbursements of one counsel for the Investor.

8. <u>Indemnification</u>.

(a) By the Company. In the event of a registration of any Registrable Securities under the Securities Act pursuant to this Agreement, the Company will indemnify and hold harmless the Investor, its directors, officers and partners, each underwriter pursuant to the applicable underwriting agreement with such underwriter, and each Person, if any, who controls the Investor or underwriter within the meaning of the Securities Act or the Exchange Act, against

any and all losses, claims, damages, expenses and liabilities (including reasonable attorneys' fees and expenses) (collectively, "Losses"), to which any such Person may become subject under the Securities Act, the Exchange Act or otherwise, insofar as such Losses (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon any untrue statement or alleged untrue statement of any material fact contained in a Registration Statement, any preliminary prospectus or final prospectus contained therein, or any amendment or supplement thereof, or arise out of or are based upon the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein (in the case of a prospectus, in light of the circumstances under which they were made) not misleading, or any violation or alleged violation by the Company of the Securities Act, the Exchange Act or any other applicable securities law, or any rule or regulation thereunder, and will reimburse the Investor, its directors, officers and partners, each such underwriter and each such controlling Person for any and all legal and other expenses reasonably incurred by any of them in connection with investigating or defending any such Loss or action or proceeding; provided, however, that the Company will not be liable in any such case if and to the extent that any such Loss arises out of or is based upon an untrue statement or alleged untrue statement or omission or alleged omission so made in conformity with information furnished by the Investor, such underwriter or such controlling Person in writing specifically for use in a Registration Statement or any prospectus contained therein or any amendment or supplement thereof. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of the Investor or any such director, officer, partner, underwriter or controlling Person, and shall survive the transfer of such Registrable Se

- (b) By the Investor. The Investor agrees to indemnify and hold harmless the Company, its directors and officers, and each Person, if any, who controls the Company within the meaning of the Securities Act or the Exchange Act to the same extent as the foregoing indemnity from the Company to the Investor solely in connection with any untrue statement or alleged untrue statement or omission or alleged omission, but only with respect to information regarding the Investor furnished in writing by or on behalf of the Investor expressly for inclusion in a Registration Statement or any prospectus contained therein or any amendment or supplement thereof relating to the Registrable Securities; *provided*, *however*, that the liability of the Investor shall not be greater in amount than the dollar amount of the proceeds (net of any selling expenses) received by the Investor from the sale of the Investor giving rise to such indemnification.
- (c) <u>Notice</u>. Promptly after receipt by an indemnified party hereunder of notice of the commencement of any action, such indemnified party shall, if a claim in respect thereof is to be made against the indemnifying party hereunder, notify the indemnifying party in writing thereof, but the omission so to notify the indemnifying party shall not relieve it from any liability which it may have to any indemnified party under this <u>Section 8</u>, except to the extent the indemnifying party is actually and materially prejudiced by such failure to give notice. The indemnifying party shall be entitled to participate in and, to the extent it shall wish, to assume and undertake the defense thereof with counsel reasonably satisfactory to such indemnified party and, after notice from the indemnifying party to such indemnified party of its election so to assume and undertake the defense thereof, the indemnifying party shall not be liable to such indemnified party under this <u>Section 8</u> for any legal expenses subsequently incurred by such indemnified party in connection with the defense thereof; *provided*, *however*, that, (i) if the

indemnifying party has failed to assume the defense and employ counsel or (ii) if the defendants in any such action include both the indemnified party and the indemnified party shall have reasonably concluded, in consultation with counsel, that there are reasonable defenses available to the indemnified party that are different from or additional to those available to the indemnifying party, or if the interests of the indemnified party are deemed, based upon the advice of counsel to the indemnified party, to conflict with the interests of the indemnifying party, then the indemnified party shall have the right to select a separate counsel and to assume such legal defense and otherwise to participate in the defense of such action, with the reasonable expenses and fees of one such separate counsel (firm) and other reasonable expenses related to such participation to be reimbursed by the indemnifying party as incurred. Notwithstanding any other provision of this Agreement, no indemnifying party shall settle any action without the prior written consent of the indemnified party, unless the settlement thereof imposes no liability or obligation on, and includes a complete and unconditional release from all liability of, the indemnified party.

(d) Contribution. If for any reason the foregoing indemnity is unavailable, or is insufficient to hold harmless an indemnified party, then the indemnifying party shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, expenses or liabilities (i) in such proportion as is appropriate to reflect the relative benefits received by the indemnifying party on the one hand and the indemnified party on the other from the registration or (ii) if the allocation provided by clause (i) above is not permitted by applicable Law, or provides a lesser sum to the indemnified party than the amount hereinafter calculated, in such proportion as is appropriate to reflect not only the relative benefits received by the indemnifying party on the one hand and the indemnified party on the other but also the relative fault of the indemnifying party and the indemnified party as well as any other relevant equitable considerations. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by or on behalf of the indemnifying party or the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such untrue statement or omission. The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 8(d) were determined by pro rata allocation or by any other method of allocation that does not take into account the equitable considerations referred to above. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation. Notwithstanding anything to the contrary contained herein, the Investor shall not be required to contribute any amount in excess of the net procee

(e) The indemnity and contribution agreements contained in this <u>Section 8</u> are in addition to any liability that an indemnifying party may have to an indemnified party. The indemnity and contribution agreements herein are in addition to and not in diminution or limitation of any indemnification provisions contained in the Investment Agreement.

9. Term.

This Agreement shall remain in effect for so long as the Investor shall retain any interest in all or any portion of the Registrable Securities; *provided* that the provisions of <u>Sections 8</u> and <u>10</u> shall survive termination of this Agreement for a period of time equal to the expiration of all relevant statutes of limitation.

10. Miscellaneous.

- (a) <u>Assignment</u>. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns; *provided* that no party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the prior written consent of each other party hereto, *provided* that the Investor shall be permitted to assign any of its rights under or the benefits of this Agreement to any Affiliate or Affiliates to whom the Investor may transfer Registrable Securities in compliance with applicable Law and the terms of this Agreement, in which event any such Affiliate shall be deemed to be an Investor for all purposes hereunder.
- (b) <u>Parties in Interest</u>. This Agreement shall be binding upon, inure to the benefit of, and be enforceable by the respective successors and permitted assigns of the parties hereto. Nothing contained herein shall be deemed to confer upon any other person any right or remedy under or by reason of this Agreement, except as otherwise provided in <u>Section 8</u> hereof.
- (c) <u>Amendment and Modification</u>. Any provision of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed, in the case of an amendment, by each party to this Agreement, or in the case of a waiver, by the party against whom the waiver is to be effective.
- (d) Entire Agreement. Except for the Investment Agreement, the Warrants and the Settlement Agreement, this Agreement contains the entire agreement by and between the Company and the Investor with respect to the transactions contemplated by this Agreement and supersedes all prior agreements and representations, written or oral, with respect thereto.
- (e) <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and by the different parties in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Counterparts delivered by electronic transmission shall be deemed to be originally signed counterparts.
- (f) <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- (g) <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party hereto. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in

good faith to modify this Agreement so as to effect the original intent of such parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the extent possible.

- (h) Specific Performance. The Investor and the Company agree that any violation or breach of the provisions contained in this Agreement will result in irreparable injury to the non-breaching party for which a remedy at law would be inadequate and that, in addition to any relief at law that may be available to the non-breaching party for such violation or breach and regardless of any other provision contained in this Agreement, the non-breaching party shall be entitled to such injunctive and other equitable relief as a court may grant, without the requirement to post bond or other security.
- (i) Notices. All notices, requests and other communications to any party hereunder shall be in writing (including facsimile transmission) and shall be given,

if to the Company, to:

ModusLink Global Solutions, Inc. 1601 Trapelo Road Suite 170

Waltham, MA 02451 Attention: Peter L. Gray

Phone: (781) 663-5024 Fax: (781) 663-5095

Email: peter_gray@moduslink.com

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 233 South Wacker Drive, Suite 5800 Chicago, IL 60606

Mark D. Gerstein Attn:

Zachary A. Judd (312) 876-7700

(312) 993-9769 Fax: Email: mark.gerstein@lw.com

zachary.judd@lw.com

if to the Investor, to:

Phone:

c/o Steel Partners II, L.P. 590 Madison Avenue New York, New York 10022 Attention: Warren Lichtenstein Leonard McGill

Tel: 212-520-2300 Fax: 212-520-2337

Email: warren@steelpartners.com len@steelpartners.com

with a copy (which shall not constitute notice) to:

Olshan Frome Wolosky LLP 65 East 55th Street New York, New York 10022 Attention: Steven Wolosky

Tel: 212-451-2300 Fax: 212-451-2222

Email: swolosky@olshanlaw.com mneidell@olshanlaw.com

Michael R. Neidell

or such other address or facsimile number as such party may hereafter specify for the purpose by notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. in the place of receipt and such day is a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed not to have been received until the next succeeding Business Day in the place of receipt.

(j) <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the law of the State of New York, without regard to the choice of law or conflict of law provision or rule that would cause the application of the laws of any jurisdiction other than the State of New York.

(k) <u>Jurisdiction</u>. Each of the parties hereto hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of any Delaware State court, or Federal court of the United States of America, sitting in Delaware, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement or the agreements delivered in connection herewith or the transactions contemplated hereby or for recognition or enforcement of any judgment relating thereto, and each of the parties hereto hereby irrevocably and unconditionally (i) agrees not to commence any such action or proceeding except in such courts, (ii) agrees that any claim in respect of any such action or proceeding may be heard and determined in such Delaware State court or, to the extent permitted by law, in such Federal court, (iii) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such Delaware State or Federal court, and (iv) waives, to the fullest extent permitted by Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such Delaware State or Federal court. Each of the parties hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each party hereto irrevocably consents to service of process in the manner provided for notices in Section 10(i). Nothing in this Agreement will affect the right of any party hereto to serve process in any other manner permitted by Law.

(l) <u>Waiver of Trial by Jury</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY ACTION ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

(m) <u>Adjustments For Stock Dividends, Stock Splits, Recapitalizations, Combinations, Etc.</u> If the Company shall at any time issue a stock dividend on its outstanding shares of Common Stock or effect a recapitalization, stock split, reverse stock split, reorganization, consolidation, split-up, combination, repurchase or exchange of shares of its Common Stock or other securities of the Company that affects the outstanding number of shares of Common Stock, all provisions set forth in this Agreement that are affected by a specified number of shares of Common Stock shall be appropriately adjusted.

[Signature Page Follows]

at above written.	
	MODUSLINK GLOBAL SOLUTIONS, INC.
	By:
	Name:
	Title:
	STEEL PARTNERS HOLDINGS LP
	By: Steel Partners Holdings GP Inc., its general partner
	By:

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective authorized officers as of the day and year

[Signature Page to Registration Rights Agreement]

Name: Title:

EXHIBIT A

Form of Demand Registration Request

CERTIFIED MAIL RETURN RECEIPT REQUESTED

ModusLink Global Solutions, Inc. 1601 Trapelo Road Suite 170 Waltham, MA 02451

Attention: Executive Vice President, General Counsel and Secretary

Ladies and Gentlemen:

Pursuant to that certain Registration Rights Agreement, dated as of , 2013 (the "<u>Registration Rights Agreement</u>"), by and among ModusLink Global Solutions, Inc. (the "<u>Company</u>") and the undersigned, this letter constitutes a Demand Registration Request (as defined in the Registration Rights Agreement) to register for public sale under the Securities Act of 1933, as amended, up to an aggregate of [fill in appropriate number of shares of Common stock so requested for inclusion] shares of the Company's common stock in accordance with Section 4 of the Registration Rights Agreement.

STEEL PARTNERS HOLDINGS L.P.

By:	Steel Partners Holdings GP Inc., its general partner
By:	
	Name:
	Title·

EXHIBIT B

THE SECURITIES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND MAY NOT BE SOLD OR TRANSFERRED UNLESS THERE IS AN EFFECTIVE REGISTRATION STATEMENT UNDER SUCH ACT COVERING SUCH SECURITIES OR THE SECURITIES ARE SOLD AND TRANSFERRED IN A TRANSACTION THAT IS EXEMPT FROM OR NOT SUBJECT TO THE REGISTRATION AND PROSPECTUS DELIVERY REQUIREMENTS OF SUCH ACT. THE SECURITIES EVIDENCED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS SET FORTH IN A SETTLEMENT AGREEMENT BETWEEN THE COMPANY AND HANDY & HARMAN LTD., DATED AS OF FEBRUARY 11, 2013. A COPY OF SUCH SETTLEMENT AGREEMENT MAY BE OBTAINED FROM THE COMPANY UPON REQUEST.

FORM OF WARRANT

TO PURCHASE SHARES OF COMMON STOCK

Λf

MODUSLINK GLOBAL SOLUTIONS, INC.

A Delaware Corporation

Warrant No.
Number of Shares:

Date: [] (the "Effective Date")

VOID AFTER [

]1

This certifies that, for value received, Steel Partners Holdings L.P., a Delaware limited partnership (the "Holder") is the registered holder of this Warrant and is hereby entitled to purchase from ModusLink Global Solutions, Inc., a Delaware corporation (the "Company"), subject to the terms and conditions set forth herein, two million (2,000,000) fully paid and nonassessable shares of common stock, par value \$0.01 per share (the "Common Stock"), of the Company, along with the associated right to acquire Series A Preferred Stock and Series B Preferred Stock issued under the Tax Plan and the Rights Plan, respectively (the "Warrant Shares") at the initial exercise price ("Warrant Price") of \$5.00 per share at any time and from time to time, until the Termination Date (as defined in Section 5 below). The Warrant Price and the number of Warrant Shares that may be purchased upon exercise of this Warrant are subject to adjustment pursuant to Section 4 hereof. This Warrant has been issued pursuant to the terms of that certain Investment Agreement, dated February 11, 2013 between the Company and the Holder (the "Investment Agreement"). For purposes of this Warrant the following terms will have the following meaning: (i) "Tax Plan" means the Tax Benefit Preservation Plan, dated as of October 17, 2011, between the Company and American Stock Transfer & Trust Company, LLC,

Five years following the Effective Date.

as Rights Agent (as amended by Amendment No. 1 to Tax Benefit Preservation Plan, dated as of March 21, 2012, by and between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent); (ii) "Tax Plan Rights" means the associated preferred share purchase rights issued pursuant to the Tax Plan; (iii) "Series A Preferred Stock" means the Series A Junior Participating Preferred Stock, par value \$0.01 per share, reserved for issuance in connection with the exercise of Tax Plan Rights issued pursuant to the Tax Plan; (iv) the "Series B Preferred Stock" means the Series B Junior Participating Preferred Stock, par value \$0.01 per share, reserved for issuance in connection with the exercise of Rights Plan Rights issued pursuant to the Rights Plan; (v) the "Rights Plan" means the Rights Agreement, dated as of March 21, 2012, between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent; and (vi) the "Rights Plan Rights" means the associated preferred share purchase rights issued pursuant to the Rights Plan.

1. Exercise.

(a) <u>Cash Exercise</u>. Subject to the terms and conditions hereof, this Warrant may be exercised by the Holder at any time or from time to time on or after the Effective Date and on or prior to the Termination Date, in whole or in part, upon delivery of a written notice in the form attached hereto as Exhibit A (the "<u>Exercise Notice</u>") to the Company during normal business hours on any Business Day at the Company's principal executive offices (or such other office or agency of the Company as it may designate by notice to the holder hereof) together with delivery of payment by (i) cash, (ii) wire transfer of immediately available funds to a bank account that shall have been specified in writing for such purpose by the Company, (iii) a certified or bank check payable to the Company, or (iv) a combination of the foregoing, for the aggregate Warrant Price of the Warrant Shares so purchased.

(b) <u>Cashless Exercise</u>. In lieu of exercising this Warrant on a cash basis pursuant to Section 1(a) hereof, subject to the terms and conditions hereof, the Holder may elect instead to exercise this Warrant at any time or from time to time, on or after the Effective Date and on or prior to the Termination Date, in whole or in part, on a cashless basis to receive upon such exercise the "<u>Net Number</u>" of shares of Common Stock determined according to the following formula (a "<u>Cashless Exercise</u>"):

Net Number =
$$\underbrace{(X \times Y) - (X \times Z)}_{Y}$$

For purposes of the foregoing formula:

X = the total number of Warrant Shares with respect to which this Warrant is then being exercised.

Y = the Market Price of the shares of Common Stock as of the Trading Day immediately preceding the date of receipt of the Exercise Notice by the Company.

Z = the Warrant Price then in effect for the applicable Warrant Shares at the time of such exercise.

To effect a Cashless Exercise, the Holder shall deliver to the Company during normal business hours on any Business Day at the Company's principal executive offices (or such other office or agency of the Company as it may designate by notice to the holder hereof) the Exercise Notice indicating the Holder's election to exercise this Warrant on a cashless basis.

- (c) <u>Surrender of Warrant Upon Exercise</u>. The Holder shall not be required to surrender this Warrant in order to effect an exercise hereunder, unless the Holder has purchased all of the Warrant Shares available hereunder and the Warrant has been exercised in full, in which case, the Holder shall surrender this Warrant to the Company at the address set forth in Section 10 for cancellation as soon as practicable after the date the final Exercise Notice is delivered to the Company. Execution and delivery of the Exercise Notice with respect to less than all of the Warrant Shares shall have the same effect as cancellation of the original Warrant and issuance of a new Warrant evidencing the right to purchase the remaining number of Warrant Shares.
- (d) Market Price; Business Day; Trading Day. For purposes of this Section 1, the "Market Price" of the shares of the Common Stock on any particular date shall have the following meaning: (i) if the Common Stock is listed for trading on a national securities exchange, the last sale price of the Common Stock reported by such exchange on that date; (ii) if the Common Stock is not listed on any such exchange and the Common Stock is traded in the over-the-counter market, the last price reported by the OTC Bulletin Board on that date; (iii) if the Common Stock is not listed on any such exchange or quoted on the OTC Bulletin Board, then the last price quoted on such date in the over-the-counter market as reported by the National Quotation Bureau Incorporated (or any similar organization or agency succeeding its functions of reporting prices) on that date; or (iv) if none of clauses (i)-(iii) are applicable, then the fair market value of the Common Stock as determined, in good faith, by the board of directors of the Company. As used herein, "Business Day" means any day other than (i) a Saturday, (ii) a Sunday, (iii) any day on which commercial banks in New York, New York are required or authorized to close by law or executive order, and (iv) the Friday after Thanksgiving Day. As used herein, "Trading Day" means any day on which the Common Stock is traded on the principal securities exchange or trading market where such security is then listed or traded; provided that "Trading Day" shall not include any day on which the Common Stock is scheduled to trade on such exchange or market for less than 4.5 hours or any day that the Common Stock is suspended from trading during the final hour of trading on such exchange or market (or if such exchange or market does not designate in advance the closing time of trading on such exchange or market, then during the hour ending at 4:00:00 p.m., New York time).
- (e) <u>Issuance of Warrant Shares Upon Exercise</u>. Upon exercise of this Warrant, the Company shall within five (5) Business Days issue and deliver to the Holder (i) a certificate or certificates for the total number of Warrant Shares (or the Net Number of Warrant Shares, as applicable) for which this Warrant is being exercised in the name of such Holder or its designee and (ii) if this Warrant is exercised with respect to fewer than all of the Warrant Shares represented by this Warrant, a new Warrant representing the number of Warrant Shares in respect of which this Warrant has not been exercised. The Warrant Shares will be deemed to have been issued, and the person in whose name any certificate representing Warrant Shares will be issuable upon the exercise of this Warrant (as indicated in the Exercise Notice) will be deemed to have become the holder of record of (and will be treated for all purposes as the record holder of) the Warrant Shares represented thereby, immediately prior to the close of business on

the Business Day upon which this Warrant is exercised in accordance with the terms hereof. The issuance of certificates for Warrant Shares upon the exercise of this Warrant will be made without charge to the Holder for any issuance tax in respect thereof; provided, however, that the Company will not be required to pay any tax which may be payable in respect of any transfer involved in the issuance and delivery of any certificate in a name other than that of the Holder.

- (f) <u>Fractional Shares</u>. Upon the exercise of the rights represented by this Warrant, the Company will not be obligated to issue fractional shares of Common Stock, and in lieu thereof, the Company will pay to the Holder an amount in cash equal to the Market Price per share of Common Stock as of the Business Day immediately preceding the date of receipt of the Exercise Notice by the Company multiplied by such fraction (rounded to the nearest cent).
- (g) Notice of Certain Events. The Company will provide the Holder with written notice at least ten (10) days prior to the date on which the Company closes its books or takes a record date (i) with respect to any dividend or distribution of any cash, assets, securities or other property or rights upon the Common Stock or (ii) for determining rights to vote with respect to any liquidation, dissolution, reclassification, merger, consolidation, sale or conveyance to another entity of all or substantially all of the property of the Company, in each case which is effected in such a way that the holders of Common Stock are entitled to receive (either directly or upon subsequent liquidation) cash, stock, securities or assets with respect to or in exchange for Common Stock.
- (h) <u>Holder's Representations and Warranties</u>. Each exercise hereof shall constitute an affirmation by the Holder that the representations and warranties contained in Article 4 of the Investment Agreement are true and correct in all material respects with respect to the Holder as of the time of such exercise.
- 2. <u>Reservation of Common Stock</u>. The Company represents and warrants that it will at all times reserve and keep available, solely for issuance upon the exercise of this Warrant, such number of authorized but unissued shares of Common Stock as will be sufficient to permit the exercise in full of this Warrant.
- 3. Transfer of Warrant. This Warrant may only be transferred or assigned by the Holder in accordance with that certain Settlement Agreement, dated as of the date hereof, by and between the Company and the Holder. Subject to the restrictions therein, the Company shall transfer this Warrant, in whole or in part, from time to time upon the books to be maintained by the Company for that purpose, upon surrender thereof for transfer properly endorsed or accompanied by appropriate instructions for transfer and such other documents as may be reasonably required by the Company, including, if required by the Company, an opinion of its counsel to the effect that such transfer is exempt from the registration requirements of the Securities Act and applicable state securities laws, and a new Warrant representing the right to purchase the number of Warrant Shares being transferred shall be issued to the transferred, a new Warrant representing the right to purchase the number of Warrant Shares not being transferred shall be issued to the Holder.

- 4. <u>Adjustments</u>. The number of Warrant Shares issuable upon the exercise of this Warrant and the Warrant Price shall be adjusted upon the occurrence of certain events, as follows:
- (a) <u>Stock Splits, Capital Stock Dividends, Combinations and Consolidations of Common Stock</u>. In the event of a stock split, capital stock dividend, subdivision or other similar event of or in respect of the outstanding shares of Common Stock occurring at any time after the date hereof, the number of Warrant Shares issuable upon exercise of this Warrant immediately prior to such stock split, capital stock dividend, subdivision or other similar event will be proportionately increased and the Warrant Price then in effect will be proportionately decreased, effective at the close of business on the date of such stock split, capital stock dividend, subdivision or other similar event, as the case may be. Conversely, in the event of a reverse stock split, combination, consolidation or other similar event of or in respect of the outstanding shares of Common Stock, the number of Warrant Shares issuable upon the exercise of this Warrant immediately prior to such reverse stock split, combination, consolidation or other similar event will be proportionately decreased and the Warrant Price will be proportionately increased, effective at the close of business on the date of such reverse stock split, combination, consolidation or other similar event, as the case may be.
- (b) Reclassifications. In the event of a reclassification or recapitalization of the Common Stock of the Company (other than as a result of a subdivision, stock dividend, stock split, or other event for which an adjustment is made under Section 4(a)) occurring at any time after the date hereof, provision shall be made so that the Holder will have the right to exercise this Warrant, and the Holder shall thereafter be entitled to receive upon exercise, in lieu of the shares of Common Stock previously issuable upon exercise of this Warrant, the same number and type of shares of stock or other securities, cash, assets or property to which a holder of the number of shares of Common Stock deliverable upon exercise of this Warrant immediately prior to such reclassification or recapitalization would have been entitled to receive in such reclassification or recapitalization. In any such case, appropriate adjustment shall be made in the application of the provisions of this Section 4 with respect to the rights of the Holder after the reclassification or recapitalization to the end that the provisions of this Section 4 (including adjustment of the Warrant Price then in effect and the number of Warrant Shares issuable upon exercise of this Warrant) shall be applicable after that event and be as nearly equivalent as practicable. The provisions of this Section 4(b) shall similarly apply to successive reclassifications and recapitalizations of Common Stock.
- (c) <u>Mergers or Consolidations</u>. In the event of, directly or indirectly, in one or a series of transactions, (i) the merger or consolidation of the Company with or into another corporation or another entity or person, or other similar business combination, or (ii) a sale or conveyance to another entity or person of all or substantially all of the property or assets of the Company, in any such case such that holders of Common Stock shall be entitled to receive (either directly or upon subsequent liquidation) stock or other securities, cash, assets or property with respect to or in exchange for the Common Stock, as a part of such merger, consolidation, business combination, sale or conveyance, provision shall be made so that the Holder will have the right to exercise this Warrant, and the Holder shall thereafter be entitled to receive upon exercise, the same number and type of shares of stock or other securities, cash, assets or property to which a holder of the number of shares of Common Stock deliverable upon exercise of this

Warrant immediately prior to such merger, consolidation, business combination, sale or conveyance would have been entitled to receive in such merger, consolidation, business combination, sale or conveyance. In any such case, appropriate adjustment shall be made in the application of the provisions of this Section 4 with respect to the rights of the Holder after the merger, consolidation, business combination, sale or conveyance to the end that the provisions of this Section 4 (including adjustment of the Warrant Price then in effect and the number of Warrant Shares issuable upon exercise of this Warrant) shall be applicable after that event and be as nearly equivalent as practicable. The provisions of this Section 4(c) shall similarly apply to successive mergers, consolidations, business combinations, sales or conveyances.

(d) Rights Upon Distribution Of Assets. If the Company shall declare or make any dividend or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property or options by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "Distribution"), at any time after the date hereof, then, in each such case, the Warrant Price in effect immediately prior to the close of business on the record date fixed for the determination of holders of shares of Common Stock entitled to receive the Distribution shall be reduced, effective as of the close of business on such record date, to a price determined by multiplying such Warrant Price by a fraction of which (i) the numerator shall be the Weighted Average Price of the shares of Common Stock on the Trading Day immediately preceding such record date minus the value of the Distribution (as determined in good faith by the Company's Board of Directors) applicable to one share of Common Stock, and (ii) the denominator shall be the Weighted Average Price of the shares of Common Stock on the Trading Day immediately preceding such record date. As used herein, "Weighted Average Price" means, for any security as of any date, the dollar volume-weighted average price for such security on the principal securities exchange or trading market where such security is then listed or traded during the period beginning at 9:30:01 a.m., New York time (or such other time as the principal securities exchange or trading market where such security is then listed or traded publicly announces is the official open of trading), and ending at 4:00:00 p.m., New York time (or such other time as the principal securities exchange or trading market where such security is then listed or traded publicly announces is the official close of trading), as reported by Bloomberg through its "Volume at Price" function or, if the foregoing does not apply, the dollar volume-weighted average price of such security in the over-the-counter market on the electronic bulletin board for such security during the period beginning at 9:30:01 a.m., New York time (or such other time as the over-the-counter market where such security is then listed or traded publicly announces is the official open of trading), and ending at 4:00:00 p.m., New York time (or such other time as the over-thecounter market where such security is then listed or traded publicly announces is the official close of trading), as reported by Bloomberg, or, if no dollar volumeweighted average price is reported for such security by Bloomberg for such hours, the average of the highest closing bid price and the lowest closing ask price of any of the market makers for such security as reported in the OTC Link or "pink sheets" by OTC Markets Group Inc. (formerly Pink OTC Markets Inc.). If the Weighted Average Price cannot be calculated for a security on a particular date on any of the foregoing bases, the Weighted Average Price of such security on such date shall be the fair market value as determined in good faith by the Board of Directors of the Company. All such determinations shall be appropriately adjusted for any stock dividend, stock split, stock combination or other similar transaction during the applicable calculation period.

- (e) <u>Certificate as to Adjustments</u>. Upon the occurrence of each adjustment (or readjustment) pursuant to this Section 4, the Company at its expense shall promptly compute such adjustment or readjustment in accordance with the terms hereof and furnish to the Holder a certificate setting forth such adjustment or readjustment and showing in reasonable detail the facts upon which such adjustment or readjustment is based. The Company shall, upon the written request at any time of the Holder, furnish or cause to be furnished to the Holder a like certificate setting forth the Warrant Price of this Warrant at the time in effect, and the number of shares of Common Stock and the amount, if any, of other property which at the time would be received upon the exercise of this Warrant.
- 5. <u>Termination Date</u>. This Warrant may not be exercised after 5:00 p.m., New York time, on []² (the "<u>Termination Date</u>") and all rights of the registered holder of this Warrant evidenced hereby shall cease after 5:00 P.M., New York time, on the Termination Date.
- 6. <u>Noncircumvention</u>. The Company hereby covenants and agrees that the Company will not, by amendment of its certificate of incorporation, bylaws or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issue or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, and will at all times in good faith carry out all the provisions of this Warrant and take all action as may be required to protect the rights of the Holder.
- 7. Governing Law; Jurisdiction; Venue; Process. This Warrant shall be governed by and construed in accordance with the law of the State of New York without regard to any choice of law or conflict of law provision or rule that would cause the application of the laws of any jurisdiction other than the State of New York. The Company and the Holder each hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of any Delaware State court, or Federal court of the United States of America, sitting in Delaware, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Warrant or for recognition or enforcement of any judgment relating thereto, and the Company and the Holder each hereby irrevocably and unconditionally (i) agrees not to commence any such action or proceeding except in such courts, (ii) agrees that any claim in respect of any such action or proceeding may be heard and determined in such Delaware State court or, to the extent permitted by law, in such Federal court, (iii) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such Delaware State or Federal court. The Company and the Holder each agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Each of the Company and the Holder irrevocably consents to service of process in the manner provided for notices in Section 10. Nothing in this Warrant will affect the right of the Company or the Holder to serve process in any other manner permitted by law.

Five years following the Effective Date.

- 8. <u>Waiver and Amendment</u>. Any term or provision of this Warrant may be waived at any time by the party entitled to the benefits thereof and any term or provision of this Warrant may be amended or supplemented at any time by agreement of the Company and the Holder. Any waiver of any term or condition, or any amendment or supplementation, of this Warrant must be in writing. A waiver of any breach or failure to enforce any term or condition of this Warrant shall not in any way affect, limit, or waive any party's rights hereunder at any time to enforce strict compliance thereafter with any term or condition of this Warrant.
- 9. <u>Severability</u>. In the event that any one or more of the provisions contained in this Warrant shall be determined to be invalid, illegal or unenforceable in any respect for any reason, the validity, legality, and enforceability of any such provision in any other respect and the remaining provisions of this Warrant shall not be in any way impaired.
- 10. <u>Notice</u>. Any notices or other communications required or permitted under, or otherwise given in connection with, this Warrant shall be in writing and shall be deemed to have been duly given (a) when delivered or sent if delivered in Person or sent by facsimile transmission (provided confirmation of facsimile transmission is obtained), (b) on the fifth (5th) Business Day after dispatch by registered or certified mail, postage prepaid, return receipt requested, (c) on the next Business Day if transmitted by national overnight courier or (d) on the date delivered if sent by email (provided confirmation of email receipt is obtained from the recipient and if followed by transmittal by national overnight courier or hand for delivery on the next Business Day). All notices shall be addressed as follows: if to the Holder or the Company, at their respective addresses as follows, or at such other address as the Holder or the Company may designate by ten days' advance written notice to the other:

if to the Company to:

ModusLink Global Solutions, Inc. 1601 Trapelo Road Suite 170 Waltham, MA 02451

Attn: Peter L. Gray Phone: (781) 663-5024 Fax: (781) 663-5095

Email: peter_gray@moduslink.com

with a copy to:

Latham & Watkins LLP 233 South Wacker Drive, Suite 5800 Chicago, IL 60606

Attn: Mark D. Gerstein

Zachary A. Judd

Phone: (312) 876-7700 Fax: (312) 993-9769 Email: mark.gerstein@lw.com

zachary.judd@lw.com

if to the Holder, to:

c/o Steel Partners II, L.P. 590 Madison Avenue New York, New York 10022 Attention: Warren Lichtenstein

Leonard McGill

Phone: (212) 520-2300 Fax: (212) 520-2337

Email: warren@steelpartners.com len@steelpartners.com

with a copy to:

Olshan Frome Wolosky LLP 65 East 55th Street New York, New York 10022 Attention: Steven Wolosky

Michael R. Neidell

Phone: 212-451-2300 Fax: 212-451-2222

Email: swolosky@olshanlaw.com mneidell@olshanlaw.com

11. Holder of Warrant not a Stockholder. Except as otherwise expressly provided herein, the Holder, solely in its capacity as a holder of this Warrant and with respect to the Warrant Shares, will not be entitled to vote on matters submitted for the approval or consent of the stockholders of the Company or to receive dividends or other distributions declared on or in respect of shares of Common Stock, or otherwise be deemed to be the holder of Common Stock or any other capital stock or other securities of the Company which may at any time be issuable upon the exercise of this Warrant for any purpose, nor will anything contained herein, except as otherwise expressly provided herein, be construed to confer upon the Holder, solely in its capacity as a holder of this Warrant and with respect to the Warrant Shares, any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter submitted for the approval or consent of the stockholders, or to give or withhold consent to any corporate action (whether upon any recapitalization, issuance of stock, reclassification of stock, merger or consolidation, conveyance, or otherwise) or to receive notice of meetings, or to receive dividends or subscription rights or otherwise, in each case, until this Warrant shall have been exercised and the Warrant Shares issuable upon the exercise of this Warrant will have become issuable as provided herein.

12. Loss, Destruction, Etc., of Warrant. In the event this Warrant is lost, stolen, destroyed, or mutilated, the Company will issue to the Holder a new Warrant, of like tenor, in lieu of such lost, stolen, destroyed, or mutilated Warrant, upon the Company's receipt of, (i) in the case of a lost, stolen, or destroyed Warrant, an affidavit reasonably satisfactory to the Company from the Holder to that effect and indemnity or security reasonably satisfactory to the Company, and (ii) in the case of a mutilated Warrant, the mutilated Warrant.

IN WITNESS WHEREOF, the Company has caused this Wa	rrant to be signed by it	s duly authorized officers on this	day of , 20)13.
By: Name: Title: President	By: Name: Title:	Secretary		
ACCEPTED AND AGREED,				
STEEL PARTNERS HOLDINGS L.P.				
By: Steel Partners Holdings GP Inc., its general partner				
By:				
Name: Title:				

EXHIBIT A

EXERCISE NOTICE

TO BE EXECUTED BY THE REGISTERED HOLDER TO EXERCISE THIS WARRANT TO PURCHASE COMMON STOCK

MODUSLINK GLOBAL SOLUTIONS, INC.

The undersigned holder hereby exercises the right to purchase of the shares of Common Stock ("<u>Warrant Shares</u>") of ModusLink Global Solutions, Inc., a Delaware corporation (the "<u>Company</u>"), evidenced by the attached Warrant to Purchase Common Stock (the "<u>Warrant</u>"). Capitalized terms used herein and not otherwise defined shall have the respective meanings set forth in the Warrant.

	1. Form of Warrant Price. The Holder intends that payment of the Warrant Price shall be made as:						
		a "Cash Exercise" with respect to	Warrant Shares; and/or				
		a "Cashless Exercise" with respect to	Warrant Shares.				
here		<u>syment of Warrant Price</u> . In the event that the holder shall pay the aggregate Warrant Price i		Exercise with respect to some or all of the Warrant Shares to be issued pursuant e Company in accordance with the terms of the Warrant.			
	3. <u>De</u>	elivery of Warrant Shares. The Company shall	deliver to the holder	Warrant Shares in accordance with the terms of the Warrant.			
Date	:	,					
		Name of Registered Holder					
By:	N.						
	Name Title:						

SETTLEMENT AGREEMENT

THIS SETTLEMENT AGREEMENT, dated as of February 11, 2013 (this "<u>Agreement</u>"), is by and between ModusLink Global Solutions, Inc., a Delaware corporation (the "<u>Company</u>") and Handy & Harman Ltd., a Delaware corporation ("<u>HNH</u>" and together with the parties listed on <u>Exhibit A</u> hereto, the "<u>Stockholder</u>"). Each of the Company and the Stockholder is referred to herein as a "<u>Party</u>" and collectively as the "<u>Parties</u>".

WHEREAS, the Stockholder beneficially owns the number of shares of the Company's common stock, par value \$0.01 per share (the "Common Stock") listed on Exhibit A hereto;

WHEREAS, on September 28, 2012, HNH delivered a letter to the Company expressing an intention to nominate director candidates (the "Nomination Letter") for election to the Company's Board of Directors (the "Board") at the 2012 annual meeting of the stockholders of the Company (including any adjournment or postponement thereof or any special meeting held in lieu thereof, the "Stockholder Meeting");

WHEREAS, the Company filed a Preliminary Proxy Statement filed with the Securities and Exchange Commission (the "SEC") on December 21, 2012 and amended prior to the date hereof (the "Preliminary Proxy Statement");

WHEREAS, the Stockholder filed a Preliminary Proxy Statement filed with the SEC on December 27, 2012 pertaining to HNH's nomination of its nominees to the Board and the other matters set forth in the Preliminary Proxy Statement (as amended prior to the date hereof, the "Stockholder Preliminary Proxy Statement"); and

WHEREAS, the Company and the Stockholder have reached an agreement with respect to certain matters related to the Stockholder Meeting, including the Nomination Letter, and certain other matters, as provided in this Agreement.

NOW, THEREFORE, in consideration of the premises and mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby agree as follows:

Section 1. Definitions.

Unless otherwise defined herein, for purposes of this Agreement, the following terms will have the meanings set forth below:

"Affiliate" of any Person means any Person that directly or indirectly controls, or is under common control with, or is controlled by, such Person. As used in this definition, "control" (including with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person (whether through ownership of securities or partnership, member or other ownership interests, by contract or otherwise). For purposes of this Agreement, Steel Partners Holdings L.P. shall be deemed an Affiliate of each Stockholder.

- "Company Bylaws" means the Third Amended and Restated Bylaws of the Company, as currently in effect.
- "Company Certificate" means the Restated Certificate of Incorporation of the Company.
- "Continuing Directors Committee" means the Continuing Directors Committee of the Board, which shall be formed as soon as practicable following the Stockholder Meeting, comprised of the Continuing Directors (as defined below) and designated such powers to carry out the rights and duties of such committee described in this Agreement.
- "Equity Interest" means any share of capital stock, partnership, member or similar equity interests in any Person, and any securities (including debt securities) convertible into, or exchangeable or exercisable for, any such shares, capital stock, partnership, member or similar equity interests, or any options, warrants or other rights of any kind to acquire or that are linked to the value of any such shares, capital stock, partnership, member or similar equity interests or such convertible or exchangeable securities, or any other ownership interest (including, without limitation, any such interest represented by Contract right), of such Person.
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Investment Agreement" means that certain Investment Agreement, dated as of the date hereof, by and between the Company and Steel Partners Holdings L.P. ("SPH").
 - "Nominating Committee" means the Nominating and Corporate Governance Committee of the Board or a duly constituted subcommittee thereof.
- "Person" means an individual, corporation, limited liability company, partnership, association, trust, unincorporated organization, other entity or group (as defined in Section 13(d) of the Exchange Act).
 - "Preferred Stock" means the preferred stock, par value \$0.01 per share, of the Company.
- "Shares" means and includes any securities of the Company the holders of which are entitled to vote for members of the Board, including, without limitation, all shares of Common Stock and all shares of Preferred Stock, by whatever name called, now owned or subsequently acquired by the Stockholder or any of its Affiliates, however acquired, whether through any share division or split, share consolidation, stock/share dividend or equivalent under Delaware law, combination, recapitalization or similar events, exercise of warrants or otherwise.
- "Transfer" means any direct or indirect sale, exchange, transfer, hypothecation, gift, bequest, conveyance in trust, pledge, mortgage, grant of a security interest in, assignment, encumbrance, or other disposition of all or any portion of the Shares, including by merger, consolidation, liquidation, dissolution, dividend, distribution or otherwise.

Section 2. Board of Directors Matters.

(a) Size and Composition of the Board.

- (i) The Board, by resolution, shall fix the size of the Board at seven (7) members immediately following the Stockholder Meeting. Warren G. Lichtenstein and Glen M. Kassan (the "<u>Stockholder Nominees</u>") shall be nominated for election by the Company as the Board's only candidates for Class I Directors (as defined in the Company Bylaws) at the Stockholder Meeting. The Board has determined that each of the Stockholder Nominees (A) conforms with the Company's director qualifications and (B) is an "independent director" as defined by Rule 5605(a)(2) of the Nasdaq Marketplace Rules.
- (ii) At the Stockholder Meeting, the Board shall (A) nominate the Stockholder Nominees for election as Class I Directors and will recommend a vote for the Stockholder Nominees (and include such recommendation in the Company's proxy statement) and support and solicit proxies from the stockholders of the Company for the election of the Stockholder Nominees at the Stockholder Meeting as such, and (B) submit to the stockholders of the Company, recommend for their approval (and include such recommendation in the Company's proxy statement) and support and solicit proxies for (i) an amendment to the Company Certificate which shall provide for the destaggering of the Board and the annual election of all directors (the "Destaggering Proposal"), with the first of such annual elections to take place at the next annual meeting following the Stockholder Meeting, and (ii) the issuance of the shares of Common Stock, the Warrants (as defined in the Investment Agreement) and the shares of Common Stock underlying the Warrants to SPH pursuant to the Investment Agreement in accordance with Nasdaq Marketplace Rule 5635(b).
- (iii) Provided that Mr. Lichtenstein is elected to the Board at the Stockholder Meeting, immediately following the issuance of the shares of Common Stock and the Warrants upon consummation of the Investment Agreement as contemplated therein, the Board shall appoint Mr. Lichtenstein as Chairman of the Board in accordance with the Company Bylaws. Subject to and effective immediately following the issuance of the shares of Common Stock and the Warrants upon consummation of the Investment Agreement as contemplated therein, (A) Edward E. Lucente and Joseph M. O'Donnell have agreed to resign from the Board and (B) the Board has appointed Francis J. Jules and Michael J. Mardy to fill such vacancies in accordance with the Company Bylaws as Class II Directors.
- (iv) For a period terminating ten (10) days preceding the opening of the advance notice period for nominations of directors for the annual meeting of stockholders of the Company for the fiscal year ending July 31, 2013 (the "Settlement Agreement Period"), the Company agrees to take all actions necessary, including, but not limited to, the filing of proxy materials, the calling of annual or special meetings, the solicitation of proxies, the removal of directors (other than the Stockholder Nominees) and the filling of vacancies on the Board, and the Stockholder agrees to waive notice, attend meetings, execute written actions, and vote of any and all of such Stockholder's and its Affiliates' Shares, in each case, so as to ensure that the size of the Board shall consist of no more than seven (7) directors.

(v) The remaining five (5) directors shall be Virginia G. Breen, and, subject to their appointment in accordance with Section 2(a)(iii), Francis J. Jules and Michael J. Mardy as Class II Directors and Jeffrey J. Fenton and Jeffrey S. Wald as Class III Directors (such directors, together with any persons (other than the Stockholder Nominees) the Nominating Committee may instead recommend for election at any special or annual meeting of the stockholders or to fill any vacancy with respect thereto, the "Continuing Directors"). Thereafter the Continuing Directors shall be designated by the Continuing Director Committee, including in the event of any vacancy created by the removal, inability to serve or resignation of any Continuing Director. The Company has obtained the agreement of each director of the Company who will be on the Board at the conclusion of the Stockholder Meeting to stand for re-election at the Company's annual meeting of stockholders for the fiscal year ending July 31, 2013 if the Destaggering Proposal is approved at the Stockholder Meeting.

(vi) During the Settlement Agreement Period, in the event of a vacancy created by (1) the removal of either of the Stockholder Nominees (other than pursuant to Section 7 hereof), (2) the inability of either of the Stockholder Nominees to serve or (3) the resignation of either of the Stockholder Nominees, the Nominating Committee shall recommend and the Board shall appoint a replacement nominee designated by the Stockholder; provided, that such director (A) conforms with the Company's director qualifications and (B) is an "independent director" as defined by Rule 5605(a)(2) of the Nasdaq Marketplace Rules; provided, further, that if at any time during the Settlement Agreement Period the Stockholder and its Affiliates Transfer (other than to an Affiliate) (A) more than 4,900,000 Shares in the aggregate of the Shares owned by the Stockholder and its Affiliates immediately after the conclusion of the Stockholder Meeting (including any Shares to be acquired pursuant to the Investment Agreement), the rights set forth in this Section 2(a)(vi) shall only apply to the director seat to be held by Mr. Lichtenstein and (B) more than 9,100,000 Shares in the aggregate of the Shares owned by the Stockholder and its Affiliates immediately after the conclusion of the Stockholder Meeting (including any Shares to be acquired pursuant to the Investment Agreement), the rights set forth in this Section 2(a)(vi) shall expire and be of no further force and effect. References in this Agreement to the Stockholder Nominees shall include their replacements. Such share thresholds shall be adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction occurring during the Settlement Agreement Period.

(b) Withdrawal of Nomination Letter; No Contests or Participation. Effective as of the date hereof, the Stockholder hereby irrevocably withdraws the Nomination Letter and the Stockholder Preliminary Proxy Statement; provided, however, that the Nomination Letter shall be reinstated and shall be deemed to be valid under the Company Bylaws in the event the Investment Agreement is terminated pursuant to Section 7.1(d) or (e) thereof. Until the earlier of (i) expiration or termination of the Settlement Agreement Period or (ii) fifteen (15) days after written notice of a material breach of any of the covenants, obligations or agreements by the Company under this Agreement, the Investment Agreement, the Registration Rights Agreement (as defined in the Investment Agreement) or the Warrants is provided to the Company by the Stockholder (unless such breach is cured within such fifteen (15) day period) (such date the "Standstill Termination Date"), the Stockholder shall not, and shall cause its Affiliates not to (A) make, or be a participant in, any "solicitation" of proxies to vote or consent, or seek to advise or influence any person with respect to the voting of, any securities of the Company, submit,

directly or indirectly, any nomination of directors pursuant to Rule 14a-11 under the Exchange Act or be or become a "participant" in any "election contest" with respect to the Company (all within the meaning of Section 14 of the Exchange Act), (B) submit any stockholder proposal (pursuant to Rule 14a-8 promulgated by the SEC under the Exchange Act or otherwise) or any notice of nomination or other business for consideration or (C) call any special meeting of the stockholders of the Company for any purpose, in each case other than for the purpose of effecting the election of the Stockholder Nominees or any Continuing Director, or the other matters set forth in Section 2(a)(ii) hereof, in accordance with the terms of this Agreement.

- (c) <u>Failure to Designate a Board Member</u>. During the Settlement Agreement Period, in the event that the Stockholder fails to recommend a replacement director (whether or not appointed to the Board) pursuant to <u>Section 2(a)(vi)</u> above within thirty (30) days of the date of the actual occurrence of a vacancy as set forth therein, such replacement director may designated by the Nominating Committee.
- (d) No Liability for Election of Recommended Directors. No Party, nor any Affiliate of any such Party, shall have any liability as a result of designating a person for election as a director for any act or omission by such designated person in his or her capacity as a director of the Company, nor shall any Party have any liability as a result of voting for or executing any written action consenting to any such designee in accordance with the provisions of this Agreement.

Section 3. Stockholder Meeting.

- (a) The Company agrees to use its reasonable best efforts to hold the Stockholder Meeting as promptly as reasonably practicable, and in any event by no later than April 30, 2013.
- (b) At the Stockholder Meeting, the Stockholder agrees to appear in person or by proxy and vote or cause to be voted all Shares in favor of (i) the election of the Stockholder Nominees and (iii) the other matters referred to in <u>Section 3(c)</u> below as recommended by the Board for approval of the stockholders of the Company (the "<u>Other Matters</u>").
- (c) The Parties acknowledge and agree that the only proposals that may be presented by the Company for consideration at the Stockholder Meeting are (i) election of the Stockholder Nominees, (ii) adoption of the Destaggering Proposal, (iii) approval, on an advisory basis, of the compensation of the Company's named executive officers, as set forth in the Preliminary Proxy Statement, (iv) ratification of the Company's independent registered public accounting firm for the current fiscal year, as set forth in the Preliminary Proxy Statement, and (v) the issuance of the shares of Common Stock, the Warrants and the shares of Common Stock underlying the Warrants to SPH pursuant to the Investment Agreement.
- (d) At any subsequent annual or special meeting of stockholders of the Company (or adjournments or postponements thereof) prior to the end of the Settlement Agreement Period, the Stockholder agrees to vote or cause to be voted all Shares in favor of the election to the Board of those director nominees nominated for election by the Nominating Committee or the Board in accordance with this Agreement and against the removal of any directors whose removal is not recommended by the Board.

Section 4. <u>Proxies</u>. Neither the Stockholder nor any of its Affiliates shall grant any proxy or enter into or agree to be bound by any voting trust with respect to any voting securities of the Company held by him, her or it, which conflicts or is inconsistent in any manner with the provisions of this Agreement, nor shall the Stockholder or any of its Affiliates enter into any stockholders agreement or other arrangement of any kind with respect to the voting securities of the Company now held or hereafter acquired by him, her or it, which conflicts or is inconsistent in any manner with the provisions of this Agreement.

Section 5. Standstill.

(a) During the period commencing on the date hereof and ending on the second anniversary of the date hereof (the "Share Acquisition Period"), without the prior consent of a majority of the Continuing Directors, neither the Stockholder nor any of its Affiliates shall acquire, or propose to acquire, "beneficial ownership" (as defined in Section 13(d) of the Exchange Act) of any equity securities, or rights or options to acquire any such securities (through purchase, exchange, conversion or otherwise), of the Company, including derivative securities representing the right to vote or economic benefits of any such securities, provided that the Stockholder and its Affiliates shall be permitted to acquire an additional 1,405,374 shares of Common Stock of the Company (as adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction prior to the expiration of the Share Acquisition Period) (the "Additional Shares"), provided, further, that the Stockholder and its Affiliates ownership in the outstanding Common Stock of the Company shall not exceed thirty percent (30%) of the then outstanding Common Stock when combined with other Equity Interests of the Company beneficially owned by HNH, the Stockholder or any of their respective Affiliates as of the date hereof (after giving effect to the transactions contemplated by the Investment Agreement (excluding the issuance or exercise of the Warrants)) (which, for purposes of such calculation, shall include shares of Common Stock issuable upon conversion of thenoutstanding preferred stock) (the "30% Cap") during the Share Acquisition Period, provided, however, that, notwithstanding the foregoing restrictions (including, without limitation, the 30% Cap), the Stockholder and its Affiliates (including HNH) shall be permitted to acquire (i) the Additional Shares, (ii) the Common Stock underlying the Warrants set forth and in accordance with the terms of the Warrants and (iii) securities as a result of, or offered pursuant to, any share division or split, share consolidation, stock/share dividend or equivalent under Delaware law, combination, recapitalization, rights offering or similar events. Notwithstanding the foregoing, in the event that the number of outstanding shares of Common Stock is either increased or decreased (other than in the case of buy-back of Common Stock made available by the Company to all of the stockholders of the Company) after the date hereof (other than in connection with the exercise of the Warrants), the number of Additional Shares that the Stockholder and its Affiliates shall be permitted to acquire shall be increased or decreased, as applicable, proportionate to the increase or decrease, as applicable, in total outstanding shares of Common Stock.

- (b) In addition to the waiver granted by the Board under Section 203 of the Delaware General Corporation Law referenced in Section 3.2(c) of the Investment Agreement (and effective as of the consummation of the issuance of Common Stock and the Warrant contemplated by the Investment Agreement), the Stockholder hereby formally requests and the Board has granted a waiver and exemption to the Stockholder and its Affiliates under and in accordance with that certain Tax Benefit Preservation Plan, dated as of October 17, 2011, between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent, as amended by Amendment No. 1 to Tax Benefit Preservation Plan, dated as of March 21, 2012, by and between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent (the "Tax Plan"), and otherwise modified that certain Rights Agreement, dated as of March 21, 2012, between the Company and American Stock Transfer & Trust Company, LLC, as Rights Agent (the "Rights Plan", and together with the Tax Plan, the "Rights Agreements") (and which waivers, exemptions and modifications shall terminate and revert back, as applicable, automatically in the event this Agreement or the Investment Agreement are terminated prior to the consummation of the issuance of such Common Stock and Warrant), to permit (x) the entering into of this Agreement and the Investment Agreement, and (y) subject to the consummation of the issuance of Common Stock and the Warrant contemplated by the Investment Agreement, the acquisition of the securities permitted to be acquired pursuant to Section 5(a), provided that for the avoidance of doubt, such waivers, exemptions and modifications to the Rights Agreements, as applicable, shall not expire upon expiration of the Share Acquisition Period as they relate to the acquisition of the Additional Shares, the Common Stock underlying the Warrants (set forth and in accordance with the terms of the Warrants) and any securities acquired pursuant to subclause (iii) of Section 5(a).
- (c) Notwithstanding anything in this Agreement to the contrary, in no event shall the Stockholder and its Affiliates be permitted to acquire shares of Common Stock or other securities or instruments in accordance with Section 5(a) that would result in an "ownership change" of the Company for purposes of Section 382 of the Internal Revenue Code of 1986, as amended.
- (d) For the avoidance of doubt, any such waiver shall not restrict the ability of the majority of the members of the Continuing Directors Committee or, if there are less than three (3) Continuing Directors, the majority of the directors on the Board who are independent and disinterested as to such matters at such time, to put in a place, a new stockholders' rights plan (which authority is hereby irrevocably devolved to such Continuing Directors Committee and directors), provided that such new plan continues to permit the Stockholder and its Affiliates to purchase the additional securities permitted by this Agreement under Section 5(a) during or following the Share Acquisition Period (including, with respect to the Warrants, at all times prior to the expiration thereof).

Section 6. <u>Issuances and Transfers of Shares</u>.

(a) In the event of any issuance or Transfer to, or purchase or other acquisition by, the Stockholder or any of its Affiliates of shares of the Company's voting securities hereafter (including, without limitation, through any share division or split, share consolidation, stock/share dividend or equivalent under Delaware law, combination, recapitalization, rights offering or similar events, exercise of warrants or otherwise), such shares shall become subject to this Agreement.

- (b) Upon any Transfer, whether directly or indirectly pursuant to the creation of a derivative security, the grant of an option or other right or the imposition of a restriction on disposition or voting, of Shares to an (i) Affiliate, or (ii) Person with whom the Stockholder has formed or joined a "group" within the meaning of Section 13(d)(3) of the Exchange Act (each, a "<u>Transferee</u>"), such Transferee shall, as a condition to such Transfer, execute a supplementary agreement with the Company agreeing to be bound by the terms and conditions of this Agreement to the same extent as the Stockholder.
 - (c) Any attempt to Transfer any Shares that is not in compliance with this <u>Section 6</u> shall be void and of no effect.
- (d) All references to numbers or percentages of shares and dollars per share in this Agreement shall be appropriately adjusted to reflect any share issuance, division or split, share consolidation, stock/share dividend or equivalent under Delaware law, combination or other recapitalization affecting Shares occurring after the date of this Agreement.

Section 7. Resignation Letter. As a condition to commencement of a term on the Board (or nomination therefor), each of the Stockholder Nominees shall provide to the Company an irrevocable letter of resignation, which shall be effective upon the earliest of (a) the date upon which a court of competent jurisdiction determines that the Stockholder materially breached its obligations to consummate the Investment Agreement in accordance with its terms and (b) (i) in the case of Mr. Kassan, the date upon which the Stockholder and its Affiliates have Transferred (other than to an Affiliate) in excess of 4,900,000 Shares in the aggregate of the Shares owned by the Stockholder and its Affiliates immediately after the conclusion of the Stockholder Meeting (including any Shares to be acquired pursuant to the Investment Agreement), and (ii) in the case of Mr. Lichtenstein, the date upon which the Stockholder and its Affiliates have Transferred (other than to an Affiliate) 9,100,000 Shares in the aggregate of the Shares owned by the Stockholder and its Affiliates immediately after the conclusion of the Stockholder Meeting (including any Shares to be acquired pursuant to the Investment Agreement). Such share thresholds shall be adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction occurring prior to the effective time of the applicable resignation letter.

Section 8. Representations and Warranties of the Company. The Company represents and warrants to the Stockholder that (a) the Company has the corporate power and authority to execute this Agreement and to bind it thereto, (b) this Agreement has been duly and validly authorized, executed and delivered by the Company, constitutes a valid and binding obligation and agreement of the Company, and is enforceable against the Company in accordance with its terms, except as enforcement thereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or similar laws generally affecting the rights of creditors and subject to general equity principles and (c) the execution, delivery and performance of this Agreement by the Company does not and will not (i) violate or conflict with any law, rule, regulation, order, judgment or decree applicable to it, or (ii) result in any breach or violation of or constitute a default (or an event which with notice or lapse

of time or both could become a default) under or pursuant to, or result in the loss of a material benefit under, or give any right of termination, amendment, acceleration or cancellation of, any organizational document, or any material agreement, contract, commitment, understanding or arrangement to which the Company is a party or by which it is bound.

Section 9. Representations and Warranties of the Stockholder. Each of HNH and the other parties listed on Exhibit A hereto jointly represents and warrants to the Company that (a) as of the date hereof, such Stockholder beneficially owns only the number of shares of Common Stock as described opposite its name on Exhibit A and Exhibit A includes all Affiliates of the Stockholder that own any securities of the Company beneficially or of record, (b) this Agreement has been duly and validly authorized, executed and delivered by such Stockholder, and constitutes a valid and binding obligation and agreement of such Stockholder, enforceable against such Stockholder in accordance with its terms, except as enforcement thereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent conveyance or similar laws generally affecting the rights of creditors and subject to general equity principles, (c) such Stockholder has the authority to execute this Agreement and to bind it thereto, (d) the execution, delivery and performance of this Agreement by such Stockholder does not and will not (i) violate or conflict with any law, rule, regulation, order, judgment or decree applicable to it, or (ii) result in any breach or violation of or constitute a default (or an event which with notice or lapse of time or both could become a default) under or pursuant to, or result in the loss of a material benefit under, or give any right of termination, amendment, acceleration or cancellation of, any organizational document, or any material agreement, contract, commitment, understanding or arrangement to which such Stockholder is a party or by which it is bound, (e) the responses to the questionnaire titled "Director Nominee Questionnaire in connection with the Stockholder Meeting of Stockholders" delivered to the Company by the Stockholder Nominees are true and correct in all material respects, at the date of this Agreement and (f) such Stockholder is not party to any agreement, arrangement, understanding or relationship, including any repurchase or similar so-called "stock borrowing" agreement or arrangement, engaged in, directly or indirectly, whether by means of derivatives or otherwise, by such Stockholder, the purpose or effect of which is to mitigate loss to, reduce the economic risk (of ownership or otherwise) of shares of any class or series of the Company by, including but not limited to "short" positions in shares of common stock, "long" puts, "short" calls, "short" forward or swap positions, manage the risk of share price changes for, or increase or decrease the voting power of, such Stockholder with respect to the shares of any class or series of the Company, or which provides, directly or indirectly, the opportunity to profit from any decrease in the price or value of the shares of any class or series of the Company.

Section 10. Mutual Non-Disparagement.

(a) The Stockholder agrees that, during the period ending on the Standstill Termination Date, neither it nor any of its Affiliates will, and it will cause each of its Affiliates not to, directly or indirectly, in any capacity or manner, make, express, transmit speak, write, verbalize or otherwise communicate in any way (or cause, further, assist, solicit, encourage, support or participate in any of the foregoing), any remark, comment, message, information, declaration, communication or other statement of any kind, whether verbal, in writing, electronically transferred or otherwise, that might reasonably be construed to be derogatory or

critical of, or negative toward, the Company or any of its directors, officers, Affiliates, subsidiaries, employees, agents or representatives (collectively, the "<u>Company Representatives</u>"), or that reveals, discloses, incorporates, is based upon, discusses, includes or otherwise involves any confidential or proprietary information of the Company or its subsidiaries or Affiliates, or to malign, harm, disparage, defame or damage the reputation or good name of the Company, its business or any of the Company Representatives. The foregoing shall not prevent (i) any statements made by a Stockholder Nominee in the course of his or her duties as a director; or (ii) the making of any factual statement in any compelled testimony or production of information, either by legal process, subpoena, or as part of a response to a request for information from any governmental authority with jurisdiction over the party from whom information is sought.

- (b) The Company hereby agrees that, during the period ending on the Standstill Termination Date, neither it nor any of its Affiliates will, and it will cause each of its Affiliates not to, directly or indirectly, in any capacity or manner, make, express, transmit, speak, write, verbalize or otherwise communicate in any way (or cause, further, assist, solicit, encourage, support or participate in any of the foregoing), any remark, comment, message, information, declaration, communication or other statement of any kind, whether verbal, in writing, electronically transferred or otherwise, that might reasonably be construed to be derogatory or critical of, or negative toward, any Stockholder, any Affiliate of any Stockholder or any of their respective directors, officers, subsidiaries, employees, agents or representatives (collectively, the "Stockholder Representatives"), or that reveals, discloses, incorporates, is based upon, discusses, includes or otherwise involves any confidential or proprietary information of any Stockholder or any of its subsidiaries or Affiliates, or to malign, harm, disparage, defame or damage the reputation or good name of any Stockholder, any Stockholder's business or any of the Stockholder Representatives.
- (c) Notwithstanding the foregoing, nothing in this <u>Section 10</u> or elsewhere in this Agreement shall prohibit any Party from making any statement or disclosure required under the federal securities laws or other applicable laws; *provided*, that such Party shall use reasonable efforts to consult with the other Parties as soon as reasonably practicable prior to making any such statement or disclosure required under the federal securities laws or other applicable laws that would otherwise be prohibited by the provisions of this <u>Section 10</u> and reasonably consider any comments of such other Parties.

Section 11. <u>Public Announcements</u>. No later than the close of business on the business day following the execution of this Agreement, the Company and the Stockholder shall jointly issue the press release (the "<u>Mutual Press Release</u>") announcing the terms of this Agreement in the form attached hereto as <u>Exhibit B</u>. Prior to the issuance of the Mutual Press Release, neither the Company nor the Stockholder shall issue any press release or public announcement regarding this Agreement or take any action that would require public disclosure thereof without the prior written consent of the other Party. Prior to the Standstill Termination Date, no Party or any of its Affiliates shall make any public statement (including, without limitation, in any filing required under Regulation 13D under the Exchange Act) concerning the subject matter of this Agreement inconsistent with the Mutual Press Release.

Section 12. <u>Legend</u>. The Stockholder agrees with the Company that the certificates evidencing the Shares will bear the following legend, which the Company shall cause to be promptly removed in connection with any Transfer not subject to Section 6(b):

"THE SECURITIES EVIDENCED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS SET FORTH IN A SETTLEMENT AGREEMENT BETWEEN THE COMPANY AND HANDY & HARMAN LTD., DATED AS OF FEBRUARY 11, 2013. A COPY OF SUCH SETTLEMENT AGREEMENT MAY BE OBTAINED FROM THE COMPANY UPON REQUEST."

Section 13. Term. This Agreement shall be effective as of the date hereof and, except as otherwise set forth in this Agreement, including in Section 2(a) and Section 5, shall continue in effect until, and shall terminate immediately upon, the earliest to occur of (a) the written agreement of the Parties to terminate this Agreement, (b) such time as the Stockholder and its Affiliates have Transferred (other than to an Affiliate) in excess of 9,100,000 Shares (as adjusted for any stock dividend, stock split, stock combination, reclassification or other similar transaction) in the aggregate (including any Shares acquired pursuant to the Investment Agreement), (c) the end of the Settlement Agreement Period (d) upon notice by either the Company or the Stockholder if the Investment Agreement is terminated in accordance with Section 7.1(d) or (e) thereof, (e) upon notice by the Company if the Investment Agreement is terminated in accordance with Section 7.1(c) thereof and (f) upon notice by either the Company or HNH if the Stockholder Nominees are not elected as directors at the Stockholder Meeting and the transactions contemplated by the Investment Agreement are not consummated. If this Agreement is terminated pursuant to the provisions of this Section 13, this Agreement shall immediately become void and there shall be no further obligations on the part of either Party or their respective directors, officers, employees, agents or representatives, except (a) with respect to the provisions of this Section 13, 15 and 16, and (b) with respect to any liabilities or damages incurred or suffered as a result of the willful and material breach prior to such termination by the Company or the Stockholder of any of their respective representations, warranties, covenants or other agreements set forth in this Agreement.

Section 14. Remedies.

- (a) The Company agrees to use its commercially reasonable best efforts, within the requirements of applicable law, to ensure that the rights granted under this Agreement are effective and that the Parties enjoy the benefits of this Agreement. Such actions include, without limitation, the use of the Company's commercially reasonable best efforts to cause the nomination and election of the directors as provided in this Agreement.
- (b) Each Party acknowledges and agrees that (a) the covenants, obligations and agreements of each Party contained in this Agreement relate to special, unique and extraordinary matters and (b) a violation of any of the covenants, obligations or agreements contained in this Agreement by a Party will cause the other Party irreparable injury for which adequate remedies are not available at law. Therefore, the Parties agree that each Party shall be entitled to an injunction, restraining order or such other equitable relief (without the requirement to post bond) as a court of competent jurisdiction may deem necessary or appropriate to restrain the other Party from committing any violation of such covenants, obligations or agreements and

to specifically enforce the terms of this Agreement, in addition to any other remedies available at law or in equity. Furthermore, the Parties hereby waive, and shall use their commercially reasonable best efforts to cause their stockholders, members, partners, Affiliates and representatives to waive, any requirement for the securing or posting of any bond in connection with such remedy.

(c) All remedies, either under this Agreement or by law or otherwise afforded to any Party, shall be cumulative and not alternative.

Section 15. Notice. Any notices, consents, determinations, waivers or other communications required or permitted to be given under the terms of this Agreement must be in writing and will be deemed to have been delivered: (i) upon receipt, when delivered personally; (ii) upon receipt, when sent by facsimile (provided confirmation of transmission is mechanically or electronically generated and kept on file by the sending Party); or (iii) one (1) business day after deposit with a nationally recognized overnight delivery service, in each case properly addressed to the Party to receive the same. The addresses and facsimile numbers for such communications shall be (or such other address for a Party as shall be specified by like notice):

To the Company:

ModusLink Global Solutions, Inc. 1601 Trapelo Road, Suite 170 Waltham, Massachusetts 02451 Attention: Peter L. Gray Tel: 781-663-5024

Fax: 781-663-5095

Email: peter_gray@moduslink.com

with a copy to:

Latham & Watkins LLP
233 South Wacker Drive, Suite 5800
Chicago, Illinois 60610
Attention: Mark D. Gerstein
Zachary A. Judd
Tel: 312-876-7700

Fax: 312-8/6-//00

Email: mark.gerstein@lw.com zachary.judd@lw.com

To the Stockholder:

c/o Steel Partners II, L.P. 590 Madison Avenue New York, New York 10022 Attention: Warren Lichtenstein Leonard McGill

Tel: 212-520-2300 Fax: 212-520-2337

Email: warren@steelpartners.com len@steelpartners.com

With a copy to:

Olshan Frome Wolosky LLP 65 East 55th Street New York, New York 10022 Attention: Steven Wolosky Michael R. Neidell

Tel: 212-451-2300 Fax: 212-451-2222

Email: swolosky@olshanlaw.com mneidell@olshanlaw.com

Section 16. Governing Law; Jurisdiction; Venue; Waiver of Jury Trial. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO ANY CHOICE OF LAW OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY JURISDICTION OTHER THAN THE STATE OF NEW YORK. Each of the Parties hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of any Delaware state court, or federal court of the United States of America, sitting in Delaware, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement or the agreements delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each of the Parties hereby irrevocably and unconditionally (i) agrees not to commence any such action or proceeding except in such courts, (ii) agrees that any claim in respect of any such action or proceeding may be heard and determined in such Delaware State court or, to the extent permitted by law, in such federal court, (iii) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such Delaware state or federal court, and (iv) waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such Delaware state or federal court. Each of the Parties agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Each Party irrevocably consents to service of process in the manner provided for notices in Section 15. Nothing in this Agreement will affect the right of any Party to serve process in any other manner permitted by law. Each Party irrevocably and unconditionally waives all right to trial by jury in any action, proceeding or counterclaim (whether based on contract, tort or otherwise) arising out of or relating to this Agreement or the actions of the Company in the negotiation, administration, performance and enforcement hereof.

Section 17. Entire Agreement. Except for the Investment Agreement and the Registration Rights Agreement (as defined in the Investment Agreement), this Agreement contains the entire agreement by and between the Company and the Stockholder with respect to the transactions contemplated by this Agreement and supersedes all prior agreements and representations, written or oral, with respect thereto.

Section 18. Receipt of Adequate Information; No Reliance; Representation by Counsel. Each Party acknowledges that it has received adequate information to enter into this Agreement, that is has not relied on any promise, representation or warranty, express or implied not contained in this Agreement and that it has been represented by counsel in connection with this Agreement. Accordingly, any rule of law or any legal decision that would provide any Party with a defense to the enforcement of the terms of this Agreement against such Party shall have no application and is expressly waived. The provisions of the Agreement shall be interpreted in a reasonable manner to effect the intent of the Parties.

Section 19. Amendments and Waivers. This Agreement may be amended, modified, superseded, cancelled, renewed or extended, and the terms and conditions of this Agreement may be waived only by a written instrument signed by the Parties or, in the case of a waiver, by the Party waiving compliance. No delay on the part of either Party in exercising any right, power or privilege pursuant to this Agreement will operate as a waiver thereof, nor will any waiver on the part of either Party of any right, power or privilege pursuant to this Agreement, nor will any single or partial exercise of any right, power or privilege pursuant to this Agreement, preclude any other or further exercise thereof or the exercise of any other right, power or privilege pursuant to this Agreement. The rights and remedies provided pursuant to this Agreement are cumulative and are not exclusive of any rights or remedies which any Party otherwise may have at law or in equity.

Section 20. <u>Successors and Assigns</u>. This Agreement will be binding upon and inure to the benefit of both of the Parties, and neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by either Party without the prior written consent of the other Party (except for assignments by the Stockholder to any of its Affiliates, subject to compliance with Section 6). Any attempt to so assign or otherwise transfer this Agreement without such consent shall be void and of no effect. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the respective successors and permitted assigns of the Parties. Nothing in this Agreement, express or implied, is intended to confer upon any party other than the Parties or their respective successors and permitted assigns any rights, remedies, obligations, or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement.

Section 21. Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of such Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the extent possible.

- Section 22. <u>Expenses</u>. Except as otherwise provided herein or in the Investment Agreement, all costs and expenses incurred in connection with the transactions contemplated by this Agreement shall be paid by the Party incurring such costs and expenses.
- Section 23. <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- Section 24. <u>Counterparts</u>. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each Party hereto shall have received a counterpart hereof signed by the other Parties hereto. Counterparts delivered by electronic transmission shall be deemed to be originally signed counterparts.
- Section 25. <u>Further Assurances</u>. At any time or from time to time after the date hereof, the Parties agree to cooperate with each other, and at the request of any other Party, to execute and deliver any further instruments or documents and to take all such further action as the other Party may reasonably request in order to evidence or effectuate the consummation of the transactions contemplated hereby and to otherwise carry out the intent of the Parties under this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, the Parties hereto have duly executed and delivered this Agreement as of the date first above written.

MODUSLINK GLOBAL SOLUTIONS, INC.

By: /s/ Peter L. Gray

Name: Peter L. Gray

Title: Executive Vice President, Chief Administrative Officer and General

Counsel

HANDY & HARMAN LTD.

By: /s/ Warren Lichtenstein

Name: Warren Lichtenstein

Title:

[Signature Page to Settlement Agreement]

EXHIBIT A

<u>Stockholder</u>	Shares of Common Stock Beneficially Owned
Handy & Harman Ltd.	5,941,170
Steel Partners, Ltd.	60,000
Steel Partners Holdings L.P.	540,015
SPH Group LLC	540,015
SPH Group Holdings LLC	540,015
Steel Partners Holdings GP Inc.	540,015
Warren G. Lichtenstein	60,000
Glen M. Kassan	0
Total:	6,541,185

EXHIBIT B

Mutual Press Release

CONSULTING AGREEMENT

THIS CONSULTING AGREEMENT ("Agreement") by and between ModusLink Global Solutions, Inc., a Delaware corporation (the "Company"), and Edward E. Lucente (the "Consultant") is made as of February 11, 2013.

WHEREAS, Consultant has been a member of the Company's board of directors (the "Board") since 2006;

WHEREAS, pursuant to the terms of that certain Settlement Agreement, dated February 11, 2013, by and among the Company, Handy & Harman, Ltd and its affiliates party thereto, (the "Settlement Agreement"), Consultant has voluntarily agreed to resign from the Board effective upon and only in the event that the Board accepts his resignation following the closing of the transactions contemplated by that certain Investment Agreement, dated February 11, 2013, by and between the Company and Steel Partners Holdings, L.P. The date such closing occurs is referred to herein as the "Effective Date"); and

WHEREAS, Consultant and the Company mutually desire for Consultant to provide consulting services to the Company and its subsidiaries and affiliates commencing on the Effective Date pursuant to the terms of this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants set forth herein, the Company and Consultant hereby agree as follows:

- **1. Consulting Period.** the Company hereby engages Consultant to perform the services described in Section 2 below during the period commencing on the Effective Date and ending on the date that is 12 months after the Effective Date (the "Consulting Period").
- **2. Services.** Consultant shall perform such services as may be requested by the Company and/or its subsidiaries from time to time during the Consulting Period, regarding the transition of his duties and responsibilities to a successor or successor member of the Board or regarding the operational and strategic plans for the Company, but not in excess of 5 hours per week. Such services will be performed at mutually agreeable times and shall not interfere with Consultant's other businesses or endeavors.
- **3. Compensation.** For his services under this Agreement, the Company shall pay Consultant a fee of \$175,000.00 on the Effective Date. Such fee shall be paid regardless of whether or not the Company or any of its subsidiaries actually requests services from Consultant under Section 2 above.

4. Non-Disclosure Agreement.

4.1 Consultant agrees that all information, whether or not in writing, of a private, secret or confidential nature concerning the Company's business, business relationships or financial affairs (collectively, "Proprietary Information") is and shall be the exclusive property of the Company. By way of illustration, but not limitation, Proprietary Information shall include discoveries, inventions, products, product improvements, product enhancements, processes, methods, techniques, formulas, compositions, negotiation strategies and positions, projects, developments, plans (including business and marketing plans), research data, financial data

(including sales, costs, profits, pricing methods), personnel data, computer programs (including software used pursuant to a license agreement), customer and supplier lists, and contacts at or knowledge of customers or prospective customers of the Company. Consultant shall not disclose any Proprietary Information to any person or entity other than employees of the Company (or such third parties as the Company may authorize) or use the same for any purposes (other than in the performance of his duties as a consultant to the Company) without written approval by an officer of the Company, either during or after the Consulting Period, unless and until such Proprietary Information has become public knowledge without fault by Consultant.

- 4.2 Consultant agrees that all files, documents, letters, memoranda, reports, records, data, sketches, drawings, models, laboratory notebooks, program listings, computer equipment or devices, computer programs or other written, photographic, or other tangible material containing Proprietary Information, whether created by Consultant or others, which have or shall come into his custody or possession, shall be and are the exclusive property of the Company to be used by Consultant only in the performance of his duties as a consultant to the Company and shall not be copied or removed from the Company premises except in the pursuit of the business of the Company. All such materials or copies thereof and all tangible property of the Company in the custody or possession of Consultant shall be delivered to the Company, upon the earlier of (i) a request by the Company or (ii) termination of the Consulting Period. After such delivery, Consultant shall not retain any such materials or copies thereof or any such tangible property.
- 4.3 Consultant agrees that his obligation not to disclose or to use information and materials of the types set forth in paragraphs 4.1 and 4.2 above, and his obligation to return materials and tangible property, set forth in paragraph 4.2 above, also extends to confidential and proprietary information, materials and tangible property of customers, suppliers and corporate affiliates of the Company, including, without limitation, parents and subsidiaries (both direct and indirect) of the Company, and other third parties who may have disclosed or entrusted the same to the Company or to Consultant ("Third Party Information"). Consultant agrees to treat Third Party Information in the same manner as Consultant is required to treat Proprietary Information under paragraphs 4.1 and 4.2. Consultant acknowledges that the Company is now and may hereafter be subject to non-disclosure or confidentiality agreements with third persons pursuant to which the Company must protect or refrain from the use of proprietary information which is the property of such third persons. Consultant agrees upon the directive of the Company to be bound by the terms of such agreements in the event Consultant has access to proprietary information protected thereunder to the same extent as if Consultant was an original individual signatory thereto.

5. Successors.

- 5.1 This Agreement is personal to Consultant and without the prior written consent of the Company shall not be assignable by Consultant. However, this Agreement shall inure to the benefit of and be enforceable by Consultant's heirs and legal representatives.
- 5.2 This Agreement shall inure to the benefit of and are binding upon the Company and its successors and assigns, and may be assigned by the Company to any subsidiary or to any successor in interest in connection with the sale of the Company or all or substantially all of the assets of the Company.

6. Independent Contractor Status.

- 6.1 the Company and Consultant do hereby acknowledge that after the Effective Date the relationship between the Company and Consultant shall be solely that of an independent contractor and that Consultant shall not be treated as an employee for any purpose. Accordingly, Consultant shall have no authority or right to bind or enter into any agreements on behalf of and is not an agent of the Company, or any subsidiary thereof.
 - 6.2 Consultant shall be responsible for all taxes, including self-employment taxes due on payments made under this Agreement.

7. Miscellaneous.

- 7.1 This Agreement may not be amended or modified otherwise than by a written agreement executed by the parties hereto or their respective successors or legal representatives.
- 7.2 The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement.
- 7.3 Any party's failure to insist upon strict compliance with any provision hereof shall not be deemed to be a waiver of such provision or any other provision hereof.
- 7.4 This Agreement may be executed simultaneously in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument.
- 7.5 This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without regard to its conflicts of laws principles.
- 7.6 This Agreement includes the entire agreement and understanding between the parties and supersedes any prior agreements, written or oral, with respect to the subject matter hereof.
- 7.7 All descriptive headings of sections and paragraphs in this Agreement are for convenience of reference only, and they form no part of this Agreement and shall not affect its interpretation.

IN WITNESS WHEREOF, the parties have executed this Consulting Agreement as of the date first above written.

ModusLink Global Solutions, Inc.

By: /s/ Peter L. Gray

Name: Peter L. Gray

Title: Executive Vice President, Chief Administrative Officer, General Counsel and Secretary

/s/ Edward E Lucente

Edward E. Lucente

[Consulting Agreement – Lucente]

MODUSLINK ANNOUNCES \$30 MILLION INVESTMENT AGREEMENT WITH STEEL PARTNERS HOLDINGS L.P.

Steel Holdings agrees to acquire 7.5 million ModusLink shares at \$4 per share, representing 45% premium

Company announces settlement with Handy & Harman Ltd. in relation to upcoming annual meeting of stockholders

WALTHAM, Mass., – **February 11, 2013** – ModusLink Global Solutions^(TM), Inc. (NASDAQ: MLNK) today announced that it has entered into an investment agreement with Steel Partners Holdings L.P. (NYSE: SPLP, "Steel Holdings"), which together with certain affiliates, including Handy & Harman Ltd. (NASDAQ: HNH, "Handy & Harman") (together, the "Steel Group"), beneficially owns 14.9 percent of ModusLink's outstanding shares.

Under the terms of the agreement, Steel Holdings would purchase 7.5 million newly issued shares of common stock at a price of \$4.00 per share, representing a cash investment in the Company, before fees and expenses, of \$30 million. The \$4.00 purchase price per share represents a 45 percent premium to the closing market price for ModusLink common stock on Friday, February 8, 2013. In addition, at the closing of the transaction the Company would issue Steel Holdings warrants to acquire 2.0 million shares at an exercise price of \$5.00 per share.

In addition, the Steel Group may purchase up to approximately 1.4 million shares of ModusLink's outstanding common stock, subject to proportionate adjustment. If all stock is purchased and all warrants are exercised as permitted under the agreement, the Steel Group would own approximately 32.6 percent of ModusLink's outstanding shares. The investment is subject to certain enumerated closing conditions, including shareholder approval of the investment pursuant to Nasdaq Listing Rule 5635(b) and the election of two Steel Group designees to the ModusLink board of directors at the Company's 2012 Annual Meeting of Stockholders (the "Annual Meeting").

In connection with the investment agreement, ModusLink also announced that it has reached a settlement agreement with Handy & Harman in relation to the Annual Meeting. Under the terms of the settlement agreement, ModusLink has agreed to nominate and solicit proxies solely for the Steel Group representatives Warren G. Lichtenstein and Glen M. Kassan for election at the Annual Meeting as Class I directors. Mr. Lichtenstein is Chairman of the Board of Directors of Handy & Harman and Chairman of the Board and Chief Executive Officer of the general partner of Steel Holdings. Mr. Kassan is Vice Chairman of the Board of Directors of Handy & Harman and Managing Director at Steel Partners LLC. Handy & Harman has agreed to withdraw its preliminary proxy statement and to end its proxy solicitation, and has entered into certain other standstill arrangements with the Company.

Upon consummation of the investment, ModusLink Directors Edward E. Lucente and Joseph M. O'Donnell would step down from the Board, and current Chairman Francis J. Jules and Director and Audit Committee Chairman Michael J. Mardy, whose terms on the Board conclude at the Annual Meeting, would each be reappointed to the Board as Class II directors. In addition, at that time, Mr. Lichtenstein would be designated Chairman of the Board of the Company. The size of the Board would be fixed at seven directors immediately following the Annual Meeting.

ModusLink expects to announce the date and location of the Annual Meeting in connection with the filing of its definitive proxy statement concerning the transaction which will, in addition to seeking the election of the Steel Group representatives and approval of the investment, seek approval of the declassification of the Company's Board of Directors such that the annual election of all directors would take place beginning at the 2013 annual meeting. The Board is submitting this latter proposal in response to the approval at the 2011 annual meeting of a shareholder proposal urging such change.

"The commitment of new capital from Steel Holdings at a significant premium to the recent price of ModusLink shares is a strong vote of confidence in ModusLink's future by our largest stockholder," said Mr. Jules. "The investment also validates the recent strategic changes and improvements the Company has undertaken, and further strengthens the Company's balance sheet. We are also pleased to reach an agreement where the Company will be nominating Warren Lichtenstein and Glen Kassan, two highly qualified nominees, for election to ModusLink's board of directors at our upcoming annual meeting. The Board looks forward to the opportunity to work with Warren and Glen to increase stockholder value."

"We have worked constructively with ModusLink to reach these agreements and are excited by the opportunity to invest further in the Company's future and to contribute significantly to the creation of value for all ModusLink stockholders," said Mr. Lichtenstein. "I believe ModusLink has considerable potential for value creation as evidenced by Steel Holdings' investment. I also believe that Glen and I are well suited to help the Company towards that objective and we look forward to working with the rest of the Board and ModusLink management to that end."

Complete details of the investment agreement and settlement agreement will be contained in the Company's Current Report on Form 8-K, which will be filed with the Securities and Exchange Commission.

Goldman, Sachs & Co. is serving as financial advisor and Latham & Watkins LLP is serving as legal counsel to ModusLink. Olshan Frome Wolosky LLP is serving as legal counsel to Steel Holdings and Handy & Harman.

About Warren G. Lichtenstein

Warren G. Lichtenstein has served as the Chairman of the Board and Chief Executive Officer of the general partner of Steel Holdings, a global diversified holding company that owns and operates businesses and has significant interests in leading companies in a variety of industries, including diversified industrial products, energy, defense, banking, and food products and services, since July 15, 2009. He is also the Chairman and Chief Executive Officer of Steel Partners LLC, a subsidiary of Steel Holdings, and has been associated with Steel Partners LLC and its affiliates since 1990. Mr. Lichtenstein has served as Chairman of the Board of Handy & Harman, a diversified global industrial company, since July 2005. He is a Co–Founder of Steel Partners Japan Strategic Fund (Offshore), L.P., a private investment partnership investing in Japan, and Steel Partners China Access I LP, a private equity partnership investing in China. He also co–founded Steel Partners II, L.P., a private investment partnership that is now a

wholly—owned subsidiary of Steel Holdings, in 1993. Mr. Lichtenstein has served as a director of GenCorp Inc., a manufacturer of aerospace and defense products and systems with a real estate business segment, since March 2008. He has served as a director of SL Industries, Inc. ("SLI"), a company that designs, manufactures and markets power electronics, motion control, power protection, power quality electromagnetic and specialized communication equipment, since March 2010. From May 2001 to November 2007, Mr. Lichtenstein served as a director (formerly Chairman of the Board) of United Industrial Corporation ("United Industrial"), a company principally focused on the design, production and support of defense systems, which was acquired by Textron Inc.

About Glen M. Kassan

Glen M. Kassan is a Managing Director and operating partner of Steel Partners LLC, a global management firm, and has been associated with Steel Partners LLC and its affiliates since August 1999. Mr. Kassan served as the Vice President, Chief Financial Officer and Secretary of a predecessor entity of Steel Holdings from June 2000 to April 2007. Mr. Kassan has served as a director of Handy & Harman since July 2005, as the Vice Chairman of the Board since October 2005 and as Chief Executive Officer from 2005 to 2012. He has served as a director of SLI since January 2002 and its Chairman of the Board since May 2008. He previously served as SLI's Vice Chairman of the Board from August 2005 to May 2008, its President from February 2002 to August 2005, its interim Chief Executive Officer from June 14, 2010 to June 29, 2010 and its interim Chief Financial Officer from June 14, 2010 to August 30, 2010. He was a director of United Industrial from October 2002 to November 2007.

About ModusLink Global Solutions

ModusLink Global Solutions Inc. (NASDAQ: MLNK) executes comprehensive supply chain and logistics services that improve clients' revenue, cost, sustainability and customer experience objectives. ModusLink is a trusted and integrated provider to the world's leading companies in consumer electronics, communications, computing, medical devices, software, luxury goods and retail. The Company's operating infrastructure annually supports more than \$80 billion of its clients' revenue and manages approximately 470 million product shipments through more than 30 sites in 15 countries across North America, Europe, and the Asia/Pacific region. For details on ModusLink's flexible and scalable solutions visit www.moduslink.com and www.valueunchained.com, the blog for supply chain professionals.

ModusLink Global Solutions is a registered trademark of ModusLink Global Solutions, Inc. All other company names and products are trademarks or registered trademarks of their respective companies.

Important Additional Information

ModusLink, its directors and certain of its executive officers and employees are participants in a solicitation of proxies in connection with its 2012 annual meeting of stockholders (the "2012 Annual Meeting"). Important information concerning the identity and interests of these persons is available in ModusLink's preliminary proxy statement filed with the Securities and Exchange Commission (the "SEC") on December 21, 2012, as amended on January 7, 2013. ModusLink plans to file with the SEC and mail to its stockholders a definitive proxy statement in connection with the 2012 Annual Meeting. Information regarding the identity of the participants, and their direct or indirect interests, by security holdings or otherwise, is set forth in the preliminary proxy statement and ModusLink's Annual Report on Form 10-K for the year ended July 31, 2012. To the extent holdings of ModusLink securities have changed since the amounts printed in the preliminary proxy statement for the 2012 Annual Meeting, such changes have been or will be reflected on Statements of Change in Ownership on Form 4 filed with the SEC.

Copies of ModusLink's preliminary proxy statement, any other relevant documents and other materials filed with the SEC concerning ModusLink, when filed, may be obtained free of charge at http://www.sec.gov and http://www.ir.moduslink.com. Stockholders should carefully read the proxy statement and the accompanying proxy card when they become available before making any voting decision.

Forward-Looking Statements

This press release contains forward-looking statements, which address a variety of subjects including, for example, the consummation and impact of the announced transaction and the Company's prospects for creating value for stockholders. All statements other than statements of historical fact, including without limitation, those with respect to the Company's goals, plans, expectations and strategies set forth herein are forward-looking statements. The following important factors and uncertainties, among others, could cause actual results to differ materially from those described in these forward-looking statements: the announced transaction is subject to certain enumerated closing conditions and there can be no assurance that such conditions will be met and the transaction will be consummated; the Company's success, including its ability to meet its revenue, operating income and cost savings targets, maintain and improve its cash position, expand its operations and revenue, lower its costs, improve its gross margins, reach and sustain profitability, reach its long-term objectives and operate optimally, depends on its ability to execute on its business strategy, including the investment and costs savings plan and the continued and increased demand for and market acceptance of its services; global economic conditions, especially in the technology sector are uncertain and subject to volatility; demand for our clients' products may decline or may not achieve the levels anticipated by our clients; the Company's management may face strain on managerial and operational resources as they try to oversee the expanded operations; the Company may not realize the expected benefits of its restructuring and cost cutting actions; the Company may not be able to expand its operations in accordance with its business strategy; the Company's cash balances may not be sufficient to allow the Company to meet all of its business and investment goals; the Company may experience difficulties integrating technologies, operations and personnel in accordance with its business strategy; the Company derives a significant portion of its revenue from a small number of customers and the loss of any of those customers could significantly damage the Company's financial condition and results of operations; the Company frequently sells to its supply chain management clients on a purchase order basis rather than pursuant to contracts with minimum purchase requirements, and therefore its sales and the amount of projected revenue that is actually realized are subject to demand variability; risks inherent with conducting international operations; tax rate expectations are based on current tax law and current expected income and may be affected by the jurisdictions in which profits are determined to be earned and taxed, changes in estimates of credits, benefits and deductions, the resolution of issues arising from tax audits with various tax authorities, including payment of interest and penalties and the ability to realize deferred tax assets; the potential tax benefits represented by the net operating loss carryforwards may not be realized and the tax benefit preservation plan may not be effective in preserving those benefits; the mergers and acquisitions and IPO markets are inherently unpredictable and liquidity events for companies in the Company's venture capital portfolio may not occur; and increased competition and technological changes in the markets in which the Company competes. For a detailed discussion of cautionary statements that may affect the Company's future results of operations and financial results, please refer to the Company's filings with the Securities and Exchange Commission, including the Company's most recent Annual Report on Form 10-K and Quarterly Reports on Form 10-Q. Forwardlooking statements represent management's current expectations and are inherently uncertain. We do not undertake any obligation to update forward-looking statements made by us.

Contact:

Robert Joyce 781-663-5120 ir@moduslink.com